DISSERTATION

OPTIMIZING RESILIENCE DECISION-SUPPORT FOR NATURAL GAS NETWORKS UNDER UNCERTAINTY

Submitted by

Mohammad Reza Ameri

Department of Civil and Environmental Engineering

In partial fulfillment of the requirements

For the Degree of Doctor of Philosophy

Colorado State University

Fort Collins, Colorado

Summer 2019

Doctoral Committee:

Advisor: John W. van de Lindt

Suren Chen Gaofeng Jia Martin Shields Copyright by Mohammad Reza Ameri 2019

All Rights Reserved

ABSTRACT

OPTIMIZING RESILIENCE DECISION-SUPPORT FOR NATURAL GAS NETWORKS UNDER UNCERTAINTY

Community resilience in the aftermath of a hazard requires the functionality of complex, interdependent infrastructure systems become operational in a timely manner to support social and economic institutions. In the context of risk management and community resilience, critical decisions should be made not only in the aftermath of a disaster in order to immediately respond to the destructive event and properly repair the damage, but preventive decisions should to be made in order to mitigate the adverse impacts of hazards prior to their occurrence. This involves significant uncertainty about the basic notion of the hazard itself, and usually involves mitigation strategies such as strengthening components or preparing required resources for postevent repairs. In essence, instances of risk management problems that encourage a framework for coupled decisions before and after events include modeling how to allocate resources before the disruptive event so as to maximize the efficiency for their distribution to repair in the aftermath of the event, and how to determine which network components require preventive investments in order to enhance their performance in case of an event. In this dissertation, a methodology is presented for optimal decision making for resilience assessment, seismic risk mitigation, and recovery of natural gas networks, taking into account their interdependency with some of the other systems within the community. In this regard, the natural gas and electric power networks of a virtual community were modeled with enough detail such that it enables assessment of natural gas network supply at the community level. The effect of the industrial makeup of a community on its natural gas recovery following an earthquake, as well as the effect of replacing conventional steel pipes with ductile HDPE pipelines as an effective mitigation strategy against seismic hazard are investigated. In addition, a multi-objective optimization framework that integrates probabilistic seismic risk

assessment of coupled infrastructure systems and evolutionary algorithms is proposed in order to determine cost-optimal decisions before and after a seismic event, with the objective of making the natural gas network recover more rapidly, and thus the community more resilient. Including bi-directional interdependencies between the natural gas and electric power network, strategic decisions are pursued regarding which distribution pipelines in the gas network should be retrofitted under budget constraints, with the objectives to minimizing the number of people without natural gas in the residential sector and business losses due to the lack of natural gas in non-residential sectors. Monte Carlo Simulation (MCS) is used in order to propagate uncertainties and Probabilistic Seismic Hazard Assessment (PSHA) is adopted in order to capture uncertainties in the seismic hazard with an approach to preserve spatial correlation. A non-dominated sorting genetic algorithm (NSGA-II) approach is utilized to solve the multi-objective optimization problem under study. The results prove the potential of the developed methodology to provide risk-informed decision support, while being able to deal with large-scale, interdependent complex infrastructure considering probabilistic seismic hazard scenarios.

ACKNOWLEDGEMENTS

Throughout the years I was working on my Ph.D., I received a great deal of support and assistance. First and foremost, I would like to express my most sincere gratitude and appreciation to one of my greatest role models, Prof. John van de Lindt, whom I was privileged to have as my Ph.D. advisor. I could not have imagined a better mentor, both for my studies and in my life, and this work would have never been possible without his tremendous support. Thank you, Dr. van de Lindt, for your inspiration, guidance, and invaluable insights. Working with you not only taught me how to be a good researcher, but also showed me the true meaning of a good person.

I would also like to recognize and thank my fantastic doctoral committee members, Dr. Suren Chen, Dr. Gaofeng Jia, and Dr. Martin Shields, for providing me with additional guidance and valuable critiques and insights.

Special thanks also goes out to the National Institute of Standards and Technology (NIST) for providing the funding for my research. I was also honored to receive two scholarships with the Jack E. Cermak Wind Engineering Graduate Fellowship and would like to take this opportunity and thank both the provider and the selection committee for choosing me for this prestigious award.

Last but not least, I would like to thank my parents whose love and support have always been with me throughout all my endeavors. I would also like to thank my beloved siblings who have always been a great emotional support in my life. I am also grateful to all my friends and family members, university staff, people I have met during this journey and anybody who has supported me along the way. I appreciate you all.

DEDICATION

To my beloved parents for their unconditional love, support and sacrifices

TABLE OF CONTENTS

ABSTRACTi
ACKNOWLEDGEMENTS iv
DEDICATION
LIST OF TABLES
LIST OF FIGURES
CHAPTER 1: INTRODUCTION
CHAPTER 2: COMPONENT MODELS FOR NATURAL GAS NETWORKS 12
Network Overview and Definitions of Components
Components Damage Models 18
CHAPTER 3: APPROACHES FOR MODELLING INTERDEPENDENCE
Empirical Approaches
Agent-Based Approaches
System Dynamics Based Approaches
Economic Theory Based Approaches
Network-Based Approaches
Topology-based methods
Flow-based methods
Interdependence Approach for NGN and Other Systems
CHAPTER 4: MODELLING EARTHQUAKE HAZARD
Formulation of PSHA
Identifying seismic sources

Characterizing the distribution of magnitudes	36
Characterizing the distribution of source-to-site distances	37
Predicting the resulting distribution of ground motion intensity	38
Combining uncertainties using the total probability theorem	40
PSHA Deaggregation	42
Cause of Damage to Natural Gas Components	44
CHAPTER 5: RESTORATION ANALYSIS	46
Methodology and Procedure	48
Mitigation Strategies	52
Optimized Network Recovery	54
CHAPTER 6: ILLUSTRATIVE COMMUNITY MODEL AND SIMULATIONS	62
The Centerville NGN and EPN	63
Illustrative Scenarios	66
Results	74
Effects of industrial makeup and HDPE pipelines	74
Effects of bi-directional interdependencies and optimized network recovery	80
CHAPTER 7: OPTIMAL DECISION MAKING UNDER UNCERTAINTY	89
Multi-Objective Optimization	91
Non-dominated Sorting Genetic Algorithm II (NSGA-II)	93
NSGA-II Procedure	95
Decision Making Framework for Disaster Preparedness and Recovery	97
Risk Assessment Methodology	98
Optimization Problem and Objective Functions	99

Retrofit and Repair Strategy for the Centerville NGN	. 100
Results	. 102
CHAPTER 8: SUMMARY, CONTRIBUTIONS, AND RECOMMENDATIONS	. 111
Summary	. 111
Contributions	. 112
Recommendations	. 114
References	. 116

LIST OF TABLES

Table 6-1. Average annual natural gas consumption for each building archetype in Centerville	73
Table 6-2. Summary of analyses types and corresponding analysis number	76
Table 6-3. Resilience metrics for the studied scenarios	79
Table 6-4. Summary of analyses types and corresponding analysis IDs	80
Table 7-1. Results for the earthquake with 10% chance in 50 years ($RP = 475$ yrs) 1	08
Table 7-2. Results for the earthquake with 2% chance in 50 years (RP = 2475 yrs) 1	08

LIST OF FIGURES

Figure 2-1: U.S. Natural Gas Consumption by Sector
Figure 2-2: Natural Gas Processing Plant
Figure 2-3: Exterior View of a Natural Gas Compressor Station
Figure 2-4: Interior View of a Natural Gas Compressor Station
Figure 2-5: Natural Gas City Gate Station
Figure 2-6: Natural Gas Processing Plant
Figure 4-1: Distribution of recordings of the NGA-West2 database with respect to moment
magnitude and rupture distance
Figure 4-2: Hazard curve for Los Angeles downtown
Figure 4-3: PSHA deaggregation for Los Angeles downtown
Figure 5-1: Flowchart for calculating functionality fragility and restoration process
Figure 5-2: Resilience loss measurement
Figure 5-3: Flowchart for fiinding the optimized restoration sequence
Figure 5-4: Example of crossover operator application in optimizing repair sequence
Figure 5-5: Example of mutation operator application in optimizing repair sequence
Figure 6-1. Plan of Centerville
Figure 6-2. Centerville Natural Gas Network
Figure 6-3. Supported Zones by each District Regulating Station
Figure 6-4. Centerville Electric Power Network
Figure 6-5. PGA map for the simulated 6.5 magnitude earthquake
Figure 6-6. PGV map for the simulated 6.5 magnitude earthquake

Figure 6-7. PGA map for the simulated 7.9 magnitude earthquake
Figure 6-8. PGV map for the simulated 7.9 magnitude earthquake
Figure 6-9. Liquefaction susceptibility map for Centerville
Figure 6-10. Bi-directional interdependency between NGN and EPN
Figure 6-11. Restoration curves for the natural gas network with steel distribution pipes in case of
magnitude 6.5 earthquake: (a) Scenario A: General, (b) Scenario B: Residential, and (c) Scenario
C: Industrial
Figure 6-12. Restoration curves for the natural gas network with HDPE distribution pipes in case
of magnitude 6.5 earthquake: (a) Scenario A: General, (b) Scenario B: Residential, and (c)
Scenario C: Industrial
Figure 6-13. Restoration curves for the natural gas network with steel distribution pipes in case of
magnitude 7.9 earthquake: (a) Scenario A: General, (b) Scenario B: Residential, and (c) Scenario
C: Industrial
Figure 6-14. Restoration curves for the natural gas network with HDPE distribution pipes in case
of magnitude 7.9 earthquake: (a) Scenario A: General, (b) Scenario B: Residential, and (c)
Scenario C: Industrial
Figure 6-15. Functionality of the natural gas network under the magnitude 7.9 earthquake in case
of using five crews
Figure 6-16. Functionality of the natural gas network under the magnitude 7.9 earthquake in case
of using one crews
Figure 6-17. Functionality of the natural gas network under the magnitude 7.9 earthquake in case
of using 25 crews

Figure 6-18. Functionality of the natural gas network under the magnitude 6.5 earthquake in case
of using one crews
Figure 6-19. Functionality of the natural gas network under the magnitude 6.5 earthquake in case
of using five crews
Figure 6-20. Functionality of the natural gas network under the magnitude 6.5 earthquake in case
of using 25 crews
Figure 6-21. Restoration curves for the natural gas network under the magnitude 7.9 earthquake in
case of using 25 crews
Figure 6-22. Resilience loss for the natural gas network following a magnitude 6.5 earthquake in
using five crews
Figure 7-1. Mapping from decision to objective space
Figure 7-2. Crowding-distance calculation
Figure 7-3. NSGA-II procedure
Figure 7-4. Simulations at day 60, for the case of an earthquake with a return period of 475 years,
a retrofit budget of \$2.8 million dollars, and two available repair crews
Figure 7-5. Pareto optimal solutions at day 60, considering an earthquake with a return period of
475 years, a retrofit budget of \$2.8 million dollars, and two available repair crews 105
Figure 7-6. Retrofit scheme for Policy 1 (green lines indicate retrofitted pipelines) 106
Figure 7-7. Retrofit scheme for Policy 2, 4, and 5 (green lines indicate retrofitted pipelines) 106
Figure 7-8. Retrofit scheme for Policy 3 (green lines indicate retrofitted pipelines) 107
Figure 7-9. Retrofit scheme that results in 100% functionality in both sectors after 90 days for the
considered scenario

CHAPTER 1: INTRODUCTION

Community resilience in the aftermath of a disaster is determined based upon the functionality of complex, interdependent infrastructure systems – physical, social, and economic. These infrastructure systems - including transportation, telecommunication, electric power, water, and natural gas can be quite vulnerable to natural hazards and their disruption can disturb the social welfare and economic vitality of a community. As an example, Gordon et al. (1998) estimated that in the 1994 Northridge earthquake in Los Angeles, approximately one-quarter of business disruption losses were due to the failure of highway bridges. Therefore, in order for a community to be resilient, interdependent lifelines should remain functional, or return to functionality rapidly after a disruptive event. When lifeline network functionality is threatened by a natural hazard, community resilience which can be defined as "the ability to prepare for and adapt to changing conditions and withstand and recover rapidly from disruptions" (PPD-21, 2013) is called into action (Guidotti et al., 2016). The word resilience is rooted from the Latin word "resiliere" which means "to bounce back". The concept of resilience has evolved and has been used in different disciplines such as ecology, material science, psychology, economics, and engineering. The concept of resilience in the engineering domain is quite new in comparison to other domains (Hosseini et al., 2016). Several researchers have provided definitions for resilience in the engineering domain. Youn et al. (2011) defined engineering resilience as the sum of the passive survival rate (reliability) and proactive survival rate (restoration) of the system. Hollnagel et al. (2006) described engineering resilience as the intrinsic ability of a system to adjust its functionality in the presence of a disturbance and unpredicted changes. The American Society of Mechanical Engineers (ASME, 2009) defined resilience as the ability of a system to sustain external and

internal disruptions without discontinuity of performing the system's function, or if the function is discontinued, to fully and rapidly recover the function.

In terms of community resilience, Koliou et al. (2018) reviewed the previous studies focusing on the effects of natural hazards on the built environment as well as social and economic systems within a community. They also discussed research needs in order to improve resilience assessment methodologies at the community level. They stated that one example of these gaps is including dependencies and interdependencies between the networks and components of a community in order to develop a methodology for community resilience assessment that properly correlates physical, social, and economic systems. Rosowsky (2019) explored the definitions of resilience in the context of civil engineering systems subjected to natural hazards. He considered a range of issues one should consider in articulating resiliency requirements for infrastructures systems, such as prioritization, triage, sequencing, and the ability to make emergency supplemental investments. He also discussed many of these dimensions and how associated decisions have political, social and economic dimensions.

Bruneau et al. (2003) defined a four-dimensional framework for community resilience, which is known as the resilience triangle model. The four dimensions they consider in their definition include: (i) Robustness, which is defined as the strength of a system to withstand any disruptive event in order to prevent losing its functionality, (ii) rapidity, which shows the speed or rate at which a system can return to its original state in order to meet an acceptable level of functionality, (iii) resourcefulness, which represents the ability to recognize and manage material (i.e., physical, technological, and information) and human resources in a case of emergency or disaster, in order to achieve predefined goals, and (iv) redundancy, which represents the extent of substitutable components or systems that can be used in the case of a disaster in order to minimize the likelihood and impact of disruptions.

Chang and Shinozuka (2004) proposed resilience measures that relate expected losses in future disasters to a community's seismic performance objectives. They used the four dimensions of community resilience previously proposed by Bruneau et al. (2003), which consists of technical, organizational, social, and economic dimensions. They demonstrated their proposed measurement framework by evaluating seismic resilience of the water delivery system in Memphis, Tennessee, comparing two retrofit strategies. They state that their resilience framework can be valuable for guiding mitigation and preparedness efforts. Miles and Chang (2003, 2004 and 2006) provided a conceptual model for disaster recovery in order to study community resilience. The basic notion of the model is based upon imperative relationships between a community's households, businesses, infrastructures, as well as its neighborhoods. They proposed five primary interrelated factors affecting recovery: (1) time, (2) space, (3) agents attributes, (4) interactions, and (5) policy. In this concept, spatial effects represent the importance of the topology of community components; meanwhile temporal effects illustrated the changes in the condition of a system during time. For example, households and businesses can be influenced by availability of water, power, natural gas, transportation, and local employment opportunities, which can have different conditions (i.e. performance/functionality) at different locations of the community during various timeframes. On the other hand, one example for the agent-attribute effects is the type of demand for businesses. The local transportation conditions and the recovery of households in neighborhoods are variables that can affect businesses with local demands, but they do not have a significant influence on businesses that export their products. Interaction effects represent the dependencies, crossdependencies, and interdependencies across community components and networks. For example,

the pumping stations in the water supply systems require electricity in order to be functional; or any disruption in the transportation system can delay repair of damaged components in, for example, the natural gas network. Households affect businesses in that not only do they provide employees for businesses but they are also consumers of their products. On the other hand, businesses affect households as they satisfy the households' demands. Finally, policy effects originate from the organizational decisions such as mitigation strategies and emergency planning before the event or restoration and resource prioritization in the aftermath of the disaster.

In terms of the infrastructure resilience, the National Infrastructure Advisory Council (PPD-21, 2013) defined the resilience of infrastructure systems as their ability to predict, absorb, adapt, and/or quickly recover from disruptive events such as natural disasters. It is noteworthy that infrastructure systems are considered a subdomain of both engineering and social domains as their lack of resilience can result in adverse impacts on well-being of communities (Hosseini et al., 2016). Moreover, infrastructure systems can have a significant influence on the economy of communities (Percoco, 2004). Since infrastructure functionality has a crucial role on social welfare and economic vitality of communities, a lot of research has recently focused on infrastructure systems' resilience. Shafieezadeh and Ivey Burden (2014) proposed a probabilistic framework for scenario-based resilience assessment of infrastructure systems. The developed method accounts for uncertainties in the process including the correlation of earthquake intensity measures, fragility assessment of structural components, estimation of repair requirements, the repair process, as well as the service demands. They applied the proposed methodology to a hypothetical seaport terminal and evaluated the system level performance of the seaport using various performance metrics. The results showed that medium to large seismic events may significantly disrupt the operation of seaports immediately after the event and that the recovery process may take months. They state

that their proposed framework enables port stakeholders to systematically assess the most-likely performance of the system during expected future earthquake events. Guidotti et al. (2019) proposed probabilistic flow-based models for estimating the capacity and demand of critical infrastructure considering different levels of resolution. The model set in a probabilistic procedure that considered the impact of the disruptive event on the infrastructure. The procedure was capable of predicting the reduction or loss of functionality of the infrastructure in terms of their ability to provide essential goods or services. Physical infrastructure and social systems were integrated in the models in order to predict the change in demand on the infrastructure. The proposed methodology was applied to the modeling of the portable water network for the small coastal community of Seaside, Oregon considering a seismic event as scenario. The results showed that neglecting the interdependency between the physical infrastructure and social systems may result in estimates of higher demands on the physical systems, slower recovery time, as well as smaller impacts on society in terms of population dislocation.

Furthermore, interdependencies play a substantial role in the performance of infrastructure, and thus the resilience of communities. During normal situations, these interdependencies can be beneficial and help networks operate near their optimized design capacity (Applied Technology Council, 2016; Guidotti et al., 2016; Nan and Sansavini, 2015; Ouyang et al., 2015). In the case of a disaster, however, interdependencies can have a substantial contribution to cascading effects and taking them into account can substantially alter the recovery process. Therefore, it is crucial to study the resilience of lifeline systems within a community including their interdependencies. An emerging body of research on functionality assessment and restoration modeling of interdependent infrastructure is ongoing. Dueñas-Osorio et al. (2007) presented the topological characterization of two interdependent small-sized real networks and investigated the effect of the degree of

coupling between networks with a tunable parameter that drove networks from independence to complete interdependence. The results showed that detrimental responses are larger when interdependencies are included in the analyses. Zhang et al. (2016) examined the vulnerability of an interdependent power-water network system by taking into account the effect of cascading failures as well as the resulting load redistribution through the power network. They introduced a critical tolerance threshold in order to control the cascading failures in the power system and its propagation into the water network. Guidotti et al. (2016) modeled the direct effects of seismic events on response and functionality of a potable water distribution network and simulated the cascading effects of the damage to the electric power network on the potable water distribution network. The results quantified the loss of functionality and the delay in the recovery process due to dependency of the potable water network on the electric power network and emphasized the importance of considering dependencies between different networks in modeling the resilience of critical infrastructure. Dong and Frangopol (2017) studied an interdependent healthcare-bridge network subjected to seismic hazard considering the effects of extra travel and waiting time along with the structural damage to the hospitals and bridges. Argyroudis et al. (2015) examined the influence of interdependencies on performance of road networks for case of a seismic hazard. The interdependencies were modeled both through the network components in addition to other external networks. It was shown that interdependencies can substantially affect the performance of a road network, specifically at the urban level where building collapse has a significant impact on network functionality. Masoomi and van de Lindt (2018) modeled a water network, electric power network, school buildings, residential buildings, and businesses along with their relative spatial distribution and system interdependencies and investigated the restoration of a community in the aftermath of a tornado.

Resilience studies of natural gas distribution networks have been primarily focused on the performance evaluation and vulnerability assessment of natural gas pipelines in the case of seismic hazards (see e.g. Choo et al., 2007; Jeon and O'Rourke, 2005; Lanzano et al., 2014; Lanzano et al., 2013; O'Rourke et al., 2014; Omidvar and Kivi, 2016; Xie et al., 2013; Psyrras and Sextos, 2018; Tsinidis et al., 2019; Badida et al., 2019). A number of researchers have also focused on studying qualitative and quantitative methodologies for risk assessment of natural gas distribution networks (see e.g. Esposito et al., 2015; Han and Weng, 2011; Liu et al., 2018; Markowski and Mannan, 2009; Cimellaro et al. 2014; Praks et al., 2017; Urlainis et al., 2015; Yuhua and Datao, 2005; Zulfikar et al., 2016). Some researchers have also studied the interconnected natural gas network and electric power network systems. Ouyang and Wang (2015) developed a resilience assessment framework for interdependent systems with the focus on modeling and resilience contribution analysis of multi-systems' joint restoration processes. As an illustrative example, the methodology was used for evaluating the resilience of the interdependent power and gas systems in Houston, Texas under hurricane hazards, using five different types of joint restoration strategies. The results revealed that under limited restoration resources, a gas-aimed restoration strategy is the best for the gas system recovery. Chiang and Zavala (2016) developed a detailed optimal control model that captured spatiotemporal interactions between natural gas and electric transmission networks. They used the model to study flexibility and economic opportunities provided by coordination and applied the methodology to a large-scale case study in the Illinois system to show the coordination can enable the delivery of significantly larger amounts of natural gas to the power grid. Liu et al. (2017) proposed a framework that combined dynamic modeling and resilience analysis of interconnected critical infrastructure. Thy performed a resilience analysis on two interconnected critical infrastructures, a gas network and an electric power system, using

numerical calculation of the resilience conditions in terms of design, operation, and control parameters values for certain failure scenarios. They state the results can be valuable to inform the decision making process of critical infrastructures operators and other stakeholders. Portante et al. (2017) proposes a framework for interdependency modeling in order to evaluate cascading failures within critical infrastructure systems. The framework permitted the integration and automation of the assessment process, including threat and hazard identification and data acquisition, as well as estimation and projection of the impact zones. The model was also capable of simulating the initial effects on infrastructure assets resulting from an initiating disruptive event and evaluation of propagating effects within each infrastructure system. Two state-level case studies were used in order to illustrate the approach in simulating the propagation of disruptions between the natural gas and electric power systems.

It should be noted, however, most of the natural gas network research has been dedicated to region-wide performance of natural gas networks and the system performance at the community level is seldom studied. Moreover, the bi-directional interdependency between natural gas and electric power network systems has been rarely studied. Modeling the natural gas network at the community level and considering interdependencies in resilience assessment of natural gas networks is important for risk-informed decision making is needed. Furthermore, decision making relating to preparedness and recovery has a crucial impact on maintaining safety, social wellbeing, and economic vitality of a community in the aftermath of a disaster. Thus, an optimal decision-making framework that quantifies the effect of different mitigation and recovery strategies needs to be developed in order to enhance the resilience of the communities after disasters.

OVERVIEW OF DISSERTATION

8

In this dissertation, a comprehensive approach is presented for optimal decision making for resilience assessment, seismic risk mitigation, and recovery of natural gas networks, taking into account their interdependency with other systems in the community. In this regard, the natural gas and electric power networks of a virtual community known as Centerville (Ellingwood et al., 2016) were modeled with enough detail such that it enables resilience assessment of natural gas network at the community level. Moreover, probabilistic seismic hazard assessment is considered in this dissertation, as it is crucial in capturing uncertainties in decision-making processes. A brief overview of the remaining chapters of this dissertation is summarized below:

Chapter 2 describes modeling natural gas networks and their components at the community level. It discusses fragility and restoration curves as well as repair rate equations that are developed based on available empirical data and expert judgment. In order to represent the performance of network components under seismic hazards, the fragility curves and repair rate equations are extracted from literature.

Chapter 3 discusses the interdependency approach for natural gas network and other systems. Modeling the interdependency between natural gas network and other systems as well as how disruption in one network can result in failures in the other are discussed. In order to consider the effect of cascading failures, a bi-directional dependence has been considered between the natural gas and electric power network in this dissertation.

Chapter 4 presents modeling of the earthquake hazard. Simulating spatial damage over the community when subjected to a seismic hazard is discussed to be employed further in the restoration analysis explained in the next chapter. In this dissertation, earthquake simulations are done using two different approaches: scenario-based and hazard-based. For scenario-based simulations, earthquake hazards are being generated using a well-known attenuation model for the

considered scenarios. For hazard-based simulations, two earthquakes with return periods of 475 years and 2475 years are considered using probabilistic seismic hazard assessment, which is done based on the active faults existing around the community and the possible earthquakes they can generate. In both cases, the effects of liquefaction are also included.

Chapter 5 introduces a community restoration algorithm in order to model the recovery process of the affected natural gas network. The algorithm is capable of representing spatial and temporal depiction of the restoration process until the full restoration is achieved. The algorithm also captures dependencies and interdependencies across components and networks and includes the effect of cascading failures in the restoration analysis such that a component remains nonfunctional until all its supplier nodes are fully recovered and functional, even if the component itself is undamaged or has been already repaired. Mitigation strategies and optimized resource allocation processes are also discussed in this chapter in order to evaluate the effect of those strategies on pace of recovery in natural gas networks.

Chapter 6 represents the illustrative community model used in this dissertation. Indeed, to illustrate the methodology proposed, a virtual community known as Centerville is subjected to the considered earthquakes (both scenario- and hazard-based), in order to examine the restoration of communities in the aftermath of an earthquake. This chapter discusses the topology and structure of the natural gas as well as electric power networks and describes the bi-directional dependence considered across the networks. It should be noted that although only natural gas network and electric power network are considered in this dissertation, other critical infrastructure systems such as transportation, telecommunication, water, and emergency services can be modeled and integrated into the current model in future studies.

Chapter 7 introduces a platform for optimal decision making, resilience assessment, seismic risk mitigation, and recovery of natural gas networks, taking into account their interdependencies with other systems. An algorithm is developed in order to find the best locations for replacing the main steel distribution pipelines with earthquake-resistant HDPE pipelines under limited budgets. Focusing on residential and business sectors in the community as two main sectors affecting the social stability area, decisions can be made on what pipelines should be retrofitted in order to improve the speed of network recovery and achieve the preliminary goals of the community.

Finally, the overall summary and the conclusions, the anticipated contribution to the profession, as well as recommendations for future research studies are provided in Chapter 8 of this dissertation.

CHAPTER 2: COMPONENT MODELS FOR NATURAL GAS NETWORKS

Natural gas is a vital component and is a major source of the energy supply in worldwide. Natural gas has many applications, including in commercial applications, in residential sectors, in industry, and even in the transportation sector. According to the U.S. Energy Information Administration (EIA, 2017) energy form natural gas accounts for 29 percent of the total energy consumed in the United States, making it a key component of the nation's energy supply. Figure 2-1 shows the natural gas consumption by sector in the United States in 2017.



Figure 2-1: U.S. Natural Gas Consumption by Sector (source: EIA, 2017)

Like other lifelines, natural gas networks (NGN) are susceptible to physical damage during earthquakes. Indeed, seismic events have highlighted the vulnerability of NGNs and the consequences of NGNs' loss of functionality on community resilience (Esposito et al., 2013). Specifically, recent earthquakes have caused a lot of damage to NGNs. As an example, the 1971 San Fernando earthquake resulted in extensive damage to welded-steel transmission pipes. The 1923 Kanto Japan earthquake also caused over four thousand breaks to gas pipelines in the Tokyo region (Esposito et al., 2013). In addition to pipelines, damage to other aboveground facilities in NGNs have also been observed. Besides the lifeline disruption due to the physical damage, there are other possible consequences such as pollution of the waterways or onset of fires and explosion. The explosion subsequent to the 1971 San Fernando earthquake, which caused crater-like depressions in the streets, is a good example. Therefore, it is important to conduct accurate vulnerability assessment of NGN components in order to design communities that are more resilient.

NETWORK OVERVIEW AND DEFINITIONS OF COMPONENTS

A natural gas system consists of two main categories of components: 1) nodal elements (e.g. natural gas processing plants and city gate stations), and 2) link elements i.e. pipelines that connect the nodal components together. The main components of the NGN include the following:

1) Natural Gas Processing Plant (NGPP): These facilities serve as a critical component in natural gas networks and their disruption can significantly interrupt the whole network's functionality. The natural gas that is brought up from the underground to the wellhead contains hydrocarbons, carbon dioxide, hydrogen sulfide, and water together with many other impurities (Poe and Mokhatab, 2016). On the other hand, major transportation pipelines usually impose restrictions on the quality and make-up of the natural gas that is allowed into the pipelines. Therefore, the raw natural gas from wellhead must be processed in a safe way and with minimal environmental effect before it can be moved into the transmission pipelines. Although some of the required processing can be accomplished at or near the wellhead (field processing), the complete processing of natural gas is done at a processing plant that is usually located in a natural gas producing region. The objective of a NGPP is to separate natural gas, associated hydrocarbon liquids, acid gases, and water from a gas-producing well and condition these fluids for sale or disposal (Poe and Mokhatab, 2016). After the liquid components and impurities are removed from the natural

gas, the natural gas can be injected into transmission and distribution systems in order to be used as fuel by residential, commercial and industrial consumers (Kidnay et al., 2011). Figure 2-2 shows a typical NGPP. It should be noted that processing plants may be different sizes and can be significantly different in their feed and product.



Figure 2-2: Natural Gas Processing Plant (source: https://www.bicmagazine.com)

2) Compressor Stations (CS): Natural gas needs to be highly pressurized while travelling through the transmission network. In order to make sure that natural gas maintains its high pressure so it can be transported from transmission pipelines to distribution systems, there are compressor stations usually located every 60 to 160 km (40- to 100-miles) along the pipelines, which supply the gas with necessary amount of pressure to keep it flowing over long distances through the network (Folga, 2007). These stations compress the natural gas using either a turbine, motor, or an engine. In addition to compressing natural gas, some compressor stations contain some type of liquid separator and filters similar to the ones

used to dehydrate natural gas during its pressing in order to capture any liquids or undesirable particles from the natural gas in the pipeline (Folga, 2007). These facilities usually include one or more compressor units, auxiliary equipment for secondary functions such as power generation or cooling of discharge gas, as well as a Supervisory Control And Data Acquisition (SCADA) system (Pitilakis et al., 2014). A typical compressor station houses the gas turbine compressor package as well as the instrumentation controls and equipment required to monitor and operate the engines and compressors in a low-rise building that are can be made from steel or reinforced concrete. These stations are usually enclosed by a chain-link fence and most have some type of additional security equipment such as cameras and motion sensors (Folga, 2007). Figures 2-3 and 2-4 show photos of the exterior and interior of a NGN compressor station.



Figure 2-3: Exterior View of a Natural Gas Compressor Station (source: https://www.entecheng.com)



Figure 2-4: Interior View of a Natural Gas Compressor Station (source: <u>http://go.jereh.com</u>)

3) City Gate Stations (CG): These stations are the connection points between the high-pressure transmission and medium/low-pressure distribution networks in a natural gas system, and are sometimes called town border or tap stations (Folga, 2007). The main function of the city gate stations is to meter the natural gas and reduce its pressure in the transmission pipeline to the distribution system. Moreover, natural gas received at the city gate may not contain odorant (the compound that gives the natural gas its specific smell). Before the gas leaves the city gate station, odorant is added to the gas if it contains insufficient or no odorant (Folga, 2007). These stations are usually cased in one-story buildings which house regulators and mechanical equipment for gas pre-heating, gas odorizing, and control through a SCADA system (Pitilakis et al., 2014). Figure 2-5 shows a typical city gate station.



Figure 2-5: Natural Gas City Gate Station (source: https://www.fiorentini.com)

4) District Regulating Stations (DR): These are local stations in the distribution networks that are used to control the pressure of the gas flowing through the distribution pipelines. They have a pressure reduction facility in order to set the gas pressure to the required level for end users. These stations are usually buried or housed in a metallic kiosk (Esposito et al., 2015). Figure 2-6 shows a district regulating station.



Figure 2-6: Natural Gas Processing Plant (source: https://www.tecnogas.es)

In general, natural gas pipelines can be divided into three main groups based on the level of pressure they operate at:

- Supra-regional transmission pipelines: these pipelines operate at extremely high pressures and have large diameters (usually up to 120 cm (48 inches) in diameter). These pipelines can cover large areas.
- Regional transmission/distribution pipelines: these pipelines also operate at higher pressure levels and are mainly used to connect the transmission networks to distribution systems.
- Local distribution pipelines: these are smaller pipelines that usually operate in the mediumto low-pressure ranges through the distribution networks.

For pipelines, the material and connection types are determining factors in performance of the pipelines, since they govern the behavior and the potential failure modes of buried pipelines in an earthquake. Reports from the American Lifeline Alliance (ALA, 2001) and HAZUS-MH (FEMA, 2003) describe some of the most common types of materials and connections used in buried pipelines. It is noteworthy that, pipelines that are specifically designed for natural gas networks are usually made of relatively ductile materials such as steel (K Pitilakis et al., 2014). Another particular type of material is polyethylene (medium or high density, i.e. MDPE or HDPE), which is used in newer networks due to its high ductility.

COMPONENTS DAMAGE MODELS

In order to capture damage that occurs to natural gas network components, fragility curves can be used. A seismic fragility describes the performance of a component or a system subjected to earthquakes in probabilistic terms, and can be defined as a mathematical function that expresses the probability that a facility or a component exceeds some previously defined limit state, as a function of a certain intensity measure such as peak ground acceleration (Porter, 2015). Fragility models are used in order to estimate the risk of earthquake excitation acting on the components of different lifeline systems such as natural gas networks. The fragility of a structural component or system is usually expressed in the form of a lognormal cumulative distribution function (CDF) as shown in Equation (2-1):

$$F_r(x) = \Phi\left[\frac{\ln(x) - \lambda}{\xi}\right]$$
(2-1)

where x = specified intensity measure (e.g. PGA or SA); $\Phi[*]$ = standard normal cumulative distribution function; $\lambda = \text{logarithmic}$ median of the capacity; and $\xi = \text{logarithmic}$ standard deviation of the capacity. Fragility functions are obtained either by structural modeling of the seismic performance of components and systems (known as analytical approach; e.g. Ellingwood, 2001; Singhal and Kiremidjian, 2002; Shinozuka et al., 2002; Noh et al., 2015), or using statistical analysis of failures observed during past earthquakes (known as the empirical approach; e.g. O'Rourke and So, 2000; Gardoni et al., 2002; Basoz et al., 2003; Straub and Der Kiureghian, 2008; Noh et al., 2015). In the analytical method, simulations include an analytical model of the structure subject to a set of ground motions having varying intensities. This process is known as incremental dynamic analysis (for more information on this see Vamvatsikos and Allin Cornell, 2002). Using the selected ground motion intensity in each simulation, the corresponding structural response/damage of the structure is recorded and used for fragility development. In the empirical method, the observed damage to the individual structure is recorded in addition to the intensity measure in case a ground motion instrument is present near that structure. However, as it is unlikely that there will be a strong ground motion instrument near the structure, usually the ground motion is estimated using ground motion prediction models or from a ground motion map such as the ShakeMaps produced by the USGS. It should be noted that, another class of fragility functions are known as expert opinion or judgment-based fragilities. An expert opinion fragility function is created by polling one or more experts in the field with the asset class in question, where the experts judge or predict failure probability as a function of environmental excitation (Porter, 2015). Examples of judgement-based fragilities can be found in Applied Technology Council (ATC, 1985).

For aboveground facilities (e.g. compressor stations and city gate stations), seismic fragility curves are usually based upon peak ground acceleration (PGA) and are often developed based upon available empirical data as well as expert judgment. These facilities may lose their functionality due to the failure of various components in these facilities, which include (K Pitilakis et al., 2014):

- Building: the collapse of the structure of the facility that covers the equipment may disrupt the functionality of the equipment;
- Electrical/Mechanical components: damage to these miscellaneous components can also result in non-functionality, especially if these components are unanchored;
- Electric power supply: lack of external power due to the electric power network disruption or the disconnection of the power lines and the facility building can also result in the facility become non-functional.

Various sources in the literature represent fragility curves for the aboveground components (ALA, 2001; Berahman and Behnamfar, 2007; FEMA, 2003; O'Rourke and So, 2000; Iervolino et al., 2004). Note that if the damage to the facility is addressed by simply considering the seismic vulnerability of the building that shelters the equipment, then the corresponding fragility curves of that building can be used in order to represent the vulnerability of the whole facility. However, if the aim is to include the mechanical and electrical component behavior in the vulnerability assessment procedure, then overall fragility curves which represent the system fragility should be

utilized in order to capture the vulnerability of the facility. The overall fragility curves are usually generated using a fault-tree analysis considering redundancies and subcomponent behavior (FEMA, 2003). In such case, Boolean logic is implicitly used within the procedure of defining each particular damage state. For example, a minor damage to a compressor station - which can result in short-time malfunction of the facility, may be represented by either the loss of electrical power or a slight damage to the building that shelters the equipment.

Like many other underground components, buried pipelines are very sensitive to permanent ground deformation (resulting from different types of ground failure, i.e. lateral spreading, liquefaction and landslide) in addition to transient ground deformation (resulting from seismic wave propagation). There is abundant literature on fragility relationships for buried pipelines. The fragilities for these components may be represented based on different intensity measures such as PGA (Chen et al., 2002; Hamada, 1991; Isoyama et al., 2000), PGV (Maison et al., 1995; Maruyama and Yamazaki, 2010; O'Rourke and Ayala, 1993; O'Rourke and Jeon, 1999; O'Rourke et al., 2012; Pineda-Porras and Ordaz-Schroeder, 2004), PGD (Eidinger and Avila, 1999; O'Rourke et al., 2012; ALA, 2001). Some researchers have also represented the fragility for buried pipelines in terms MMI (macroseismic intensity) or PGS (peak ground strain) (Chen et al., 2002; O'Rourke and Deyoe, 2004; O'Rourke and Jeon, 1999). Nonetheless, representing fragility curves for buried pipelines as equations based on peak ground displacement (PGD) and peak ground velocity (PGV) are the most common (ALA, 2001; FEMA, 2003). These equations are referred to as RR equations. In essence, these semi-empirical equations represent the number of repairs per unit length of the pipeline. The equations are usually developed according to the number of repairs recorded by field crews (e.g. ALA, 2001; O'Rourke and Ayala, 1993) and do not clearly demonstrate all possible failure modes for buried pipelines subjected to ground deformations. The

probability of having n repairs in a pipe segment with length L can be evaluated using a Poisson probability distribution as in Equation (2-2):

$$P(N = n) = \frac{[(RR)(L)]^n}{n!} e^{-(RR)(L)}$$
(2-2)

However, it should be noted that the main source of damage to buried pipelines is usually the permanent ground deformation, as proved by previous earthquakes (K Pitilakis et al., 2014). During the 1971 San Fernando earthquake, the steel pipelines withstood significant ground shaking, but suffered severe damage due to abrupt vertical and lateral dislocation as a result of lateral spreading (EERI, 1986; O'Rourke, 1988). In the 1906 San Francisco earthquake, lateral spreading accounted for only around 5% of the built-up area affected by the earthquake; meanwhile, approximately 50% of the damage occurred to the pipelines within one block of these zones, which shows the high impact of ground failure on pipeline damage (O'Rourke and Liu, 1999). In the 1964 Niigata earthquake, liquefaction resulted in substantial damage to pipelines, where the average failure ratio for one of the pipeline systems was as high as 0.97 per km with all kinds of failure types such as pipe body breaks, weld breaks, and joint separation (K Pitilakis et al., 2014).

The pipelines used in natural gas network are usually continuous pipelines like weldedsteel pipes. These pipelines typically fail due to compressive strain that induces buckling of the pipe body, or warping and wrinkling of the pipe wall (ALA, 2001; K Pitilakis et al., 2014). Moreover, pipeline damage in the natural gas network are categorized as either leakages or breaks, with different *RR* equations. If ground failure is the primary reason for damage to the pipeline, it is reasonable to assume that the proportions of leaks and breaks are 0.2 and 0.8, respectively; in the meantime, if the major cause of damage is ground shaking, leaks and breaks proportions will be 0.8 and 0.2, respectively. Moreover, for each network component, HAZUS-MH also provides restoration curves that provide the conditional probability of restoration for a certain initial damage at a given time after failure, which is also adopted in this dissertation.
CHAPTER 3: APPROACHES FOR MODELLING INTERDEPENDENCE

As previously mentioned, interdependencies can have a substantial contribution in cascading failures in the case of a disaster, and it is crucial to study the resilience of lifeline systems within a community including their interdependencies. A number of studies have utilized different approaches in order to model interdependency within the community. Rinaldi et al. (2001) defined four main classes of interdependencies between infrastructures, including physical, cyber, geographic, and logical. In this context, a physical interdependency arises from a physical linkage between the networks and two infrastructure systems are considered physically interdependent if the state of functionality in one network is dependent on the material output(s) of the other. For cyber interdependency, an infrastructure has a cyber interdependency if its functionality is dependent on the information transmitted from another infrastructure. A geographic interdependence occurs when elements of multiple infrastructures are in close spatial proximity such that a local environmental event can result in change of functionality status for all of the infrastructures. Finally, two infrastructures are logically interdependent if the state of each network depends on the state of the other system via a mechanism that is not classified as physical, cyber, or geographic. As an example, human decisions and interventions play an important role in logical interdependencies.

Researchers have used the physical definition of interdependency in order to study the response of interconnected networks. Dueñas-Osorio et al. (2007b) investigated the effect of seismic disruptions on the performance of physically interdependent networks. Several degrees of interconnectedness were explored and interdependent network fragility curves were introduced in order to display the effect of different strengths of coupling. Based on the results, mitigation

24

actions were proposed and propagation of their effects were studied. Robert and Morabito (2008) presented a set of tools in terms of consequence curves and flexible cartographic representations that enabled the management of physical interdependencies among critical infrastructures. Based on the resources exchanged through the interdependent networks, these tools allowed the visualization of the evolution of domino effects in time and space, giving the infrastructure managers the potential to set up convenient preventive and protective measures in order to avoid their propagation. Poljanšek et al. (2012) developed a probabilistic reliability model in order to generate network fragility curves of spatially distributed interconnected network systems subjected to natural hazards. The developed methodology was applied to the interconnected European gas and electricity transmission networks such that the gas-fired power plants formed the physical connections between the two networks. Modeling the physical interdependencies, the propagation of failures in gas networks were identified using fragility curves and the effect of gas network failures on electricity network were captured. Wang et al. (2012) also developed a framework for vulnerability analysis of interdependent infrastructure systems. Considering the physical interdependency between the networks, the methodology was applied to the power and water systems of a major city in China and the results were used for ranking critical components for protection purposes. Using the available restoration curves from the 2011 Tohoku Earthquake in Japan, Cimellaro et al. (2014) proposed a method for calculation of regional resilience considering different degrees of physical interdependency between the infrastructures. Kong and Simonovic (2019) developed a methodology for multiple hazard spatiotemporal resilience assessment of infrastructure systems considering the physical interdependencies. They applied the methodology on a simplified two-layer interdependent infrastructure network as a case study, showing the physical interdependence among infrastructures further magnifies the impact on resilience value.

On the other hand, some researchers have studied geographical, cyber, and logical interdependence between the infrastructure systems (e.g. Amin, 2010; Healy and Palepu, 2003; Johansson and Hassel, 2010; Ramachandran et al., 2015). Although the classification used by Rinaldi et al. (2001) provides a basic foundation for modeling infrastructure interdependencies, the clear way of modeling interdependencies between the infrastructures is not discussed in their study, and the authors emphasize the need for further research on developing explicit ways to model interdependencies.

Several researchers have developed different approaches that are potentially capable of explicitly modeling interdependencies between the infrastructures. Ouyang (2014) classified these explicit approaches into six categories: empirical approaches, agent-based approaches, system dynamics based approaches, economic theory based approaches, network-based approaches, and other approaches. These are briefly explained below.

EMPIRICAL APPROACHES

In empirical approaches, the critical infrastructure systems' interdependencies are analyzed according to historical accident or disaster data and expert experience. Empirical approaches provide the opportunity to identify significant and frequent failure patterns (e.g. Chou and Tseng, 2010; McDaniels et al., 2007), quantify interdependency strength metrics to inform decision makers (e.g. Zio and Sansavini, 2011), and perform empirically based risk analyses (e.g. Utne et al., 2011).

AGENT-BASED APPROACHES

In agent-based approaches, it is assumed that the complex behavior of interdependent infrastructures emerge from many individual and relatively simple interactions of autonomous agents (Kaegi et al., 2009). In this context, most critical infrastructures' components can be

considered as agents, and each agent interacts with other agents and its environment based upon a set of rules that mimics the way a real counterpart of the same type would respond. Agent-based approaches have been widely used in order to model the interdependencies between infrastructures (e.g. Barton et al., 2004; Basu et al., 1998; North, 2001), and are capable of modeling behaviors of decision-makers, capturing all types of interdependencies, providing scenario-based what-if analysis, and can also be integrated with other modeling techniques in order to provide more comprehensive analysis (Ouyang, 2014).

SYSTEM DYNAMICS BASED APPROACHES

System dynamics-based approaches take a top-down method in order to analyze and model complex interdependent infrastructures. The key concepts that are used in these approaches are feedback, stock and flow. Feedback loops represent the connections between infrastructure components, including the direction of interdependencies. Stocks show different states of the system, and flow rates control the levels of these states over time. Using a causal-loop diagram that captures the causal influence among different variables, and a stock-and-flow diagram that describes the flow of information through the system, system dynamic approaches are capable of modeling the interdependencies between critical infrastructures (Brown et al., 2004; Stapelberg, 2008).

ECONOMIC THEORY BASED APPROACHES

Economic theory-based approaches generally work based on interactions between households and producers. Households provide labor and capital to producers in exchange for income. The producers, on the other hand, provide the households with goods and services by the use of not only that labor and capital, but also other types of materials and services. Generally, two types of economic theories exist in the literature that can be utilized in order to model interdependencies between critical infrastructures: the input-output (I-O) model and the computable general equilibrium (CGE) model. The I-O models represent the interdependencies between different sectors of a national economy or different regional economies (Dietzenbacher and Lahr, 2004). The fist I-O model was proposed by Leontief (1986), which was a static and linear model. Since then various researchers have developed more complicated I-O models (e.g. Haimes et al., 2005; Haimes and Jiang, 2001). CGE models are an extension of the I-O models and inherit the main features of I-O models, including the consideration of interdependencies between economic sectors, while they overcome most of their limitations, such as linear interdependencies among sectors, lack of consumers' and producers' behavioral responses to markets and prices subject to labor, as well as resource and capital constraints (Rose, 1995 and 2004; Ouyang, 2014). Extending the capacities of the I-O models, CGE models have been used by researchers in order to model interdependencies between infrastructures (e.g. Zhang and Peeta, 2011).

NETWORK-BASED APPROACHES

Interdependent critical infrastructures can also be modeled as networks, where components of the networks can be represented as nodes, and the physical and relational connections between these nodes can be represented by links. Network-based approaches model each infrastructure by networks and then describe the interdependencies between these networks by inter-links, providing intuitive representations along with detailed descriptions of their topologies and flow patterns if desired. Network-based approaches also provide the opportunity to evaluate response of critical infrastructures under different hazards by modeling the component failures at component level first, and then simulating the cascading failures within and across the infrastructures at system level. Depending on whether the modeling procedure includes particle flow of the infrastructures, network-based approaches can be broadly divided into topology-based and flow-based methods.

Topology-based methods

In order to investigate the performance of interdependent critical infrastructures under natural hazards, topology-based simulation methods can be used. A topology-based model, which has also been adopted in this dissertation, defines the interdependencies between the infrastructures based upon their topologies, with discrete states of functionality for each component (i.e. nodes and links), and generally with considering two states: functional or failed. Failure of a component might happen directly due to the failure of the component itself, or indirectly due to the failure of other components in the network that result in disconnections from its source nodes.

In this methodology, a number of different metrics can be defined in order to evaluate the performance of the networks, such as the number of normal or failed components (e.g. Johansson and Hassel, 2010), the inverse characteristic path length (e.g. Ouyang et al., 2009), the connectivity loss (e.g. Dueñas-Osorio et al., 2007a), as well as the redundancy ratio and cluster related metrics (e.g. Dueñas-Osorio et al., 2007b). Furthermore, by incorporating the level of service provided to the customers and system-level performance, some other metrics can be defined such as lost service hours (e.g. Johansson and Hassel, 2010) or the fraction of customers affected (e.g. Poljanšek et al., 2012). The topology-based methods mainly capture topological features of the interdependent infrastructures and can be used to provide suggestions on robustness improvements from the topological perspective.

Flow-based methods

The key difference between topology- and flow-based methods is that in flow-based methods, the flow and the amount of services provided by the critical infrastructures is also taken

into account. In this context, some capacities are defined for each component in the networks that should be produced, loaded and delivered to the end users. Flow-based models capture the flow characteristics of interdependent infrastructures, and are capable of providing realistic descriptions on their operation mechanisms. This type of methods can also identify critical infrastructures' components and provide emergency protection provisions against natural hazards.

INTERDEPENDENCE APPROACH FOR NGN AND OTHER SYSTEMS

As the world struggles to find new sources of energy, natural gas plays a key role in the foreseeable future. Natural gas is used in different sectors – including residential, commercial, industrial, and electrical power sectors. The residential sector uses natural gas for heating buildings, heating water, cooking, and also for drying clothes. The commercial sector uses natural gas to heat buildings, heat water, operate refrigeration and cooling equipment, dry clothes and provide outdoor lighting. The industrial sector uses natural gas as a fuel for process heating as well as for combined heat and power systems, and as a raw material to produce different products. Finally, the electric power sector uses natural gas to generate electricity.

In particular, both natural gas and electric power networks show strong interdependency which is well recognized (Portante et al., 2017). For example, EPNs with gas-fired power plants need natural gas to produce electricity. An interruption or pressure loss in gas network systems may lead to a loss of multiple gas-fired electric generators, which can substantially decrease the supplied power and endanger the power system security (Shahidehpour et al., 2005). EIA's January 2018 Short-Term Energy Outlook (STEO) estimates that natural gas will remain the primary source of electricity generation in the United States, for at least the next two years, and it is likely that the use of natural gas as a fuel source for electricity will continue to grow (<u>https://www.eia.gov</u>). On the other hand, electricity is essential for natural gas processing plants

and sometimes for other stations such as electric-driven compressor stations. Interruption of electric service to such facilities can have significant negative safety and financial consequences observable long after the electric power network has recovered (EAC, 2011).

It is worth mentioning that, the issues related to interdependencies between natural gas and electric power networks are not only limited to those described above, but these interdependencies can also have a negative impact on the economic and social systems within a community. For example, in severe weather situations, the natural gas and electricity demands may reach the highest point together, which in turn, can result in a spike in energy prices. In such situations, gas price hikes can push up the marginal cost of gas-fired generating units, which would directly transform into higher market prices for electricity (Shahidehpour et al., 2005). For example, in the 2000-2001 winter, gas and electricity markets in California reached their capacity limits simultaneously, and the result was a dramatic increase in energy costs (Pope, 2002).

Therefore, it is important to consider the interactions between the natural gas and electric power networks in restoration modeling and recovery of a community in the aftermath of a disaster. On the other hand, natural gas has a crucial role in maintaining continued occupancy as a disruption in the natural gas network may disturb the functionality of residential, commercial, and industrial sectors, and consequently result in population dislocation and outmigration, especially during cold weather.

Modeling the interdependencies between different networks can be accomplished via an adjacency matrix A. As described in Wallace et al. (2003), a general network system k may be defined by $n^{(k)}$ nodes and $m^{(k)}$ links connecting the nodes, while the links are considered to be bidirectional (i.e. not pointing in a specific direction). In the case of K interdependent systems, each network can be represented as a symmetric $n^{(k)} \times n^{(k)}$ adjacency matrix, namely $A^{(k)} =$

 $[a_{ij}^{(k)}], (i, j \in k; i, j = 1, ..., n^{(k)})$, where $a_{ij}^{(k)}$ is equal to 1 if there is a link (element) between nodes *i* and *j*, and is 0 otherwise. The *K* adjacency matrices representing each network can then be placed along the main diagonal of *A*, and the off-diagonal matrices are used to represent connections between the nodes of different networks. For instance, considering two network systems *p* and *q*, the $n^{(p)} \times n^{(q)}$ matrix $A^{(p,q)} = [a_{ij}^{(p,q)}]$ represents the connections between the nodes of the two networks, where $a_{ij}^{(p,q)}$ is equal to 1 if there is a link between node *i* of network *p* and node *j* of network *q*, and is 0 otherwise. The adjacency matrix *A* can be represented as follows:

$$A = \begin{bmatrix} A^{(1)} & \cdots & A^{(1,p)} & A^{(1,q)} & \cdots & A^{(1,K)} \\ \vdots & \ddots & \vdots & \vdots & \ddots & \vdots \\ A^{(p,1)} & \cdots & A^{(p)} & A^{(p,q)} & \cdots & A^{(p,K)} \\ A^{(q,1)} & \cdots & A^{(q,p)} & A^{(q)} & \cdots & A^{(q,K)} \\ \vdots & \ddots & \vdots & \vdots & \ddots & \vdots \\ A^{(K,1)} & \cdots & A^{(K,p)} & A^{(K,q)} & \cdots & A^{(K)} \end{bmatrix}$$
(3 - 1)

In essence, in the connectivity matrix A, if $a_{ij}^{(k)} = 1$, then a failure of node i in the network results in a certain failure of node j, assuming that there is no redundancy in the network and that is the only connection between the nodes. It should be noted that, however, the dependency between two nodes in a network may not be mutual (Guidotti et al., 2016).

CHAPTER 4: MODELLING EARTHQUAKE HAZARD

Earthquakes are one of the most devastating disasters and can result in considerable loss of life and damage to property. As an example, the 1906 San Francisco earthquake caused more than 3,000 deaths and left 225,000 people homeless (USGS, n.d.). In general, earthquakes were responsible for 1.87 million deaths in the twentieth century and resulted in an average of 2,052 fatalities per event between 1990 and 2010 (Doocy et al., 2013; Wisner et al., 2003). Events such as the 1994 Northridge and 1995 Kobe earthquakes are reminders that losses to a community are not limited to the immediate aftermath of an earthquake only. In fact, socioeconomic losses accumulate over, as the recovery process gets longer and more complex (Miles and Chang, 2006). On the other hand, despite the sustained efforts on "earthquake prediction", which can be defined as issuing a science-based alarm of an imminent damaging earthquake with enough accuracy and reliability in order to justify measures such as evacuations, there is still no scientifically plausible way to predict the occurrence of earthquakes (Mulargia et al., 2017).

Two commonly used approaches for seismic hazard analysis include deterministic and probabilistic methods (Gupta, 2007). Both approaches are commonly used in the field and one has different advantages over another, but the question of what method should be used in a particular situation depends on the basic notion of the problem to be solved. In particular, factors affecting this choice include the decision to be made (i.e. the goal of the hazard and risk assessment), seismic environment (i.e. seismicity level of the location), and the purpose of the assessment (i.e. whether one is evaluating the risk at a site, a multi-site, or a region) (McGuire, 2001).

In the deterministic approach, different intensity measures are estimated for a specific considered earthquake scenario (Krinitzsky, 2002). An earthquake scenario can be a representative

of a potential future earthquake by assuming a particular magnitude, location, and fault-rupture geometry. The shaking maps from that earthquake scenario can be estimated using ground motion prediction models. The deterministic approach can be used for generating realistic earthquakes that are considered likely to happen, or are a desirable measure to use in design and analysis of infrastructure. Further, these can be used to conduct training exercises, and can be used by local government, utility companies and other organizations for disaster preparation and emergency response planning. The USGS Earthquake Hazards Program has a product known as ShakeMap that provides near-real-time maps of ground motion and shaking intensity following significant earthquakes, and can also be used as a tool for scenario-based earthquake modeling. Although convenient and easy to model, deterministic approaches might not be able to provide a realistic picture of seismic hazard at a specific site, as there is a great deal of uncertainty about the location, magnitude, and resulting shaking intensity of future earthquakes.

Probabilistic seismic hazard assessment (PSHA) aims to account for the uncertainties in the occurrence and intensity of future earthquakes, essentially their location and magnitude, and combines these uncertainties in order to produce an explicit description of the distribution of future possible earthquakes that may occur at a specific site (Baker, 2008). The concept of PSHA was first introduced by (Cornell, 1968). Since then, various researchers have used the methodology proposed by Cornell (1968) as a basis to develop more advanced models for PSHA (e.g. Abrahamson, 2000; Akkar et al., 2018; Baker, 2005; Delavaud et al., 2012; Jayaram and Baker, 2010; Jibson et al., 2000; Lapajne et al., 2003; Refice and Capolongo, 2002; Shahi and Baker, 2011; Stein et al., 2003; Tothoug and Cornell, 2007; Visini et al., 2019). While performing a PSHA may add some complexity to the seismic hazard assessment procedure due to incorporation of

uncertainties into the calculations, the results will be much more reliable especially in engineering decision-making applications (Baker, 2008).

FORMULATION OF PSHA

The PSHA considers all possible earthquake events and resulting ground motions, in addition to their associated probabilities of occurrence. The final output of a PSHA is a full distribution of levels of ground motion intensities along with their associated rates of exceedance. These results can later be used to identify a ground motion intensity having a certain probability of exceedance. The steps for performing a PSHA at its most basic level are explained in the following subsections (Baker, 2008).

Identifying seismic sources

The first step in performing a PSHA is to identify all earthquake sources capable of producing damaging ground motions. In contrast to the deterministic approach, in PSHA the interest is in all earthquake sources capable of producing ground motions. Ideally, the sources are faults, which are typically planar surfaces identified using various means such as by observations of past earthquake locations and geological evidence. However, if individual faults are not identifiable due to lack of knowledge about all the faults, then earthquake sources may be represented as areas that encompass several faults and are associated with a specific geological structure such as uplifts, rifts, folds, or volcanos that localize the seismic activity within that area. On the other hand, if there is no information on either active faults or geological structures (this may happen in less seismically active areas), then another type of seismic source referred to as "tectonic province" can be used in practical applications (Gupta, 2007). In this case, the area can be represented as a large geographic area of diffused seismicity with no identifiable active faults or geological structures, and earthquakes can occur anywhere within the considered area.

Seismicity sources in a region are usually identified based upon geological, geophysical, geodetic and seismotectonic data (Gupta, 2007). Nonetheless, if the available data is not adequate in most cases, expert knowledge and judgment will play an important role in identifying seismic sources for the region of interest. Once all possible sources of seismicity are identified for the region, one can identify the distribution of magnitudes and source-to-site distances associated with potential earthquakes from each source.

Characterizing the distribution of magnitudes

After the source of damaging earthquakes are identified, the next step is to identify the distribution of magnitudes for the potential earthquakes (i.e. the rates at which earthquakes with various magnitudes are expected to occur). Different seismic sources are capable of producing earthquakes with various magnitudes. The annual rate of occurrence for these earthquakes can be defined using recurrence relationships. Gutenberg and Richter (1944) first studied the magnitudes of earthquakes and noted that in general the number of earthquakes in a region greater than a given size follows a particular distribution as shown in Equation (4-1):

$$\log_{10} \lambda_m = a - bm \tag{4-1}$$

where *a* and *b* are constants and λ_m represents the rate of earthquakes with magnitudes greater than *m*. In this formulation, constant *a* represents the overall rate of the earthquakes in a region, and constant *b* indicates the relative ration of small and large magnitudes. These constants can be estimated using statistical analysis of historical observations, with additional constraining data provided by other types of geological evidence (Hainzl et al., 2006; Baker, 2008).

Equation (4-1) can be used in order to compute the cumulative distribution function (CDF) for magnitudes of earthquakes that are larger than a minimum magnitude m_{min} . The minimum magnitude is used as a condition as earthquakes with magnitudes smaller than m_{min} may not have

any engineering importance (Baker, 2008). By defining $\beta = b \ln (10)$, the cumulative distribution function for magnitudes can be written as:

$$F_M(m) = p(M \le m \mid M > m_{min}) = 1 - 10^{-b(m - m_{min})}$$
(4-2)

In practice, it is necessary to consider an upper limit on earthquake magnitudes, m_{max} , due to the finite size of the source faults. If a maximum magnitude is determined, then equation (4-2) can be rewritten as follows:

$$F_M(m) = p(M \le m \mid m_{min} < M < m_{max}) = \frac{1 - 10^{-b(m - m_{min})}}{1 - 10^{-b(m_{max} - m_{min})}}$$
(4-3)

In PSHA equations, the continuous distribution of magnitudes can be converted into a discrete set of magnitudes. Later, the former equation should be used in order to calculate the probabilities of occurrence for different magnitudes, as follows:

$$P(M = m_j) = F_M(m_{j+1}) - F_M(m_j)$$
 (4-4)

where m_j shows the discrete set of magnitudes, ordered in the way $m_j < m_{j+1}$. So long as the discrete magnitudes are finely spaces, the approximation will be accurate enough (Baker, 2008). It should be noted that, other alternative recurrence relationships exist in the literature that may be used instead of Gutenberg-Richter model (e.g. Burroughs and Tebbens, 2002; Kagan, 1991, 2002b, 2002a; Kijko, 2004; Lomnitz-Adler and Lomnitz, 1979; Main, 1996; Main and Burton, 1984; Merz and Cornell, 1973; Schwartz and Coppersmith, 1984; Youngs and Coppersmith, 1985). Some of these models have a faster decay for larger magnitudes and propose that some faults have repeated occurrences of a characteristic earthquake with a reasonably consistent magnitude (Gupta, 2007; Baker, 2008). This characteristic magnitude occurs more often than predicted by the Gutenberg-Richter model (Baker, 2008).

Characterizing the distribution of source-to-site distances

The PSHA also accounts for the differing probabilities of observing earthquakes at different locations. For this purpose, after the sources and corresponding earthquake magnitudes are determined, the distribution of source-to-site distances should be identified. It is usually assumed that earthquakes can occur with equal probability at any location within the earthquake source. Based on this assumption, the distribution of source-to-site distances can be identified using only the geometry of the sources. It is important to mention that there are different definitions of distance that are commonly used in PSHA (e.g. distance to epicenter or hypocenter, distance to the closet point on the rupture surface, distance to the closest point on the surface projection of the rupture only, some definitions account for the depth of the rupture. The choice of distance definition will depend upon the required input to the ground motion prediction model. For example, in some models, it is required to assume ruptures occur over a plane rather than at a single point in space, and the definitions should account for the depth of the rupture (e.g. Campbell and Bozorgnia, 2014).

Predicting the resulting distribution of ground motion intensity

The next step involves using ground motion prediction models in order to predict the resulting distribution of desired ground motion intensity as a function of many parameters such as earthquake magnitude, distance, faulting mechanism, the near-surface site soil conditions, the potential presence of directivity effects, and so on. These prediction models are developed mainly by statistical regression analysis of thousands of observed ground motion intensities from dozens of previous earthquakes. For example, Campbell and Bozorgnia (2014) used 11,125 records from 245 earthquakes of $3.0 \le M < 5.5$ and 4,396 records from 77 earthquakes of $5.5 \le M < 7.9$, as a subset of the PEER NGA-West2 database (Ancheta et al., 2013 and 2014) updated to include

earthquakes that occurred through 2011. Figure 4-1 shows the distribution of the recordings with respect to moment magnitude and rupture distance.



Figure 4-1: Distribution of recordings of the NGA-West2 database with respect to moment magnitude and rupture distance (excerpted from Campbell and Bozorgnia (2014))

As shown in Figure 4-1, there is- significant scatter in observed ground motions, and therefore, the ground motion prediction models need to provide a probability distribution for intensities rather than simply identifying a single intensity. This is important as PSHA calculations need to account for the possibility of unlikely events including those with extreme intensities much larger than the predicted mean values (Baker, 2008; Bommer and Abrahamson, 2006). In order to describe this probability distribution, ground motion prediction models take a general form as shown in Equation (4-5):

$$\ln IM = \overline{\ln IM} (M, R, \theta) + \sigma(M, R, \theta) \cdot \varepsilon \qquad (4-5)$$

where $\ln IM$ shows the natural logarithm of the desired ground motion intensity measure, which is modeled as a random variable with a normal distribution, and $\overline{\ln IM}(M, R, \theta)$ and $\sigma(M, R, \theta)$ are the mean and standard deviation of $\ln IM$, respectively. These terms are functions of earthquake magnitude (*M*), distance (*R*), and other parameters referred to as θ . Moreover, ε is a normal random variable that captures the variability in ln *IM*. Campbell and Bozorgnia (2014) proposed the following predictive model for the natural logarithm of PGA, PGV, and PSA:

$$\ln Y = \begin{cases} \ln PGA; & PSA < PGA \text{ and } T < 0.25 \text{ sec} \\ f_{mag} + f_{dis} + f_{flt} + f_{hng} + f_{site} + f_{sed} + f_{hyp} + f_{dip} + f_{atn}; \text{ otherwise} \end{cases}$$
(4 - 6)

where Y is the IM of interest and the f-term represent the scaling of ground motion with respect to earthquake magnitude, geometric attenuation, style of faulting, hanging wall geometry, shallow site response, basin response, hypocentral depth, fault dip, and anelastic attenuation, respectively (Campbell and Bozorgnia, 2014). The reader is referred to Campbell and Bozorgnia (2013) for detailed explanations of the abovementioned terms used in Equation (4-6).

The probability of exceeding any *IM* level can be estimated using the standard normal cumulative distribution function, as follows:

$$P(IM > x \mid m, r) = \int_{x}^{\infty} f_{IM}(t)dt \qquad (4-7)$$

where $f_{IM}(t)$ is the probability density function of PGA, given *m* and *r*. The next step will be to combine all the information from previous steps in order to find the full distribution of levels of ground motion intensities along with their associated rates of exceedance.

Combining uncertainties using the total probability theorem

The final step in performing a PSHA is to combine all the uncertainties associated with the earthquake size, location, and ground motion intensity, using a calculation known as the total probability theorem. Having the probability distributions for distance and magnitude and assuming that the magnitudes and distances of events are independent, equation (4-7) can be written as:

$$P(IM > x) = \int_{m_{min}}^{m_{max}} \int_{0}^{r_{max}} P(IM > x \mid m, r) f_M(m) f_R(r) dr dm \qquad (4-8)$$

where P(IM > x | m, r) is calculated from the ground motion prediction model, $f_M(m)$ and $f_R(r)$ are the probability distribution functions for magnitude and distance. Equation (4-8) represents the probability of exceedance given an earthquake happens and does not include any information about how often earthquakes occur based on the considered sources. In order to calculate the rate of exceedance (instead of probability of exceedance), Equation (4-8) can be written as follows:

$$\lambda(IM > x) = \lambda(M > m_{min}) \int_{m_{min}}^{m_{max}} \int_{0}^{r_{max}} P(IM > x \mid m, r) f_M(m) f_R(r) dr dm \qquad (4-9)$$

where $\lambda(M > m_{min})$ shows the rate of occurrence of earthquakes bigger than m_{min} , and shows the rate of IM > x. Acknowledging the fact all seismic sources contribute to the rate of IM > x, Equation (4-9) will look like as follows:

$$\lambda(IM > x) = \sum_{i=1}^{n_{sources}} \lambda(M_i > m_{min}) \int_{m_{min}}^{m_{max}} \int_0^{r_{max}} P(IM > x \mid m, r) f_{M_i}(m) f_{R_i}(r) dr dm$$
(4-10)

where $n_{sources}$ shows the number of sources included in hazard analysis, and M_i and R_i represent the magnitude and distance distributions of source *i*. In practice, the distributions of magnitude and distance are discretized Equation (4-10) can be rewritten as follows:

$$\lambda(IM > x) = \sum_{i=1}^{n_{sources}} \lambda(M_i > m_{min}) \sum_{j=1}^{n_M} \sum_{k=1}^{n_R} P(IM > x \mid m_j, r_k) P(M_i = m_j) P(R_i = r_k)$$
(4-11)

where n_M and n_R show the range of possible M_i and R_i , respectively. The abovementioned equation is the most common formulation used in PSHA and integrates the knowledge about occurrence rates of earthquakes, possible magnitudes and source-to-site distances from those earthquakes, as well as distribution of ground shaking intensity due to the possible earthquakes. The result can be represented as a graph known as a hazard curve, which represents the rate of exceeding different *IM* levels at a certain site. Figure 4-2 represents the hazard curve in Los Angeles downtown, for an earthquake with 10% probability of exceedance in 50 years (Return Period = 475 years), which is generated using the USGS Unified Hazard Tool (USGS, 2017). The results derived from a PSHA can be useful for engineering decision-making purposes.



Figure 4-2: Hazard curve for Los Angeles downtown (generated by Unified Hazard Tool) PSHA DEAGGREGATION

Despite the advantage PSHA provides by accounting for all possible earthquake sources in an area when computing seismic hazards, the contribution of each earthquake scenario in the hazard is not immediately obvious after all scenarios are aggregated together in the PSHA calculations. Perhaps the bigger issue is that the spatial correlations of the intensities over a community or region are lost since hazard intensity collects contributions from different sources. Nonetheless, the question of which earthquake scenario is most likely to result in exceedance of a prescribed level of earthquake ground shaking at a site can be answered through a process known as deaggregation (Baker, 2008; Bazzurro and Cornell, 1999). Deaggregation of seismic hazard helps to understand the contribution of different possible earthquakes to the hazard at a point location, by partitioning the earthquake sources into bins of distance and magnitude. The conditional joint distribution of magnitudes and distances can be found using the following equation:

$$P(M = m, R = r \mid IM > x) = \frac{\lambda(IM > x, M = m, R = r)}{\lambda(IM > x)}$$
(4 - 12)

The numerator of the above equation can be computed using the basic PSHA equation but not summing over the magnitude or distance, as follows:

$$\lambda(IM > x, M = m, R = r) = \sum_{i=1}^{n_{sources}} \lambda(M_i > m_{min}) P(IM > x \mid m_j, r_k) (M_i = m) P(R_i = r)$$

$$(4 - 13)$$

An example of this conditional distribution of *M* and *R* given IM > x is shown in Figure 4-3, for an earthquake with 10% probability of exceedance in 50 years (Return Period = 475 years) in Los Angeles downtown, which was generated using the USGS Unified Hazard Tool (USGS, 2017).



Figure 4-3: PSHA deaggregation for Los Angeles downtown (generated by Unified Hazard Tool)

The seismic hazard deaggregation is an essential part of many PSHA analyses and deaggregation results should be provided as part of the output from any PSHA calculations (Baker, 2008).

CAUSE OF DAMAGE TO NATURAL GAS COMPONENTS

Damage to the natural gas components from earthquakes occur due to either 1) ground shaking associated with traveling seismic waves, or 2) large permanent ground deformations. The former is often referred to as transient ground deformation (TGD), while the latter is generally known as permanent ground deformation (PGD). It should be noted that permanent ground deformations are usually the result of liquefaction, landslides or surface fault ruptures. The relative influence of different earthquake-induced PGDs on a natural gas network system may depend on different soil features such as morphologic, geological, and geotechnical conditions of the subsoil (Esposito et al., 2015). In general, fault displacement can be evaluated through semi-empirical equations that correlate displacement with earthquake magnitude (e.g. Petersen et al., 2011). Meanwhile, for liquefaction and landslide, the ground displacement can be estimated by using models that relate the ground displacement to a certain ground motion intensity measure. The approach of HAZUS-MH can be utilized for the purpose of risk assessment of large networked systems such as natural gas networks, as this methodology requires limited information about the geotechnical characterization of the region. In that methodology, the first stage for evaluating liquefaction hazard is to determine the liquefaction susceptibility, and then calculate the probability of liquefaction (i.e. the likelihood that the earthquake will initiate the phenomenon). Eventually, given that liquefaction occurs at certain locations, the amount of PGD (i.e., displacement) is calculated using predictive graphs. Similarly, the calculation of downslope displacement starts by first evaluating the landslide vulnerability, which is related to the morphologic and groundwater conditions of the area, in addition to the geologic source and the strength characteristics of soils. These factors contribute to the determination of the critical acceleration (which is known as k_c), which is defined as the minimum shaking intensity needed to overcome the slide resistance of the slope.

CHAPTER 5: RESTORATION ANALYSIS

Rapid recovery and restoration is a vital element of community resilience. Communities are composed of complex systems and highly coupled networks, and therefore, failure of one network or any of its key components can result in malfunction of other systems and loss of functionality of all or parts of the community. For example, the Rhode Island gas outage of 2019 resulted in a potentially dangerous loss of pressure caused by a supplier's faulty valve in a single natural gas pipeline, and left thousands of people without heat resulting in school closures (U.S. News, 2019). The Northeast blackout of 2003 that started from a single point source also affected approximately 50 million people in both U.S. and Canada (Farmer and Allen, 2006). These examples highlight that in addition to the direct losses such as injuries, fatalities, and property damage, indirect losses that are indeed the consequence of cascading failures need to be accurately addressed in community resilience assessment, as they play an important role in economic vitality and the social well-being of people. Various researchers have studied the damage to infrastructure systems and cascading failures resulting from damage in the aftermath of disasters (e.g. Argyroudis et al., 2015; Dong and Frangopol, 2017; Fang and Zio, 2019; Heracleous et al., 2017; Javanbakht and Mohagheghi, 2014; Shafieezadeh et al., 2014; Wu et al., 2016; Zanini et al., 2017; Zhang et al., 2016). Nonetheless, the duration of these failures within the community is of significant importance on the amount of indirect losses and can be quantified by modeling the restoration analysis of the community (Masoomi and van de Lindt, 2018a).

Restoration analysis in community resilience has received significant attention because of its critical role, but it is perhaps the least understood part of the recovery process for a community. Various researchers have modeled the post-disaster recovery of different infrastructure systems in order to estimate the expected restoration time for systems (e.g. Francis and Bekera, 2014; Guikema et al., 2014; Liu et al., 2007; Nateghi et al., 2011; Shinozuka et al., 2003). Some researchers have also compared different restoration strategies (e.g. Buzna et al., 2007; Cağnan et al., 2006). Ouyang et al. (2012) evaluated the performance of an electric power network under hurricane hazards and investigated the effect of the number of recovery resource units on the restoration process. Çagnan et al. (2004) categorized lifeline restoration models into two theoretical approaches known as Markov processes and network models, as well as two empirical approaches called statistical curve fitting and deterministic resource constraint. Moreover, they proposed a simulation-based model for post-disaster restoration process and called it the discrete event simulation (DES) method, which was built based upon past studies yet overcame some of their limitations. The method is capable of considering rules, constraints, and decisions from utility companies for restoration process including recovery prioritization plans, mutual aid agreements, as well as number of available repair crews and materials. The method also captures uncertainties associated with the parameters such as inspection time, repair time, and amount of available resources by defining these parameters as random variables with specific probability distributions. Cagnan and Davidson (2007) used the DES method to simulate the post-earthquake restoration process for Los Angeles Department of Water and Power (LADWP) electric power system and simulated the restoration curve for the Northridge earthquake. Çağnan et al. (2006) used the DES approach and investigated several restoration improvement strategies to improve the seismic resilience of the LADWP electric power system. Tabucchi et al. (2010) also used the DES approach to the water supply system of LADWP in order to simulate the restoration curve and spatial distribution of the restoration in the aftermath of Northridge earthquake. D'Uffizi et al. (2015) used the DES as decision support for planning different strategies of action to apply in

emergency and risk situations due to man-made threats and natural hazards. Huling and Miles (2015) used the DES for modeling home reconstruction in the aftermath of disasters and stated that the DES is a novel approach that can be utilized to support pre- and post-disaster decision making for improved community disaster resilience.

The recovery pace of a network system is a function of different parameters such as the recovery strategy and the amount and prioritization of resource allocation (Guidotti et al., 2016; Jia et al., 2017; Sharma et al., 2017). In order to evaluate the system functionality and recovery after the occurrence of an earthquake, a reliability approach should be used that includes both hazard modeling and physical and functional models of the damaged network system (Ameri and van de Lindt, 2018b). Moreover, the possibility of multiple component failures and the cascading effects due to different internal and external network dependencies and interdependencies should be taken into account for understanding the functionality assessment of a damaged network. In this dissertation, the DES approach is combined with network-based approach and is applied to a virtual community known as Centerville (Ellingwood et al., 2016). The methodology used enables considering the effect of cascading failures in the analysis such that if a component has no damage due to the disruptive event or is already physically repaired, it remains non-functional until all its supplier nodes are recovered and back to functionality. Furthermore, the methodology allows taking into account dependencies and interdependencies, as well as priorities for repair among networks and network components using evolutionary algorithm optimization techniques.

METHODOLOGY AND PROCEDURE

The methodology proposed herein combines the methods previously described with damage, functionality, and recovery models. Specifically a damage model refers to the simulation of physical damage to network components and a network functionality model represents an analysis used to investigate the damage impact on the network. Finally, a recovery model refers to the restoration process of the damaged network. The methodology is also capable of modeling network dependencies and interdependencies. Figure 5-1 depicts the flowchart of the methodology.



Figure 5-1: Flowchart for calculating functionality fragility and restoration process

The entire process can be summarized in six steps as follows:

1) Network modeling: The first step in resilience assessment of network systems is to model different networks components, including their internal and external dependencies and

interdependencies throughout the networks. This is worth-mentioning that different levels of resolution may be considered for modeling the networks based on the scale of interest and desired computational complexity.

- 2) Hazard modeling: The second step of the procedure is to create the hazard model with spatial variations in intensity for different components of the networks. Seismic hazard ground motion prediction models can be used for generating desired spatial maps of intensity measures. These include maps of peak ground acceleration (PGA), peak ground velocity (PGV), and peak ground displacement (PGD).
- 3) Direct damage assessment: Once the community topology and hazard are defined, the next step is to measure the physical damage that occurs to networks' components. This is evaluated probabilistically through fragility curves and *RR* equations for nodal and link elements, respectively. Fragility curves represent the conditional probability of exceeding a predetermined performance level for a given level of hazard intensity measure (e.g. Ditlevsen and Bjerager, 1986; Gardoni et al., 2003; Masoomi et al., 2017; Masoomi et al., 2018; Memari et al., 2017). *RR* equations, on the other hand, represent the expected number of leakage and breaks per unit length of the pipe, for a given hazard intensity measure (e.g. ALA, 2001; O'Rourke and Ayala, 1993; O'Rourke and Deyoe, 2004; Pitilakis et al., 2014).
- 4) Indirect damage assessment: In order to determine the damage states for each network system as a whole, the cascading effects due to internal dependencies of the network components as well as the interdependencies between different networks should be

included in the process. This step determines the damage status of the networks' components by including both the direct damage (from the previous step), in addition to the cascading effects due to dependencies and interdependencies.

- 5) Functionality assessment: In this step, natural gas network functionality is evaluated in the immediate aftermath of the earthquake. Functionality assessment of an NGN can be carried out at two different levels: a) connectivity-based analysis, and b) flow-based analysis. The former method concentrates on finding paths connecting source nodes to supplier nodes, and allows the serviceability assessment in terms of, for instance, the number of supplier nodes that remain accessible from at least one supply node after the occurrence of an earthquake. The flow-based method focuses on the network's capacity and considers extra constraints/factors in the serviceability evaluation; e.g., the minimum gas pressure at each supplier node.
- 6) Restoration analysis: The time duration that component failures remain within the community can have a dramatic effect on indirect losses, and can be estimated by performing a community restoration analysis. After determining the functionality status of the network following the seismic event, the next step is to evaluate the level of functionality at each time step t after the earthquake. The recovery time of a certain component in the network is a function of both its recovery time and that of the supporting elements through different networks. For each time step t, the method implemented herein updates the functionality status of the elements in the network based on the initial damage state and corresponding restoration functions. It should be noted that in the literature, there

are different examples of restoration functions that are either postulated (e.g. Bocchini et al., 2012; Titi et al., 2015), or obtained based on more fundamental recovery activities (Sharma et al., 2017).

In the restoration analysis in this dissertation, it is assumed that the recovery process of all networks are started immediately after the disaster occurs and simultaneously. This simultaneous assumption is felt to be logical as each network has its own crews for inspection, assessment, and repair.

MITIGATION STRATEGIES

Mitigation refers to measures that can be taken in order to minimize the destructive and disruptive effects of hazards, and thus reduce loss of life and property and lessen the impact of hazards in general. It includes some advance measures taken in order to avoid, decrease, or eliminate the risk that natural or man-made hazards impose to human lives and property. Mitigation strategies can be of different types, ranging from structural measures such as retrofitting the infrastructure components that are vulnerable to disasters, to non-structural measures such as developing and implementing education and outreach programs to increase public awareness. Although mitigation most often refers to actions against potential disasters before they happen, mitigation activities can take place at any time before the disaster, during an emergency, or after the disaster and during recovery and reconstruction (Maskrey, 1989).

The structural measures are important as immediate injury and damage results from the physical infrastructure systems failure (Godschalk, 2003). Structural measures focus on protecting people, property, and the environment from disruptive events in a number of different ways. Examples of structural measures include identifying hazards and risks in order to perform smart urban growth by directing new developments away from hazardous areas, retrofit and strengthen

vulnerable buildings and lifelines, as well as other activities such as flood control works, slope stabilization, and shoreline hardening.

Various researchers have focused on mitigation strategies for community resilience. Albert et al. (2004) studied the power grid from a network perspective and determined its ability to transfer power between generators and customers when certain nodes were disturbed. They realized that disturbances affecting key transmission substantially reduced its ability to function, and that increasing the redundancy and capacity of the existent structure can be used as an effective mitigation strategy to increase resilience and reduce blackouts effects on the population. Pitilakis et al. (2006) proposed a methodology for seismic risk assessment of utility systems and transportation infrastructures. Their proposed methodology provided a uniform basis for the reduction of the consequences of lifeline damage in urban areas and an efficient mitigation strategy and periodization for pre- and post-earthquake actions. Davis (2008) summarized practices implemented as well as other practices identified as needing to be implemented by operational water systems worldwide in order to improve their seismic resilience. He presented several mitigation strategies for reducing component damage and effects on system functionality, such as providing normal and backup seismic resistant power supply, implementing block distribution system, as well as providing system redundancy to expected damage areas. Bana e Costa et al. (2008) developed a multi-criteria value model that enabled the prioritization of bridges and tunnels according to their structural vulnerability and strategic importance as a mitigation strategy in the case of seismic events. The model was subsequently explored to prioritize the bridges and tunnels of a zone in Lisbon, Portugal with high seismicity. Kishawy and Gabbar (2010) summarized typical threats to pipeline integrity and techniques used in management and monitoring pipelines, and presented a pipeline integrity management system design for failure

prevention, inspection and repair, which allowed for risk mitigation and long-term optimized performance of pipeline systems. Zulfikar et al. (2016) described a real-time risk mitigation system for Istanbul natural gas distribution network. The system is capable of providing integrated ground shaking and damage maps for the natural gas network components immediately after an earthquake, which can be used for dispatching rapid response teams to high damage areas.

For the case of a natural gas network resilience against earthquakes specifically, using high-density polyethylene (HDPE) pipelines in the distribution network is one effective mitigation strategy, as the absence of damage in these pipelines in the Canterbury earthquake (O'Rourke et al., 2012) leads to a reasonable assumption of very good performance of HDPE pipelines in earthquakes. HDPE pipelines are a type of flexible plastic pipe that can carry portable water, wastewater, hazardous wastes, as well as oil and compressed gases (Sajwan et al., 2008). They are built from polyethylene, which is a very strong and durable, yet flexible material. Due to their lower life cycle costs (e.g. corrosion resistance and leak tight) and their reduced installation costs, HDPE pipes are being used more in new gas distribution networks as well as in replacement of aging steel pipes (Pitilakis et al., 2014). In this dissertation, the effect of replacing steel pipes with ductile HDPE pipelines has been investigated as part of an effective mitigation strategy against seismic hazard.

OPTIMIZED NETWORK RECOVERY

Resources are usually limited and scarce in the immediate aftermath of an earthquake. Therefore, developing optimized network recovery plans in order to effectively allocate available resources to damaged components of a disrupted network can speed up the restoration process, and hence, enhance the community resilience. Various studies have focused on optimized restoration of disrupted lifelines for a faster and more efficient recovery. Bryson et al. (2002) used a mixed integer programming approach for selecting a set of restoration plans, with the aim to give the greatest benefit to business operation and maximize the total value of the coverage provided by the set of selected plans. Casari and Wilkie (2005) discussed sequencing lifeline repairs after an earthquake and represented ways to integrate economic inventive into the management of natural disasters. The proposed decisional procedures could be employed by a planner in order to set repair priorities and help to coordinate lifeline firms in the post-earthquake reconstruction. Lee et al. (2007) proposed a model for minimizing the operating costs involved in temporary emergency restoration of highly interconnected lifelines. The model allowed representation of infrastructures under different conditions of normal operation, post-disruption impact assessment, as well as restoration, and permitted the development and use of algorithms for identifying solutions to problems associated with disruption of interdependent infrastructures. Xu et al. (2007) developed a methodology to determine how to schedule inspection, damage assessment, and repair tasks in order to optimize the post-earthquake restoration of the LADWP electric power system. The objective of the optimization was to minimize the average time each customer was without power, and a genetic algorithm was used to solve the problem. Matisziw et al. (2010) proposed a multi-objective optimization approach for network restoration during disaster recovery. The goal of the optimization was to ensure that facility restoration was prioritized so that system performance was maximized over a planning horizon. The proposed model permitted tradeoffs between two objectives to be evaluated and was applied to a telecommunication network to illustrate significance of optimization. Nurre et al. (2012) proposed an integrated network design and scheduling problem that modeled optimized restoring of infrastructure systems after an extreme event. They used integer programming to formulate the problem and utilized residual network optimality conditions in order to create a disputing rule and solve the problem. Applying

the models and algorithms on different testbeds, they indicated the developed dispatching rule can be utilized in real-time restoration planning activities. González et al. (2016) developed a model that optimized the reconstruction strategy of a damaged system of interdependent infrastructure networks, considering the existence of limited resources including budget, time, and manpower among others, as well as interdependencies between the networks, such as physical and geographical, among others. Maya Duque et al. (2016) developed a dynamic programming algorithm as well as an iterated greedy-randomized constructive procedure to solve the problem of network repair crew scheduling and routing problem during the aftermath of disasters. Vodák et al. (2018) introduced a modified ant colony optimization algorithm in order to increase the speed of the road network recovery process in the aftermath of disasters. Fang and Sansavini (2019) investigated the effects of uncertain repair time and resources on the post-disruption restoration of critical infrastructure. They proposed a stochastic programming for infrastructure restoration under uncertainty by developing a multi-mode component repair model, and proposed a tailored Benders decomposition to effectively solve the model.

In case of a disruptive event, utility companies start taking actions immediately after the disaster in order to recover their systems and its service. The number of available resource units for repairing the network is a crucial factor in this process. Meanwhile, the way in which these resources are being allocated to the damaged components in a network is another important factor that can substantially alter the entire recovery process. For a damaged network, different repair sequences can be defined based on the number of available resources, which can result in different restoration times and resilience. Accordingly, optimizing the restoration sequence to minimize the resilience loss is a good strategy to recover faster from a disaster.

Bruneau et al. (2003) proposed a deterministic static metric for measuring the resilience loss of a community in the aftermath of an earthquake as follows:

$$RL = \int_0^{t_1} [100 - Q(t)]dt \qquad (5-1)$$

where t_1 is the time when the infrastructure is completely recovered, and Q(t) represents the functionality of the infrastructure in percent. In this context, the performance of the damaged infrastructure system is compared to the target infrastructure performance during the recovery process. Resilience Loss (*RL*) can be illustrated simply as the shaded area in Figure 5-2. Accordingly, a larger *RL* value means a lower resilience while a smaller value of *RL* indicates higher resilience. Herein, this definition is used in order to measure community resilience in the aftermath of an earthquake.



Figure 5-2: Resilience loss measurement (excerpted from Bruneau et al. (2003))

It is usually difficult for standard optimization techniques to solve such problem due to the complicated cascading effects within and across multiple systems (Ouyang et al., 2015). Nonetheless, a Genetic Algorithm (GA) is a powerful optimization technique and has been successfully used in the literature for optimizing restoration sequences of damaged components (Xu et al., 2007; Ouyang and Wang, 2015). A Genetic Algorithm methodology has been used in

this dissertation in order to study the effect of optimized network recovery on recovery pace of natural gas network.

Genetic Algorithms, inspired by Darwin's theory of natural selection and initially developed by Holland (1975), is a well-known and accepted optimization method. In this method, a population of candidate solutions (aka individuals) is evolved to find a better solution for the problem. Each individual has a set of properties and is encoded as a string which represents a chromosome. The evolution begins by randomly generating a population of individuals, and it is a repetitive procedure, with the population at each iteration being called a generation. In each generation, a fitness value – which is usually the value of the objective function in the optimization problem being solved – is calculated for each individual in the population. Forming a new population takes place using three main operators: selection, crossover, and mutation. The selection operator chooses individuals based on their fitness values, meaning that the more fit an individual is, chances are higher they are selected. Two selected individuals (parents) then produce two descendants (offspring) by using the crossover and mutation operators. Crossover operator exchanges substrings of the chromosomes in the chosen individuals based on a crossover probability. Mutation operators alter the chromosomes of the offspring with a mutation probability. This process is repeated until enough offspring are available to create the next generation. The new population of candidates is then used in the next iteration to seek better solutions. It should be noted that, the algorithm usually terminates after a certain maximum number of generations has been reached, or after a satisfactory level of fitness is achieved. The procedure to optimize the restoration sequence for damaged components is shown in Figure 5-3.



Figure 5-3: Flowchart for fiinding the optimized restoration sequence

The procedure can be described in five steps as follows:

- Generate candidate solutions: The first step in this procedure is to number the damaged components in the network. Then, a restoration sequence may be expressed as a string composed of those numbers. For example, if five components are damaged in the network and the restoration sequence is in the order of 1 → 2 → 3 → 4 → 5, a chromosome such as 12345 represents that specific restoration sequence. It is obvious that in this context, the length of the chromosome is equal to the number of damaged components.
- 2) Calculate fitness values: The second step of the procedure is to evaluate the fitness value for each of the individuals built in the previous step. For this purpose, for each chromosome which corresponds to a restoration sequence, the restoration process is simulated and the *RL* is calculated. Then, a linear normalization technique is used in order to calculate the fitness values. In this formulation, after calculating the *RL* for each of the candidate
solutions, the *RL* values are arranged by the order starting from the smallest *RL* to the biggest one. Then, the fitness value for the *i*-th individual can be calculated as f(i) = max(a - bi, 1), in which *i* is the order of the chromosome, *a* is the maximum fitness value, and *b* is the reducing ratio (Ouyang and Wang, 2015). The values for *a* and *b* can be set to 100 and 2.5, respectively.

- 3) Selection: The roulette wheel selection techniques can be utilized in order to select superior chromosomes based on their fitness values. In this technique, a cumulative sum of all fitness values is calculated in order to make a summing fitness value sequence. Then, a uniformly distributed random number *r* is chosen between 0 and the total sum of fitness values. This random number is then compared with the elements of the summing sequence, and the first individual with its element in the summing sequence exceeding *r* is chosen.
- 4) Crossover: For a pair of selected individuals in the previous step, if a uniformly distributed random number is less than the crossover probability, then this operator is called into action. In this context, a number between 1 and the length of the chromosome is randomly selected as the cutting point. The first offspring inherits the longer substring from the first parent and replaces the genes of the shorter substring in the order they appear in the second parent. The second offspring inherits the longer substring from the first parent and replaces the genes of the shorter substring in the order they appear in the second parent. The second offspring inherits the longer substring from the first parent. For example, if 63281745 and 24876153 are the selected parents, assuming a cutting point 5 is considered, the first offspring inherits the genes 63281 from the first parent, and the remaining substring of 745 is replaced by 475, which is the order of genes, appeared in the

second parent. Hence, the first offspring will be 63281475 and the second offspring will be 24876315. This is shown in Figure 5-4.



Figure 5-4: Example of crossover operator application in optimizing repair sequence

 Mutation: For each offspring generated in the previous step, if a uniformly distributed random number is less than the mutation probability, then this operator is called into action. In this context, two randomly selected genes in the chromosome of an offspring are replaced with each other. For examples, if 63281745 is the selected offspring for mutation, assuming that the 3rd and 6th genes are selected to be replaced, then the mutated chromosome will be 63781245. This is shown in Figure 5-5.



Figure 5-5: Example of mutation operator application in optimizing repair sequence

Steps 3 to 5 are repeated until enough number of offspring individuals are made for the next generation. The whole process is repeated until a certain maximum number of generations has been reached. Finally, the chromosome with the minimum *RL* corresponds to the optimal restoration sequence for recovery of the network system.

CHAPTER 6: ILLUSTRATIVE COMMUNITY MODEL AND SIMULATIONS

To illustrate the application of the probabilistic procedure described before, the natural gas and electric power networks of a virtual community known as Centerville are considered and subjected to two different earthquake scenarios. The Centerville virtual community was developed as a testbed for the NIST-funded Center of Excellence for Community Resilience (van de Lindt et al., 2015) with the goal for examining procedures and methodologies (Ellingwood et al., 2016). Centerville is a typical mid-size community with a population of 50,000, located near an earthquake fault in the United States, and approximately 8 km by 13 km (5 miles by 8 miles) in size. The physical infrastructure of the Centerville built environment was initially modeled with four physical infrastructure components/systems – buildings, transportation system, water system, and electric power system. A natural gas network was later designed and added to the community (Ameri and van de Lindt, 2018a). The Centerville building portfolio contains about 30,000 buildings with residential, commercial, and industrial occupancies. It also includes critical facilities such as fire stations, hospitals, schools, and government offices. These are distributed in in seven residential zones (including a mobile home park), two commercial zones, and two industrial zones (one light and one heavy industry). All of the systems are essential to the health and welfare of a community, and play a significant role in the community resilience assessment, regardless of community size or its location. The schematic view of the Centerville is shown in Figure 6-1.



Figure 6-1. Plan of Centerville (after Ellingwood et al., 2016) [Inco. = Income; and Den. = Density]

THE CENTERVILLE NGN AND EPN

The natural gas network of the Centerville supports residential, commercial, and industrial sectors of the community through subsystems that include supply, transmission, and distribution (Ameri and van de Lindt, 2018a). These subsystems are modeled using graph theory with nodal elements (e.g. processing plant, city gate stations, etc.) and link elements (i.e. the transmission and distribution pipelines). The network is characterized by a natural gas processing plant, which is the major source of natural gas for Centerville, two compressor stations, two city gate stations, one local distribution company, 20 district regulating stations, and steel pipelines that connect these components together. The model of the natural gas system is illustrated in Figure 6-2. Among the 20 DR stations, one provides natural gas for the electric power processing plant, two stations deliver gas to the industrial sectors (heavy and light), and each of the other stations support two main sectors as well as their nearby critical facilities. It is noteworthy that even though the DR stations demand the natural gas from the transmission networks, herein they are considered

supplier nodes as they supply the natural gas for different sectors of the Centerville. The map of regions supported by each DR station is shown in Figure 6-3.



Figure 6-2. Centerville Natural Gas Network



Figure 6-3. Supported Zones by each District Regulating Station

Only large diameter transmission and distribution pipelines are represented in the model, as this simplified modeling is felt to be of sufficient resolution for assessing the system performance at the community level. Moreover, in designing the natural gas network system it was assumed that the natural gas system would be congruent to the major roadway system (Ameri and van de Lindt, 2018a). It is common practice in urban development in the U.S. to utilize the right-of-way for transportation systems to locate other distributed infrastructure systems (Ellingwood et al., 2016).

The electric power network of the Centerville includes one power plant, one transmission substation, one main grid substation, two distribution substations, three sub-distribution substations and 24 overhead poles (Unnikrishnan and van de Lindt, 2016) These are connected via transmission, distribution, and sub-distribution lines. The model of the electric power system is illustrated in Figure 6-4. As mentioned earlier, in case of a seismic event, damage occurs primarily to power plants, substations and buried power lines. Thus, in Centerville, the power plant and the substations are the only components of EPN that are prone to earthquakes.



Figure 6-4. Centerville Electric Power Network

ILLUSTRATIVE SCENARIOS

The methodology described in Section 2 was applied to Centerville in order to estimate damage and simulate recovery of the natural gas network when subjected to earthquakes. In order to evaluate the functionality and restoration of the natural gas network in the community, two earthquakes of magnitude 6.5 and 7.9, both located approximately 10 km northwest of the Centerville, were considered as two separate examples. Figure 6-5 to 6-8 show the PGA and PGV maps for each simulated earthquake from the attenuation relationship developed by Campbell and Bozorgnia (2014).



Figure 6-5. PGA map for the simulated 6.5 magnitude earthquake



Figure 6-6. PGV map for the simulated 6.5 magnitude earthquake



Figure 6-7. PGA map for the simulated 7.9 magnitude earthquake



Figure 6-8. PGV map for the simulated 7.9 magnitude earthquake

The simplified methodology explained in HAZUS-MH was adopted in order to simulate liquefaction effects. As was mentioned earlier, the first step for liquefaction hazard evaluation is to determine the relative liquefaction susceptibility of the soil and geologic conditions of the region. Based on the age, depositional environment, and soil properties in the area, a relative liquefaction susceptibility rating – that can range from very low to very high – can be assigned to each soil type in the region. The liquefaction susceptibility map assigned to Centerville is shown in Figure 6-9. Based on this map the PGD values can be determined for each location within Centerville for each hazard scenario.



Figure 6-9. Liquefaction susceptibility map for Centerville

The purpose of the natural gas network is to deliver gas from transmission systems to the distribution system, and then from there to the supplier nodes (i.e. DR stations), thus the network performance is evaluated under the simulated earthquakes. Two different scenarios are considered:

 Independent NGN: In this case, it is assumed the natural gas network is independent from the electric power network. In a connectivity-based analysis which is the scope of this dissertation, it is assumed the DR stations remain functional if they are properly connected to a source node. In this context, for nodal elements, it was assumed that the connectivity between the source and supplier node is lost if these components are in extensive damage stations according to the adopted fragility curves. 2) Interdependent NGN and EPN: In this case, the bi-directional interdependencies between the natural gas and electric power networks are considered. In this context, for the DR stations to remain functional, not only they need to be properly connected to the source nodes, but also the natural gas processing plant and the city gate stations need to be properly connected to and fed by the electric power network system.

The bi-directional interdependency between the NGN and EPN is represented in Figure 6-10. As shown in Figure 6-10, the required natural gas for electricity production is first produced in the natural gas processing plant (NGPP) and then delivered to the EPN power plant through NGN transmission pipelines. The NGPP also depends on EPN power plant for electricity. The city gate stations in the NGN depend on the nearby substations for electricity as shown in Figure 6-10.



Figure 6-10. Bi-directional interdependency between NGN and EPN

In the case of link elements it is assumed that a pipeline cannot supply gas to the DR stations if it has at least one break. The compressor stations and the local distribution company were removed from the model for fragility analysis as the natural gas can still flow through the network without these components being functional (although unquestionably a failure in the compressor station will result in lower pressures, but pressure alteration is not the scope of this dissertation). The susceptibility of DR stations were also neglected which is consistent with the work done by Esposito et al. (2015).

Accordingly, the NGPP, CG stations, and pipelines were considered as the vulnerable elements for the natural gas network. The CG stations were assumed to be modeled with the same fragility as of unanchored compressor stations (e.g. Chang and Song, 2007; Esposito et al., 2015). Moreover, for NGPP, the fragility curves for small oil refineries were adopted from HAZUS-MH in terms of lognormal cumulative distribution functions. For the electric power network, the power plant and the substations were considered as the vulnerable elements of the network, and the fragility curves were adopted from HAZUS-MH accordingly.

For buried pipelines, the *RR* equations were adopted from Pitilakis et al. (2014) for steel pipelines as functions of PGV and PGD as follows:

$$RR = K_1 \, 0.002416 \, PGV \tag{6-1}$$

$$RR = K_2 \ 2.5829 \ PGD^{0.319} \tag{6-2}$$

where RR is given in 1/km, PGV and PGD are expressed in cm/s and cm, respectively, and K_1 and K_2 are coefficients that are selected based on the pipe diameter and material. In order to calculate the RR for each pipeline, the pipe is first divided into separate segments. The RR for each segment is evaluated at the end nodes for both PGV and PGD intensity measures, and the mean RR is calculated by taking the average values at the end points of the pipe segments. Finally, the calculated RRs are multiplied by 0.2 and 0.8 for PGV and PGD, respectively, as only breaks are considered in a connectivity-based analysis, and the largest number is taken as the final RR. It is noteworthy that, the probability of having n repairs in a pipe segment with length L can be evaluated using a Poisson probability distribution as in Equation (6-3):

$$P(N = n) = \frac{[(RR)(L)]^n}{n!} e^{-(RR)(L)}$$
(6-3)

There are different indicators that can be used in a connectivity-based analysis for performance assessment of natural gas networks, among which is the serviceability ratio (*SR*) (see Esposito (2011) for other possible performance indicators and more details). The *SR* indicator, originally defined for water supply systems (see e.g. Adachi and Ellingwood, 2008), is associated with the number of supplier nodes in the distribution network that remain accessible from the supply facilities in the transmission network after the hazard occurrence. The *SR* may be calculated as:

$$SR = \sum_{i=1}^{n} (w_i \cdot x_i) / \sum_{i=1}^{n} w_i$$
 (6-4)

where w_i is the weighting factor for the supplier node *i*, x_i represents the functionality status of the supplier node *i* (which is modeled as the outcome of a Bernoulli trial, i.e., it is one if the supplier node is functional and it is zero otherwise), and *n* is the number of the supplier nodes. The weighting factor for each supplier node is calculated based on the amount of natural gas consumed by the buildings supported by that certain DR station (Ameri and van de Lindt, 2018a). For this purpose, the amount of natural gas consumption for each building archetype in Centerville was estimated based on the available data from the U.S. Energy Information Administration (EIA) (<u>https://www.eia.gov</u>). Table 6-1 shows the amount of natural gas consumption for each building archetype in Centerville.

Sector	Consumption (MCF*)			
Residential - R1	60			
Residential - R2	100			
Residential - R3	135			
Residential - R4	100			
Residential - R5	2100			
Mobile home	25			
Retail/Businesses	1075			
Government offices	1340			
School	2980			
Hospital	9420			
Fire Station	400			
Walmart/Home Depot	2500			
Light Industrial	5000			
Heavy Industrial	20000			

Table 6-1. Average annual natural gas consumption for each building archetype in Centerville

*MCF = the volume of 1,000 cubic feet

Moreover, two different networks are considered for each of the aforementioned scenarios:

- Normal networks: In this scenario the natural gas and electric power networks are modeled realistically, meaning that all of the components in the networks are reasonably susceptible to the earthquake hazard.
- 2) Fortified networks: In this scenario, the networks are considered to be more secured, in the way that the key elements in each network (i.e. natural gas processing plant in the NGN and electric power plant in EPN) are fortified, so that if they get damaged during the earthquakes they can be repaired much faster.

For each of the aforementioned scenarios, two different restoration strategies are considered: Random Restoration Strategy and Optimized Restoration Strategy. The different restoration strategies are being considered in order to evaluate how optimized resource allocation can speed up the recovery process. Moreover, in order to perform a thorough community resilience assessment, the effect of having different numbers of resource units on the restoration process are also taken into account. For natural gas network restoration, a number of available resource units, *r*, as generic work teams including repair crews, vehicles, equipment and replacement components are considered for assignment to damaged components (Ameri and van de Lindt, 2018a; Masoomi and van de Lindt, 2018; Ouyang et al., 2012). For electric power network restoration, two available resource units are considered, as this number seems reasonable according to the number of components in the EPN. It is assumed that only one resource unit is needed to repair each damaged component in the networks. When the repair is completed for a certain component, the resource unit is immediately shifted to the next damaged component in the network until the entire network is restored. Moreover, the effect of the industrial makeup of a community on its natural gas recovery following an earthquake, as well as the influence of replacing conventional steel pipes with high-density polyethylene (HDPE) ductile pipelines as part of the mitigation strategy plan has been examined for the case of an independent normal network with random restoration.

RESULTS

Effects of industrial makeup and HDPE pipelines

Considering the independent normal network under random restoration, in order to investigate the effect of the industrial makeup of a community on the natural gas recovery and the influence of replacing conventional steel pipes with high-density polyethylene (HDPE) ductile pipelines, three scenarios were considered in this dissertation for functionality assessment of the natural gas network of the Centerville:

Scenario A: General

In this scenario, the natural gas is realistically distributed between different buildings and sectors (i.e. residential, commercial, and industrial) of Centerville based on the number and size of buildings.

Scenario B: Residential

In this scenario, the natural gas is still realistically distributed among different buildings/sectors, however, the focus is on residential building (including mobile homes) restoration only. Therefore, the results in this simulation will represent the pace at which natural gas is restored to residential buildings as the measure being tracked for recovery.

Scenario C: Industrial

In this scenario it is assumed that the main portion of the natural gas (about 70%) is consumed by the industrial (both heavy and light) sectors of Centerville.

Furthermore, two different cases were considered for each of the abovementioned scenarios. First, it is assumed all the pipelines are made of steel and the fragilities are adopted accordingly. In the second case, the distribution network pipelines are assumed to be made of high-density polyethylene (HDPE) pipelines, which are being used more in new gas distribution networks due to their high ductility (K Pitilakis et al., 2014). There are no fragilities available for HDPE pipelines but the absence of damage in these pipelines in the Canterbury earthquake (O'Rourke et al., 2012) leads to a reasonable assumption of very good performance of HDPE pipelines in earthquakes. Therefore, in this dissertation it is assumed that HDPE pipelines in Centerville do not break in an earthquake due to their high ductility. Table 6-2 presents a summary of the 12 analyses presented in this dissertation for evaluating the effect of industrial makeup and HDPE pipelines.

Earthquake	Saanaria	Pipeline		
	Scenario	Steel	HDPE	
M _w = 6.5	Scenario A	1	4	
	Scenario B	2	5	
	Scenario C	3	6	
$M_w = 7.9$	Scenario A	7	10	
	Scenario B	8	11	
	Scenario C	9	12	

Table 6-2. Summary of analyses types and corresponding analysis number

For each scenario considered in Table 6-2, the intensity measures for simulated earthquakes were applied to each element of the natural gas network and their damage state and repair time were calculated using the methodology described in Chapter 5. The network damage and functionality status are then updated from the previous time step. The reader is reminded that even if a component has no damage or is physically repaired, it remains non-functional until all its supplier components are recovered and become functional, consistent with a standard network analysis. A Monte Carlo Simulation (MCS) was used to propagate uncertainties with simulations. In essence, the whole procedure described in Figure 5-1 in Chapter 5, is a single run within the MCS. In each run of the MCS, a uniformly random number between 0 and 1 is generated and used to sample from the statistical distributions for each component within the network models. Each run provides a probability of being in one of four damage states for each component. A repair time is then assigned to each damaged component based on its damage state and the demand percentage satisfied by the network in the immediate aftermath of the earthquake is calculated. Subsequently, a recovery sequence list is generated and the available resources (crews) are allocated to the damaged components in the generated recovery sequence list. After the first components in the list are repaired, the status of the repaired components is updated to functional, the new demand percentage satisfied by the network is calculated, and the associated resources are released and assigned to the next damaged components in the recovery sequence list. The process continues

until the system is fully recovered and the restoration curve can be determined. This whole process is repeated for 10,000 times and the mean result is eventually presented as the final answer.

The results for recovery of the natural gas network under the magnitude 6.5 earthquake are illustrated in Figures 6-11 (Analyses 1-3) and 6-12 (Analyses 4-6) for steel and HDPE pipeline cases considered. Figures 6-13 (Analyses 7-9) and 6-14 (Analyses 10-12) show the results for the magnitude 7.9 earthquake with r indicating the number of resource units in the analysis.



Figure 6-11. Restoration curves for the natural gas network with steel distribution pipes in case of magnitude 6.5 earthquake: (a) Scenario A: General, (b) Scenario B: Residential, and (c) Scenario C: Industrial



Figure 6-12. Restoration curves for the natural gas network with HDPE distribution pipes in case of magnitude 6.5 earthquake: (a) Scenario A: General, (b) Scenario B: Residential, and (c) Scenario C: Industrial



Figure 6-13. Restoration curves for the natural gas network with steel distribution pipes in case of magnitude 7.9 earthquake: (a) Scenario A: General, (b) Scenario B: Residential, and (c) Scenario C: Industrial



Figure 6-14. Restoration curves for the natural gas network with HDPE distribution pipes in case of magnitude 7.9 earthquake: (a) Scenario A: General, (b) Scenario B: Residential, and (c) Scenario C: Industrial

As shown in the figures, utilizing a different number of crews (resource units) was considered in the recovery process models. It is assumed that each crew for the natural gas network is capable of working on two damaged components simultaneously. For Scenario A in the 7.9 magnitude earthquake with steel pipelines (Analysis 7), for instance, the natural gas network will be 50% operational after 321 days (on average) if only one resource unit (r) is available. Obviously,

having only one work crew available after a major earthquake is unreasonable but is included with each scenario to serve as a benchmark. Increasing r from one to five significantly decreases the functionality time to 70 days for 50% functionality. It should be noted that increasing the number of crews beyond r = 10 does not considerably improve the recovery process. This is due to several factors. Namely, in the case where one of the damaged components has a higher repair time in comparison to the other network components. In the cases where the HDPE pipelines are used for distributing (Analysis 10) the natural gas network will be 50% operational after 64 days if only one resource unit is available. Increasing r from one to five decreases the functionality time to 21 days with HDPE. Meanwhile, increasing the number of crews beyond five does not have any influence on the recovery process if HDPE pipeline is used due to the same reasons explained earlier.

A basic resilience metric was able to be defined as the ratio of the area under the restoration curve to the area under the fully functioning curve corresponding to the fully functional system in line with Bruneau et al. (2003). Table 6-3 represents the approximate mean value of the resilience metrics for each scenario that are normalized to the maximum amount of resilience metric observed among all scenarios for each earthquake.

		Relative Resilience Index (%)					
Earthquake	Scenario	Steel Distribution Pipes			HDPE Distribution Pipes		
	-	r = 1	r = 5	r = 25	r = 1	r = 5	r = 25
M _w = 6.5	Scenario A	25	88	97	92	99	99
	Scenario B	25	87	97	91	98	98
	Scenario C	26	88	98	93	100	100
$M_{\rm w} = 7.9$	Scenario A	2	79	96	86	98	98
	Scenario B	1	78	94	85	97	97
	Scenario C	2	80	96	87	100	100

 Table 6-3. Resilience metrics for the studied scenarios

Note: the results for r = 2 and r = 10 are not shown in the table for brevity.

As shown in Table 6-3, and as expected, the most resilient system is the system with the HDPE pipelines for the distribution portion of the network. This is clear from the table that for such a system using more than five resource units does not have any influence on the recovery process as also noted earlier. However, for the system with steel pipes, increasing the number of crews can have a significant influence on the recovery process. Such a resilience index could be used to assist in community-level decisions regarding the replacement of steel NGN pipeline in high seismic regions. Each resilience index can be tied to a cost for a community, thereby supporting scenario-based decision.

Effects of bi-directional interdependencies and optimized network recovery

In order to investigate the effect of bi-directional interdependencies between natural gas and electric power networks on recovery rate of natural gas network system, as well as evaluating the effect of optimized resource allocation on recovery and restoration of natural gas network, several scenarios were considered. Table 6-4 presents a summary of the analyses performed for this purpose. In this context, "I" and "ID" stand for independent and inter-dependent scenarios, "N" and "F" stand for normal and fortified networks, and "R" and "O" stand for random and optimized restoration strategies, respectively. Also, "65" and "79" stand for 6.5 and 7.9 magnitude earthquakes, respectively.

Table 0-4. Summary of analyses types and corresponding analysis ins					
Sconorio	Network -	Restoration Strategy			
Scenario		Random	Optimized		
Indonandant	Normal	INR-65	INO-65		
maependent	Fortified	IFR-65	IFO-65		
Inter-	Normal	IDNR-65	IDNO-65		
dependent	Fortified	IDFR-65	IDFO-65		
Indonandant	Normal	INR-79	INO-79		
Independent	Fortified	IFR-79	IFO-79		
Inter-	Normal	IDNR-79	IDNO-79		
dependent	Fortified	IDFR-79	IDFO-79		
	Scenario Independent Inter- dependent Independent Inter- dependent	ScenarioNetworkIndependentNormalInter-NormaldependentFortifiedIndependentFortifiedIndependentFortifiedInter-NormalGependentFortifiedInter-NormalFortifiedFortifiedInter-NormalFortifiedFortified	ScenarioNetworkRestorati RandomIndependentNormalINR-65Inter-NormalIDNR-65dependentFortifiedIDFR-65IndependentFortifiedIDFR-65IndependentFortifiedINR-79IndependentFortifiedIFR-79Inter-NormalIDNR-79Inter-NormalIDNR-79dependentFortifiedIDFR-79		

 Table 6-4. Summary of analyses types and corresponding analysis IDs

Figure 6-15 shows the functionality of the natural gas network under the magnitude 7.9 earthquake, when five crews are utilized. In Figure 6-15, the vertical axis represents different scenarios under study, the horizontal axis shows the day (time), and the colors represent the functionality of the network according to the color bar at the top of the figure.



Figure 6-15. Functionality of the natural gas network under the magnitude 7.9 earthquake in case of using five crews

As shown in Figure 6-15, implementing an optimized recovery process for the natural gas network is beneficial in almost every single scenario considered. For example, in the case of an independent normal network, using an optimized recovery process will result in achieving around 90% functionality after 90 days, whereas the absence of an optimized recovery process reduces functionality to about 60% (compare INR and INO) in the same time horizon. In the case of an interdependent normal network, however, optimization does not have a significant influence on network recovery, as in this case interdependencies dominate the entire recovery process regardless of how the resources are being allocated to the damaged components (compare IDNR

and IDNO). Moreover, fortifying the networks has a significant advantage and accelerates the entire recovery process. This is especially important in the case of interdependent networks, as fortifying the key elements in the networks will result in more resilient systems. For example, in the case of using optimized restoration strategy for interdependent networks, a fortified network will be 95% recovered after 90 days; meanwhile a normal network will recover up to around 40% during that time frame (compare IDNO and IDFO). The results for other scenarios are shown in Figures 6-16 to 6-20.



Figure 6-16. Functionality of the natural gas network under the magnitude 7.9 earthquake in case of using one crews



Figure 6-17. Functionality of the natural gas network under the magnitude 7.9 earthquake in case of using 25 crews



Figure 6-18. Functionality of the natural gas network under the magnitude 6.5 earthquake in case of using one crews



Figure 6-19. Functionality of the natural gas network under the magnitude 6.5 earthquake in case of using five crews



Figure 6-20. Functionality of the natural gas network under the magnitude 6.5 earthquake in case of using 25 crews

As shown in these figures, using only one repair crew following a 7.9 magnitude earthquake is ineffective, regardless of the type of network (i.e. normal or fortified) and the method of restoration (i.e. random or optimized). Furthermore, in the case of using 25 repair crews, there is not much difference between random and optimized restoration methods; although in the case of the interdependent fortified networks, implementing an optimized restoration strategy can still be beneficial to some extent (compare IDFR and IDFO).

Figure 6-21 compares the different types of restoration curves for the natural gas network following a magnitude 7.9 earthquake, with 25 repair crews.



Figure 6-21. Restoration curves for the natural gas network under the magnitude 7.9 earthquake in case of using 25 crews

As shown in Figure 6-21, using an optimized recovery strategy does not have any influence on the recovery process for the independent NGN. Nevertheless, a fortified independent network has a better performance than a normal independent network as one would expect. In the case of interdependent networks, however, using an optimized recovery strategy always improves the recovery process. It should be noted that fortified interdependent networks have better performance when comparison to normal interdependent networks, even if a random restoration strategy is adopted for network recovery. In addition, in the case of interdependent normal networks, even though the systems eventually become functional at the same time (the graphs merge at a functionality of about 70%), the optimized restoration strategy still provides a better recovery curve and reduces the amount of resilience loss.

Finally, Figure 6-22 shows the data for resilience loss generated via Monte Carlo Simulations for the natural gas network under magnitude 6.5 earthquake while using five crews for recovery of the network. The unit for resilience loss is functionality-days



Figure 6-22. Resilience loss for the natural gas network following a magnitude 6.5 earthquake in using five crews

As can be seen from Figure 6-22, the average resilience loss is minimum for the independent fortified network when using an optimized restoration strategy; meanwhile, the average resilience loss is maximum for the interdependent normal network with a random

restoration strategy, which follows intuition but can now be quantified for the NGN. It is worth mentioning that the high variation in resilience loss for the interdependent normal network is due to the fact that for such systems, when both the natural gas processing plant and electric power plant are severely damaged during the earthquake, it can take a long time for the natural gas network to return to being functional. Of course, this means the resilience loss will be substantial. Meanwhile, in scenarios where the electric power plant - or generally the electric power network - is slightly damaged, then the natural gas network can function quite independently, meaning that the amount of resilience loss will be much lower and similar to that observed in the independent normal network.

The results provided here propose that considering bi-directional interdependency between the natural gas and electric power networks can alter the entire recovery process of the natural gas network in the aftermath of an earthquake. Fortifying the key elements in the natural gas and electric power networks can guarantee a much more resilient system and accelerate the recovery process, specifically when interdependencies are included in the analyses. Moreover, using an optimized network restoration strategy can speed up the recovery process of the natural gas network. Increasing the number of crews does not necessarily improve the recovery process post a certain number of crews, and in case of having sufficient number of repair crews, using an optimized recovery strategy does not necessarily enhance the recovery process. The failure of only one component in any of the natural gas or electric power network may impair the functionality of a large number of other components consistent with network analysis. Therefore, functionality cannot be used by itself in the decision-making process and should be considered in addition to the restoration process. The methodology proposed here can be utilized for weighing the cost of fortifying key elements in the networks or replacing steel distribution pipelines with HDPE compared with the summation of direct and indirect losses for different scenario events and decision making under uncertainty. The ability to quantify the effect of investment in improving the seismic performance of buried lifeline networks is key to deciding where these investments should be made for overall community resilience strategies. The methodology can be implemented to investigate the use of different decisions and policies by a community for a disaster scenario in order to prioritize disaster planning and mitigation, and for pursuing strategic decisions regarding whether to proactively retrofit or reactively repair steel pipelines in the distribution network of a NGN, with the objective of minimizing the network functionality loss, and therefore, disruptions in the society and economy. These will be further discussed in the next chapter.

CHAPTER 7: OPTIMAL DECISION MAKING UNDER UNCERTAINTY

In the context of risk management and community resilience, critical decisions should be made not only in the aftermath of a disaster in order to immediately respond to the event and properly repair the damage and restore services, but preventive decisions should be made in order to mitigate the adverse impacts of disasters before they happen. This usually involves significant uncertainty related to the basic notion of the hazard itself as well as disaster realization, and usually involves mitigation strategies such as strengthening components or preparing required resources for post-event activities. In essence, instances of risk management problems that encourage a framework for coupled decisions before and after disasters include how to allocate resources before the disruptive event so as to maximize the efficiency for their distribution in order to repair damage in the aftermath of the disaster, and how to determine which network components require preemptive investments in order to enhance their performance in case of a disaster (Gomez and Baker, 2019; Gomez et al., 2015).

A number of researchers have worked on addressing decision-making and optimization problems within risk assessment and management of complex infrastructure systems. Lim and Song (2012) proposed an efficient reliability assessment approach for complex infrastructure networks and introduced an algorithm called the selective recursive decomposition algorithm, which preferentially identified critical disjoint cut sets and link sets to calculate the probabilities of network disconnection events with a significantly reduced number of identified sets. Baroud et al. (2014) defined network resilience along dimensions of reliability, vulnerability, survivability, and recoverability, and proposed a framework to optimize network resilience strategies using a non-parametric stochastic ranking technique. Hu et al. (2015) developed a framework in order to determine optimal maintenance plans for large networks with many bridges, with the objective to minimize the extra travel distance caused by potential bridge failures over a planning horizon and under a budget constraint.

Frangopol and Bocchini (2011) proposed a methodology that used resilience as an optimization criterion for bridge rehabilitation, considering the maximization of the transportation network resilience in addition to minimizing the total rehabilitation cost, relying on bi-objective genetic algorithms for constructing an efficient Pareto front. Pereira et al. (2016) used multi-objective evolutionary algorithms to find resilient routing configurations that are robust to changes in traffic demands and are able to maintain desired performance in the case of arc failure events. Alderson et al. (2017) investigated the resilience of the transportation network in the San Francisco Bay Area based on a sequential game strategy and from the perspective of users. The model was used to quantify the operational resilience of the system, as well as for characterizing trade-offs in resilience considering different investments. Fan et al. (2009) proposed a stochastic programming model in order to optimize retrofit decision for highway systems so that damage caused by future earthquakes would be minimized. Liao et al. (2018) proposed a mathematical model to optimize transportation network resilience in case of a disaster. The model was used in order to prioritize preparedness and recovery activities under the constraints of budget and traversal time.

Peeta et al. (2010) addressed a pre-disaster planning problem that sought to strengthen a highway network whose links were subject to random failures due to a disaster. Considering the goal to select the links to invest in under a limited budget with the objective of maximizing the post-disaster connectivity and minimizing traversal costs between the origin and destination nodes which was modeled using a stochastic programming approach. Chang et al. (2012) proposed a methodology to find the optimal bridge retrofit program that aimed to maximize the post-disaster

90

network evacuation capacity, addressing the uncertainties of earthquake intensity, bridge structural damage, as well as bridge traffic-carrying capacities using a Monte Carlo simulation. Zhang et al. (2019) addressed the problem of evacuating residents out of affected areas in the aftermath of a natural disaster, and formulated a mathematical optimization model with the goal of maximizing the transportation network resilience and minimizing the total travel time for all users through different network reconfiguration schemes.

MULTI-OBJECTIVE OPTIMIZATION

Multi-objective optimization has a remarkable practical importance as almost every realworld optimization problem can be logically developed by considering multiple conflicting objectives. In the past, in order to solve a multi-objective optimization problem (MOOP), these problems used to be converted to a single-objective problem using weighting factors due to the lack of suitable solution techniques (Deb, 2014). The difficulty arises due to such problems giving rise to a set of trade-off optimal solutions known as Pareto-optimal solutions. Nevertheless, various methods have been proposed that are capable of solving such problems without the need to simplify and convert them to a single-objective problem (e.g. Deb et al., 2002; Fonseca and Fleming, 1993; Zitzler and Thiele, 1999).

In general, a multi-objective optimization problem can be defined as follows:

$$Optimize \ F(x) = (f_1(x), f_2(x), \dots, f_n(x)); \ x \in D$$
(7-1)

where *n* is the number of objectives $(n \ge 2)$, $x = (x_1, x_2, ..., x_k)$ is the vector of decision variables, *D* is the set of feasible solutions and F(x) is the function to be optimized. Each variable x_i in the feasible decision space can be mapped to a point in objective space defined as $F(x_i)$ as shown in Figure 7-1. It should be noted that although Figure 7-1 is in two-dimensional space to simplify visualization, in practice mapping usually takes place from a k-dimensional decision vector to an n-dimensional objective vector.



Figure 7-1. Mapping from decision to objective space

In contrast to a single-objective optimization problem, often there is no unique solution for a MOOP. Instead, there are usually a set of solutions representing the best possible trade-offs between the objectives. This set of solutions referred to as Pareto optimal solutions. Many researchers have developed algorithms to find the Pareto optimal solutions for MOOPs. As most MOOPs are mathematically complex (i.e. NP-hard), exact methods are usually effective for smallscale problems only. For medium- and large-size problems, it is necessary to use approximation methods such as metaheuristics. Multi-objective metaheuristics can be broadly classified into two categories:

• Scalar approaches: In this methodology, the MOOP is transformed into one or more singleobjective problems. These methods require prior knowledge of the problem in order to define preferences among objectives, and most of the time they produce a single solution per each run. The aggregation method (Ishibuchi and Murata, 1998) and ε -constraint method (Hertz et al., 1994) are examples of scalar approaches.

Population-based approaches: These methods use different candidate solutions in each run as a population that evolves in a way that guarantees some level of diversity among solutions. Among these methods, some are Pareto-based, in which the selection mechanism incorporates the concept of Pareto optimality. Some of the well-known methods in this category include: Pareto-archived evolution strategy (PAES) (Knowles and Corne, 1999), strength-Pareto evolutionary algorithm (SPEA) (Zitzler, 1999), and non-dominated sorting genetic algorithm (NSGA-II) (Deb et al., 2002).

It should be noted that a non-dominated sorting genetic algorithm (NSGA-II) approach has been used in this dissertation, and thus, this method is discussed in more details in the following sub-section.

Non-dominated Sorting Genetic Algorithm II (NSGA-II)

The non-dominated sorting genetic algorithm II (NSGA-II) proposed by Deb et al. (2002) is one of the most popular multi-objective optimization algorithms. With its three special and innovative features, i.e. a fast non-dominated sorting process, a fast crowded distance estimation procedure, and a simple crowded comparison operator, NSGA-II has been proved to outperform other contemporary multi-objective optimization techniques in terms of finding a more diverse set of optimal solutions (Yusoff et al., 2011; Deb et al., 2002). The aforementioned features are briefly explained in the following sub-sections.

Fast Non-dominated Sorting Approach

In order to sort a population into different non-domination levels in this methodology, at first two entities are calculated for each solution: 1) domination count n_p i.e. the number of solutions that dominate the solution p, and 2) S_p , a set of solutions that the solution p dominates. In this context, all solutions in the first non-dominated front will have their domination count as zero. For each solution p with $n_p = 0$, each member (q) of its set S_p is visited and the domination count for those members is reduced by one. In doing so, if the domination count becomes zero for any member q, that member is placed in a separate list Q, which represents the second nondominated front. This procedure is repeated on each member of Q in order to find the third nondominated front, and the process is continued until all fronts are identified.

Crowded Distance Estimation Procedure

Along with convergence to the Pareto-optimal solutions, it is also important to maintain a good spread of solutions in the Pareto front. For this purpose, a crowded comparison approach is used in NSGA-II. In this method, in order to estimate the density of solutions surrounding a specific solution in the population, the average distance of two points on either side of that solution is calculated along each of the objectives. This quantity, known as the crowding distance, can serve as an estimate of the perimeter of the cuboid formed by using the closest neighbors as the vertices. Figure 7-2 shows the crowding distance of the *i*-th solution in its front (marked in filled circles) as the average side length of the cuboid (showed as a dashed box). In order to compute the crowding distance for each solution, the population is first sorted in ascending order of magnitude according to each objective function value. Afterwards, for each objective function, the solutions with smallest and largest function values are assigned an infinite distance value. All other intermediate solutions are assigned a distance value equal to the absolute normalized difference in the values of two adjacent solutions. This calculation is then repeated for other objective functions and the overall crowding distance is reported as the sum of individual distance values corresponding to

each objective. The estimated crowding distance is then used in the crowded comparison operator that guides the selection procedure at various stages of NSGA-II algorithm.



Figure 7-2. Crowding-distance calculation

Crowded Comparison Operator

After all members in the solution population are assigned a crowding distance, the crowded comparison operator directs the selection of candidate solutions toward a uniformly spread-out Pareto optimal front. In this context, between two solutions selected from different fronts, the one with a better rank is desired; nonetheless, if the solutions have the same ranking (i.e. they belong to the same front), the solution with a bigger crowding distance is desired, as a bigger crowding distance represents a lesser crowded region.

NSGA-II Procedure

In this methodology, a random parent population P_0 with size N is initially created and sorted based on the non-domination ranking. Each random solution in the population P_0 is assigned
a ranking based upon its level of fitness. Using this initial population, a set of offspring population Q_0 is generated using the typical binary tournament selection, crossover, and mutation operators. The parent and offspring populations are then combined together in order to embrace the concept of elitism for generating next populations. From there, the process to generate next offspring populations using NSGA-II is shown in Figure 7-3.



Figure 7-3. NSGA-II procedure

As shown in Figure 7-3, in order to create a new population P_{k+1} (of size N), the parents and offspring from previous generations need to be combined first in order to ensure elitism ($R_k = P_k \cup Q_k$). The combined population R_k (of size 2N) is then sorted according to non-domination, and solutions that belong to the best non-dominated fronts are transferred to the new population P_{k+1} based on their rankings (i.e. the solutions are transferred first from set F1, then F2, and so on). Considering that the new population should have exactly N members, the process is continued until no more sets can be completely transferred to the new population. In this case, the members of the last set that cannot be completely transferred to the new population and will be sorted using the crowded-comparison operator, and then the best solutions from that last front will be selected in order to fill in the population P_{k+1} . The new population P_{k+1} will then be used for selection, crossover, and mutation in order to create another population Q_{k+1} . This process is usually repeated until a satisfactory level of non-dominated solutions are found for the specific multiobjective problem under study.

DECISION MAKING FRAMEWORK FOR DISASTER PREPAREDNESS AND RECOVERY

As described before, in order to improve the resilience of an infrastructure component or system, courses of action might be taken either before the event occurs (preparedness) or in the aftermath of the event (repair/recovery). Moreover, the actions that are taken in the aftermath of a disaster are often dependent upon the preceding actions that are taken beforehand. For example, a typical decision before an event occurs is to whether or not invest in strengthening a system's components in order to prevent or lessen the negative consequences of a potential future disaster, and how these investments are made will affect the repair and recovery process in the aftermath of the event. In this dissertation, an optimization framework that integrates probabilistic risk assessment of coupled infrastructure systems and evolutionary algorithms is proposed in order to determine cost-optimal decisions before and after a seismic event, with the objective of making the network recovery faster, and thus the community more resilient. Specifically, strategic decisions are pursued regarding which distribution pipelines should be retrofitted in the Centerville natural gas network under limited budgets, considering the objectives of minimizing the number of people without natural gas in the residential sector, as well as minimizing the business losses due to the lack of natural gas in non-residential sectors. MCS is used in order to propagate

uncertainties with simulations and PSHA is adopted in order to capture uncertainties in the seismic hazard. A non-dominated sorting genetic algorithm (NSGA-II) approach was utilized to solve the multi-objective optimization problem in the present study, and the results can provide risk-informed decision support as will be shown later.

Risk Assessment Methodology

The decision-making framework proposed in this dissertation adopts a methodology that combines probabilistic and deterministic seismic risk assessment methodologies. As mentioned in Chapter 4, the results from a PSHA are much more reliable due to incorporation of uncertainties into calculations, especially in engineering decision-making applications. Nevertheless, using the results of a PSHA merely may not be a suitable option for evaluating the serviceability of spatially distributed infrastructure networks, as the variation in spatial intensity in seismic demand from a particular earthquake is lost through the aggregation process (McGuire, 2001; Miller and Baker, 2015). In contrast to PSHA, a deterministic scenario-based hazard assessment has the ability to better capture the spatial correlation in seismic intensity, as it is based upon a single earthquake rather than an aggregation of earthquakes. Nonetheless, using a single earthquake scenario with a very low and unknown rate of occurrence cannot provide sufficient information for estimating losses due to a spectrum of possible future earthquakes, or for determining proper investments in retrofit of existing infrastructures (Adachi and Ellingwood, 2009). Therefore, implementing a method that includes desirable features of both PSHA and scenario-based seismic hazard assessment is required. For this purpose, the methodology developed by Adachi and Ellingwood (2009) is adopted in this dissertation. In this context, all possible earthquakes affecting the serviceability of an infrastructure at a stipulated seismic hazard should be included in the analyses.

For the case of the Centerville coupled NGN and EPN systems, it is assumed the seismicity in the Centerville is affected by earthquakes that can occur from two separate seismic source zones. One seismic zone is located with its center at a distance approximately 15 km from the northeast side of Centerville, and with the potential to generate magnitude 7.0 to 8.0 earthquakes that occur at a rate of $\lambda = 0.002$ times per year. The other seismic zone is located with its center at a distance approximately 10 km from the southwest of the Centerville, with potentially producing magnitudes 6.0 to 7.0 earthquakes that occur at a rate of $\lambda = 0.01$ times per year. Accordingly, in order to evaluate the serviceability of the interconnected NGN and EPN of the Centerville under a specified seismic hazard, both of the abovementioned seismic sources should be taken into account.

Two different return periods of 475 and 2475 years are considered in this dissertation. The first step in the process of seismic hazard assessment for Centerville - that combines both PSHA and scenario-based methodologies together, is to deaggregate the seismic hazard at each return period of interest, as explained in Chapter 4. Next, the performance of the networks under all possible earthquake scenarios is evaluated in order to capture the spatial variation in intensity values accurately. Finally, the components' functionality and the networks' serviceability are determined by re-aggregating the seismic damage, weighted by the relative contribution of each earthquake at each return period of interest.

Optimization Problem and Objective Functions

Considering the coupled NGN and EPN in Centerville, the goal is to minimize the adverse consequences due to the unavailability of natural gas in the case of an earthquake, by retrofitting the most vulnerable components of the natural gas network under limited budgets before the seismic event, and effectively allocate the limited resources available for repairing the damaged components of the network in the aftermath of the earthquake. Thus, acknowledging the limited budgets for retrofit and scarcity of available resources for repair, strategic decisions need to be made regarding how to proactively retrofit beforehand, and then reactively repair the damaged components afterwards, in order to maximize community resilience. For this purpose, two different objective functions are considered for minimization: 1) the number of people affected by unavailability of natural gas in the residential sectors, and 2) the amount of business losses due to unavailability of natural gas in the non-residential sectors. In order to calculate the business losses, the same concept of resilience loss proposed by Bruneau et al. (2003) is utilized in this dissertation, as described in Chapter 5. In this context, the business loss can be evaluated using the following equation:

$$BL = \int_0^{t_T} [100 - F_2(t)]dt \qquad (7-2)$$

where t_T is the target time i.e. the time until when the business losses are being calculated, and $F_2(t)$ represents the functionality of the non-residential sectors as a percent. This metric can be later transformed to a monetary measure, by evaluating the amount of money that is lost due to unavailability of natural gas in the non-residential sectors. For the case of Centerville, it is assumed that 100% unavailability of natural gas in non-residential zones can result in losses equivalent to \$1,000,000 per day. The reader is reminded that this amount is just an assumption made in this dissertation for illustrative purposes only, and one may require performing a detailed investigation of exact amount of monetary losses due to unavailability of natural gas in non-residential sectors. *Retrofit and Repair Strategy for the Centerville NGN*

As shown in Chapter 6, replacing steel pipes with HDPE pipelines in the natural gas distribution network can effectively increase the resilience of the system. Hence, retrofitting the steel pipes with HDPE pipelines can be regarded as an effective mitigation strategy. In order to find what combinations of retrofit and repair strategies result in the Pareto optimal solutions based on the defined objectives in this dissertation, a random population of retrofit parents (i.e. retrofit schemes) is first generated. Each retrofit parent is built such that steel pipelines are randomly selected from the distribution network to be replaced with HDPE pipes, until there is no money available for retrofitting more steel pipelines. Afterwards, for each retrofit parent, NSGA-II is implemented in order to find the best repair strategies that result in Pareto optimal solutions for that specific retrofit scheme. Considering the limited number of available repair crews, selection, crossover, and mutation operators are used as described earlier in Chapter 5 in order to find the best repair sequences for each retrofit scheme. After the best combinations of retrofit and repair strategies are found, a new population of retrofit parents needs to be generated. In this dissertation, in order to generate new retrofit offspring populations, only selection and mutation are used. The reason is that the retrofit optimization problem considered in this dissertation is not a sequential problem (i.e. the order of selected pipelines for retrofitting does not matter in the optimization process), and therefore, exchanging different segments of retrofit chromosomes does not necessarily result in a more diverse population. Moreover, there is also the danger of violating budget constraints in swapping different segments of selected retrofit chromosomes. Therefore, using only selection and mutation operators seems to be a better option for the specific problem defined herein. Therefore, in this dissertation, a mutation operator is designed with the aim to guarantee building a more diverse population for next generations. In this context, first a number (p) between 1 and half of the length of the selected chromosome is randomly selected. Then, p elements of the selected retrofit chromosome are randomly selected and substituted with other pipelines that do not already exist in the chromosome. Combining this mutation operator with the selection methodology used in NSGA-II algorithm, the retrofit parents for next generations can be built. Then, for each newly built retrofit parent, the procedure explained above is repeated again,

in order to find the best repair strategies for each retrofit parent. This whole process is repeated until enough number of non-dominated Pareto optimal solutions are found. It should be noted that NSGA-II has already been used by other researchers adopting a selection and mutation operator only (e.g. Coello et al., 2009). Moreover, it is proven that depending on the basic notion of the problem under study, a search strategy that is merely based upon selection and mutation can be used as a powerful search procedure (Michalewicz, 2013). It is also noteworthy that technically, an algorithm that only works based upon selection and mutation operators is no longer considered a genetic algorithm, but may be referred to as an evolution strategy or evolutionary programming (Simon, 2013).

Results

In order to show the results of the multi-objective optimizations, it is important to capture the bi-directional interdependencies between the NGN and EPN systems. Moreover, as the results from Chapter 6 proved the importance of functionality of key elements (i.e. natural gas processing plant in the NGN and electric power plant in EPN) in the resilience of the natural gas network, retrofitting these key elements should be prioritized in any mitigation plan. Thus, a fortified interdependent natural gas network is selected in order to show the optimization results.

In terms of the costs for replacing the pipelines, Farahmandfar et al. (2016) obtained the cost of pipeline replacement from the literature. Assuming that pipe bursting technique is used for pipeline replacements, they used a cost of \$1.04/mm/m (\$8 in./ft.) for pipeline replacements, which was derived by adjusting the costs provided by Boyce and Bried (1998) for inflation from 1998 to 2014. In this dissertation, the same cost of \$1.04/mm/m is used in order to calculate the costs of pipeline replacements. There are approximately 44 km of pipelines in the natural gas distribution network of the Centerville. In order to retrofit the natural gas network distribution pipelines, two

different budgets of \$2.8 and \$5.6 million dollars are considered for illustrative purposes, which are sufficient for retrofitting approximately 25% and 50% of the total pipelines in the distribution network, respectively. Moreover, three different number of repair crews (two, five, and ten) are considered in the recovery process of the natural gas network, in order to capture the effect of having a different number of available resources for repair in the aftermath of an earthquake. Considering two different earthquakes with return periods of 475 and 2475 years, three different target days of 30, 60, and 90 days have been selected. These are used as a target in time domain, at which the objective functions are being optimized.

As an example, consider a retrofit budget of \$2.8 million dollars and two repair crews available to repair the damaged components under an earthquake with 475 years return period. After 30 days, both residential and non-residential sectors remain completely non-functional, regardless of how the gas distribution network is retrofitted before the earthquake and how the damaged components are repaired afterwards. Nonetheless, after 60 days, even though the sectors cannot reach 100% functionality, there are certain combinations of retrofit and repair that can produce some optimal solutions. Figure 7-4 shows the process of moving toward optimal solutions via millions of simulations for the considered scenario. The Pareto front for these simulations is also plotted in Figure 7-5. As shown in Figure 7-5, each dot on the plot represents a certain policy (P) of retrofit and repair that results in Pareto optimal solutions. The retrofit schemes for these policies (i.e. P₁ to P₅) are depicted in Figures 7-6 to 7-8. The green lines in the figures show the pipelines that have been retrofitted. It is noteworthy that the retrofit schemes for Policies 2, 4, and 5 are similar, and the difference in the results is due to adopting different repair strategies in the aftermath of the earthquake.



Figure 7-4. Simulations at day 60, for the case of an earthquake with a return period of 475 years, a retrofit budget of \$2.8 million dollars, and two available repair crews



Figure 7-5. Pareto optimal solutions at day 60, considering an earthquake with a return period of 475 years, a retrofit budget of \$2.8 million dollars, and two available repair crews



Figure 7-6. Retrofit scheme for Policy 1 (green lines indicate retrofitted pipelines)







Figure 7-8. Retrofit scheme for Policy 3 (green lines indicate retrofitted pipelines)

It should be noted that after 90 days, there are certain combinations of retrofit and repair strategies that can provide 100% functionality in both residential and non-residential sectors. Figure 7-9 shows one of the retrofit schemes that provides complete functionality of the sectors, considering that an optimized network recovery will be done after the earthquake.



Figure 7-9. Retrofit scheme that results in 100% functionality in both sectors after 90 days for the considered scenario

Tables 7-1 and 7-2 show several polices that represent different combinations of retrofit and repair schemes for the Centerivlle NGN under all scenarios considered in this dissertation. For simplifications, for each scenario, at most three Pareto optimal solutions have been selected to be shown in the tables as possible solutions for policy-makers to consider: one that minimizes the number of people affected, one that minimizes the amounts of business losses, and one that is inbetween the two.

	Repair Crew	Policy	Day					
Retrofit Budget			30 days		60 days		90 days	
			People	BL	People	BL	People	BL
\$2.8	R = 2	P1	50,000	\$30	11,500	\$47.7	0	\$49.6
		P2	50,000	\$30	9,500	\$57.5	0	\$62.4
		P3	50,000	\$30	15,000	\$43.7	0	\$44.2
	R = 5	P4	15,500	\$30	13,500	\$33.6	0	\$33.8
		P5	50,000	\$30	0	\$42.2	0	\$42.2
	R = 10	P6	13,500	\$30	0	\$43.8	0	\$44.3
		P7	50,000	\$30	0	\$38.7	0	\$38.7
\$5.6	R = 2	P8	26,500	\$28	13,000	\$43.4	0	\$43.5
		P9	50,000	\$30	0	\$44.3	0	\$44.5
		P10	50,000	\$28	2,500	\$43.4	0	\$43.4
	R = 5	P11	13,500	\$28	0	\$33.5	0	\$33.5
		P12	0	\$30	0	\$38.6	0	\$38.6
	R = 10	P13	13,500	\$28	0	\$33.5	0	\$33.6
		P14	0	\$30	0	\$38.6	0	\$38.6

Table 7-1. Results for the earthquake with 10% chance in 50 years (RP = 475 yrs)

Note: the retrofit budget and business losses (BL) are in million dollars

Retrofit Budget	Repair Crew	Policy	Day					
			30 days		60 days		90 days	
			People	BL	People	BL	People	BL
\$2.8	R = 2	P15	50,000	\$30	50,000	\$60	28,500	\$76.3
		P16	50,000	\$30	50,000	\$60	21,500	\$89.9
		P17	50,000	\$30	50,000	\$60	44,500	\$72.1
	R = 5	P18	50,000	\$30	15,500	\$55.2	13,500	\$65.3
		P19	50,000	\$30	37,500	\$47.1	0	\$84.5
	R = 10	P20	50,000	\$30	13,500	\$51.1	0	\$63.1
\$5.6	R = 2	P21	50,000	\$30	50,000	\$60	17,500	\$71.9
		P22	50,000	\$30	50,000	\$60	9,500	\$83.1
		P23	50,000	\$30	50,000	\$60	48,500	\$61.2
	R = 5	P24	50,000	\$30	13,500	\$56.3	0	\$57.5
		P25	50,000	\$30	29,500	\$48.1	0	\$49.8
	R = 10	P26	50,000	\$30	13,500	\$56.4	0	\$57.7

Table 7-2. Results for the earthquake with 2% chance in 50 years (RP = 2475 yrs)

Note: the retrofit budget and business losses (BL) are in million dollars

As shown in the tables, each policy (P) provides information with regard to the trade-offs between the number of people affected and the amount of business losses at different time frames, considering different budgets for retrofit, the number of repair crews, and two different design earthquakes. The results from these tables can provide useful information for decision making purposes. For example, consider policies P4 and P5, which represent the results for the case of an earthquake with 475 years return period, when a budget of \$2.8 million dollars and five repair crews are available for retrofit and repair, respectively. Comparing these policies together, adopting policy P4 results in a smaller number of people being affected after 30 days (15,500 people instead of 50,000) and reduces the amounts of business losses in the long run (\$33.8 million dollars after 90 days). Nonetheless, using this policy results in 13,500 people not having access to natural gas after 60 days. Therefore, if the goal is to have 100% functionality of natural gas network in the residential sectors by the day 60, then policy P5 should be implemented, even though the amount of business losses after 90 days will increase for this case (\$42.2 million dollars).

Furthermore, for the earthquake with 475 years return period, the results show that when a budget of 5.6 million dollars are available for retrofit, arranging more than five repair crews would be unnecessary for repairing the damaged components. Nontheless, considering five crews can be much more effective than two crews for this case. Moreover, if under certain circumstances only two crews can be arranged and devoted for repairing the damaged components in the aftermath of the earthquake, then doubling the retrofit budget to strengthen the natural gas distribution pipelines does not seem to be a reasonable option, as there is not much difference between the results for these cases. Instead, it may be more reasonable to save the extra budget and invest it in either increasing the number of available repair crews after the event, or fortifying the key elements of the networks more extensively before the event occurs. Similar reasoning can also be made for the case of earhtquake with 2475 years return period.

Moreover, the question for which earthquake the decisions should be made is completely up to the decion-maker. A risk-taker may only consider the results for the earthquake with a return period of 475 years. Nontheless, if the decision-maker is a risk-averse person, they may select planning for the earthquake with 2475 years return period, which is a low-probability yet highconsequence event. Regardless, for someone who is risk neutral, both of the above-mentioned tables can provide sufficient information for decision making purposes at the community committee level.

Finally, the methodology developed in this dissertation can provide decision makers with useful information in order to prioritize retrofit actions. For example, in most scenarios under study, pipelines L_6 and L_{38} have been selected for retrofit. This reveals the importance of these pipelines as the key connections between the transmission and distribution networks. Thus, the developed methodology is capable of recognizing the critical components in the network that need to be prioritized in retrofit and planning projects at the community level.

CHAPTER 8: SUMMARY, CONTRIBUTIONS, AND RECOMMENDATIONS

SUMMARY

In this dissertation, a methodology was proposed and described for optimizing resilience decision making for natural gas networks under uncertainty. An optimization framework that integrated probabilistic seismic risk assessment of coupled infrastructure systems and evolutionary algorithms was proposed in order to determine cost-optimal decisions before and after a seismic event, with the goal of making the network recovery faster and thus contributing to a more resilient community. Specifically, strategic decisions were pursued regarding which pipelines in the natural gas distribution network of a community should be retrofitted under limited budgets, and how the limited available resources should be allocated to damaged components, considering minimizing two competing objective functions: the number of people without natural gas in the residential sector, and business losses due to a lack of natural gas in non-residential sectors. The proposed methodology can be used to find the best mitigation and recovery policies or to master-plan a new community or network.

As a basis for the proposed decision-making methodology, component models for natural gas networks were discussed by defining the network components at the community level and their damage models in chapter 2. In order to model the interdependencies between the natural gas network and other networks, approaches for modeling these interdependencies were explained in chapter 3. In chapter 4, earthquake hazard modeling was discussed and deterministic and probabilistic seismic hazard assessment methodologies were explained including an approach to maintain spatial correlation while utilizing hazard with a specified return period. The causes of damage to natural gas network components were also discussed in chapter 4. Chapter 5 introduced

a community restoration algorithm, primarily the natural gas network combined with electrical power, in order to model that portion of the recovery process in the aftermath of an earthquake. The developed algorithm was capable of representing the restoration process both spatially and temporally until full restoration of the network was achieved. Mitigation strategies and optimization of resource allocation processes were discussed in this chapter.

In order to illustrate the methodology proposed, chapter 6 represented a virtual community model that was used in this dissertation and subjected to earthquakes in order to examine the restoration of communities in the aftermath of an earthquake. The chapter discussed the topology and structure of the natural gas as well as electric power networks and described the bi-directional interdependency that should be considered across the networks. Finally, chapter 7 introduced a platform for optimal decision making, resilience assessment, seismic risk mitigation, and optimized recovery of natural gas networks, taking into account their interdependencies with other systems. An algorithm was developed in order to find the best locations for replacing the main steel distribution pipelines with earthquake-resistant HDPE pipelines for mitigation before the event and under limited budgets, as well as finding the best way to allocate the available yet limited repair crews to damaged components for recovery.

CONTRIBUTIONS

Modelling NGN at the community level

A robust simulation methodology was proposed for modeling natural gas networks including their interdependencies with other systems at the community level. The causes of damage to natural gas network components due to earthquakes were discussed and component damage models were provided. This simulation model can be coupled with any restoration model in order to investigate the recovery of natural gas networks under seismic hazards at the community level.

Optimized restoration analysis methodology

An algorithm was developed in this dissertation, which can model interdependencies between natural gas networks and other systems, and includes cascading failures in restoration of natural gas networks in the aftermath of an earthquake. The proposed restoration algorithm combines discrete event simulation (DES) and network-based approaches and can represent spatial and temporal depiction of the restoration process until the full restoration is achieved. Another advantage of the developed algorithm is its ability to perform optimized network recovery and prioritized resource allocation using genetic algorithms.

Optimizing resilience decision-making for NGN under uncertainty

A comprehensive approach for optimal decision making for resilience assessment, seismic risk mitigation, and recovery of natural gas networks, taking into account their interdependencies with other systems was developed. The optimization framework is capable of integrating sophisticated probabilistic descriptions of potential disruptive scenarios derived from seismic risk assessment with evolutionary algorithms in order to determine cost-optimal decisions before and after a seismic event. Using this platform, strategic decisions can be pursued regarding which pipelines in the natural gas distribution network in a community should be retrofitted under limited budgets before the hazard event, and how the limited available resources should be allocated to damaged components in the aftermath of the event, considering different competing objective functions. Therefore, the proposed methodology can be used for risk-informed prioritization of investments incorporating different levels of risk preferences.

Integrating the Platform into IN-CORE

The platform for optimal decision making under uncertainty has been developed in Python and can be integrated into the Interdependent Networked Community Resilience Modeling Environment (IN-CORE) in a straightforward fashion, in order to be coupled with other models for further analyses and investigations.

RECOMMENDATIONS

Incorporate community-level metrics as objective functions

Community-level metrics may be used as more comprehensive objective functions in the decision-making approach developed in this dissertation. Examples include using a socioeconomic resilience metric such as population outmigration as an alternative for the number of people not having access to natural gas.

Include direct and indirect business losses

The simplified methodology to evaluate business losses used in this dissertation can be replaced by a more complex model that can accurately relate the unavailability of natural gas to direct and indirect economic losses. Examples may include incorporating computable general equilibrium (CGE) models into the decision-making platform in order to make a more accurate estimation of the amount of business losses due to unavailability of natural gas in the community. It would also be appropriate to include other systems/networks in the community in the decision-making framework (e.g. water, food supply chains, etc.)

Combine additional mitigation approaches

In addition to the mitigation strategies adopted in this dissertation, other mitigation schemes may be added to the decision-making platform. One example is the installation of new pipelines in the natural gas distribution network in order to make the system more resilient by adding more redundancy to it.

Reduce computational costs

Despite the strength and capabilities of the algorithm developed in this dissertation, the amount of time it takes to converge to the optimal solution is lengthy. This is due to the stochastic nature of the models/methods (i.e. Monte Carlo simulation, probabilistic seismic hazard, and multi-objective genetic algorithm). It is therefore recommended to investigate other approaches that may reduce the complexity of the simulations. One suggestion may be replacing Monte Carlo simulation with Latin Hypercube sampling to reduce computational demand.

REFERENCES

ASME (2009). Innovative Technological Institute (ITI). Washington, D.C.: ASME ITI, LLC.

- Abrahamson, N. A. (2000). Effects of rupture directivity on probabilistic seismic hazard analysis.
 In Proceedings of the Sixth International Conference on Seismic Zonation: Managing Earthquake Risk in the 21st Century, Palm Springs, CA, 12-15 November 2000.
- Adachi, T., & Ellingwood, B. R. (2008). Serviceability of earthquake-damaged water systems:
 Effects of electrical power availability and power backup systems on system vulnerability.
 Reliability Engineering and System Safety. https://doi.org/10.1016/j.ress.2006.10.014
- Adachi, T., & Ellingwood, B. R. (2009). Serviceability assessment of electrical power transmission systems under probabilistically stated seismic hazards: Case study for Shelby County, Tennessee. *Structure and Infrastructure Engineering*. https://doi.org/10.1080/15732470701773358
- Akkar, S., Kale, Ö., Yakut, A., & Çeken, U. (2018). Ground-motion characterization for the probabilistic seismic hazard assessment in Turkey. *Bulletin of Earthquake Engineering*. https://doi.org/10.1007/s10518-017-0101-2
- ALA. (2001). Seismic Fragility Formulations for Water Systems: Part 1 guideline, American lifelines alliance. Retrieved from http://www.americanlifelinesalliance.org/
- Albert, R., Albert, I., & Nakarado, G. L. (2004). Structural vulnerability of the North American power grid. *Physical Review E - Statistical, Nonlinear, and Soft Matter Physics*. https://doi.org/10.1103/PhysRevE.69.025103

Alderson, D. L., Brown, G. G., Carlyle, W. M., & Wood, R. K. (2017). Assessing and Improving

the Operational Resilience of a Large Highway Infrastructure System to Worst-Case Losses. *Transportation Science*. https://doi.org/10.1287/trsc.2017.0749

- Ameri, M. R., & van de Lindt, J. W. (2019). Seismic Performance and Recovery Modeling of Natural Gas Networks at the Community Level Using Building Demand. *Journal of Performance of Constructed Facilities*, 33(4), 04019043.
- Amin, S. M. (2010). Electricity infrastructure security: Toward reliable, resilient and secure cyberphysical power and energy systems. In *IEEE PES General Meeting*, *PES 2010*. https://doi.org/10.1109/PES.2010.5589488
- Ancheta, T. D., Darragh, R. B., Stewart, J. P., Seyhan, E., Silva, W. J., Chiou, B. S.-J., ... Donahue, J. L. (2013). *PEER NGA-West2 Database*.
- Ancheta, T. D., Darragh, R. B., Stewart, J. P., Seyhan, E., Silva, W. J., Chiou, B. S.-J., ... Donahue,J. L. (2014). NGA-West2 Database. *Earthquake Spectra*, *30*(3), 989–1005.
- Applied Technology Council (ATC). (1985). ATC-13, Earthquake damage evaluation data for California. *Applied Technology Council*.
- Argyroudis, S., Selva, J., Gehl, P., & Pitilakis, K. (2015). Systemic seismic risk assessment of road networks considering interactions with the built environment. *Computer-Aided Civil and Infrastructure Engineering*. https://doi.org/10.1111/mice.12136
- Baker, Jack W. (2008). An introduction to Probabilistic Seismic Hazard Analysis (PSHA). US Nuclear Regulatory Commission.
- Baker, Jack Wesley. (2005). Vector-Valued Ground Motion Intensity Measures for Probabilistic
 Seismic Demand Analysis. PEER report, 1547-058X; 2006/08 TA -.
 https://doi.org/10.1016/j.matlet.2016.05.135

- Bana e Costa, C. A., Oliveira, C. S., & Vieira, V. (2008). Prioritization of bridges and tunnels in earthquake risk mitigation using multicriteria decision analysis: Application to Lisbon. *Omega*. https://doi.org/10.1016/j.omega.2006.05.008
- Baroud, H., Ramirez-Marquez, J. E., Barker, K., & Rocco, C. M. (2014). Stochastic Measures of Network Resilience: Applications to Waterway Commodity Flows. *Risk Analysis*. https://doi.org/10.1111/risa.12175
- Barton, D. C., Eidson, E. D., Schoenwald, D. A., Cox, R. G., & Reinert, R. K. (2004). Simulating Economic Effects of Disruptions in the Telecommunications Infrastructure. *Sandia Report*.
- Baso"z, N. I., Kiremidjian, A. S., King, S. A., & Law, K. H. (2003). Statistical Analysis of Bridge
 Damage Data from the 1994 Northridge, CA, Earthquake. *Earthquake Spectra*.
 https://doi.org/10.1193/1.1586027
- Basu, N., Pryor, R., & Quint, T. (1998). ASPEN: A Microsimulation Model of the Economy. Computational Economics. https://doi.org/10.1023/A:1008691115079
- Bazzurro, P., & Cornell, C. A. (1999). Disaggregation of seismic hazard. Bulletin of the Seismological Society of America.
- Berahman, F., & Behnamfar, F. (2007). Seismic fragility curves for un-anchored on-grade steel storage tanks: Bayesian approach. *Journal of Earthquake Engineering*. https://doi.org/10.1080/13632460601125722
- Bocchini, P., Decò, A., & Frangopol, D. M. (2012). Probabilistic functionality recovery model for resilience analysis. *Bridge Maintenance, Safety, Management, Resilience and Sustainability Biodini & Frangopol (Eds)*. https://doi.org/10.1201/b12352-283

Bommer, J. J., & Abrahamson, N. A. (2006). Why do modern probabilistic seismic-hazard

analyses often lead to increased hazard estimates? *Bulletin of the Seismological Society of America*. https://doi.org/10.1785/0120060043

- Boyce, G. M., & Bried, E. M. (1998). Social cost accounting for trenchless projects. In *Conf. on North American No-Dig* '98 (pp. 3–12).
- Brown, T., Beyeler, W., & Barton, D. (2004). Assessing infrastructure interdependencies: the challenge of risk analysis for complex adaptive systems. *International Journal of Critical Infrastructures*, 1(1). https://doi.org/10.1504/ijcis.2004.003800
- Bruneau, M., Chang, S. E., Eguchi, R. T., Lee, G. C., O'Rourke, T. D., Reinhorn, A. M., ... Von
 Winterfeldt, D. (2003). A Framework to Quantitatively Assess and Enhance the Seismic
 Resilience of Communities. *Earthquake Spectra*, 19(4), 733–752.
 https://doi.org/10.1193/1.1623497
- Bryson, K. M., Millar, H., Joseph, A., & Mobolurin, A. (2002). Using formal MS/OR modeling to support disaster recovery planning. *European Journal of Operational Research*. https://doi.org/10.1016/S0377-2217(01)00275-2
- Burroughs, S. M., & Tebbens, S. F. (2002). The upper-truncated power law applied to earthquake cumulative frequency-magnitude distributions: Evidence for a time-independent scaling parameter. *Bulletin of the Seismological Society of America*. https://doi.org/10.1785/0120010191
- Buzna, L., Peters, K., Ammoser, H., Kühnert, C., & Helbing, D. (2007). Efficient response to cascading disaster spreading. *Physical Review E - Statistical, Nonlinear, and Soft Matter Physics*. https://doi.org/10.1103/PhysRevE.75.056107

Cagnan, Z., & Davidson, R. A. (2007). Discrete event simulation of the post-earthquake restoration

process for electric power systems. *International Journal of Risk Assessment and Management*. https://doi.org/10.1504/ijram.2007.015298

- Çağnan, Z., Davidson, R. A., & Guikema, S. D. (2006). Post-earthquake restoration planning for Los Angeles electric power. *Earthquake Spectra*. https://doi.org/10.1193/1.2222400
- Çagnan, Z., Davidson, R., & Guikema, S. (2004). Post-earthquake restoration modeling of electric power systems. In *Proceedings of the 13th World Conference on Earthquake Engineering*.
- Campbell, K. W., & Bozorgnia, Y. (2013). NGA-West2 Campbell-Bozorgnia Ground Motion Model for the Horizontal Components of PGA, PGV, and 5%-Damped Elastic Pseudo-Acceleration Response Spectra for Periods Ranging from 0.01 to 10 sec.
- Campbell, K. W., & Bozorgnia, Y. (2014). NGA-West2 ground motion model for the average horizontal components of PGA, PGV, and 5% damped linear acceleration response spectra. *Earthquake Spectra*. https://doi.org/10.1193/062913EQS175M
- Casari, M., & Wilkie, S. J. (2005). Sequencing lifeline repairs after an earthquake: An economic approach. *Journal of Regulatory Economics*. https://doi.org/10.1007/s11149-004-4418-9
- Chang, Laing, & Song, J. (2007). Matrix-based system reliability analysis or urban infrastructure networks: a case study of MLGW natural gas network. 5th China-Japan-US Trilateral Symposium on Lifeline Earthquake Engineering.
- Chang, Liang, Peng, F., Ouyang, Y., Elnashai, A. S., & Spencer, B. F. (2012). Bridge Seismic Retrofit Program Planning to Maximize Postearthquake Transportation Network Capacity. *Journal of Infrastructure Systems*. https://doi.org/10.1061/(asce)is.1943-555x.0000082
- Chang, S. E., & Shinozuka, M. (2004). Measuring improvements in the disaster resilience of communities. *Earthquake Spectra*, 20(3), 739–755. https://doi.org/10.1193/1.1775796

- Chen, W. W., Shih, B. J., Chen, Y. C., Hung, J. H., & Hwang, H. H. (2002). Seismic response of natural gas and water pipelines in the Ji-Ji earthquake. *Soil Dynamics and Earthquake Engineering*. https://doi.org/10.1016/S0267-7261(02)00149-5
- Chiang, N.-Y., & Zavala, V. M. (2016). Large-scale optimal control of interconnected natural gas
 and electrical transmission systems. *Applied Energy*.
 https://doi.org/10.1016/j.apenergy.2016.01.017
- Choo, Y. W., Abdoun, T. H., O'Rourke, M. J., & Ha, D. (2007). Remediation for buried pipeline systems under permanent ground deformation. *Soil Dynamics and Earthquake Engineering*, 27(12), 1043–1055. https://doi.org/10.1016/j.soildyn.2007.04.002
- Chou, C.-C., & Tseng, S.-M. (2010). Collection and Analysis of Critical Infrastructure Interdependency Relationships. *Journal of Computing in Civil Engineering*. https://doi.org/10.1061/(asce)cp.1943-5487.0000059
- Cimellaro, G. P., Villa, O., & Bruneau, M. (2014). Resilience-Based Design of Natural Gas Distribution Networks. *Journal of Infrastructure Systems*. https://doi.org/10.1061/(asce)is.1943-555x.0000204
- Cimellaro, Gian Paolo, Solari, D., & Bruneau, M. (2014). Physical infrastructure interdependency and regional resilience index after the 2011 Tohoku Earthquake in Japan. *Earthquake Engineering and Structural Dynamics*. https://doi.org/10.1002/eqe.2422
- Coello, C. C., Dhaenens, C., & Jourdan, L. (2009). Advances in Multi-Objective Nature Inspired Computing. Studies in Computational Intelligence. https://doi.org/10.1007/978-3-642-11218-8

Cornell, C. A. (1968). Engineering seismic risk analysis. Bulletin of the Seismological Society of

America, 58(5), 1583–1606. https://doi.org/http://dx.doi.org/10.1016/0167-6105(83)90143-5

- Council., A. T. (2016). Critical assessment of lifeline system performance: Understanding societal needs in disaster recovery. Gaithersburg, MD: Prepared for US Department of Commerce National Institute of Standards and Technology, Engineering Laboratory. NIST GCR 16-917-39.
- D'Uffizi, A., Simonetti, M., Stecca, G., & Confessore, G. (2015). A simulation study of logistics for disaster relief operations. In *Procedia CIRP*. https://doi.org/10.1016/j.procir.2015.06.029
- Davis, C. (2008). Seismic practices to improve water system resilience. In *14th world conference on earthquake engineering*.
- Deb, K. (2014). Multi-objective optimization. In *Search methodologies* (pp. 403–449). Boston, MA: Springer.
- Deb, K., Pratap, A., Agarwal, S., & Meyarivan, T. (2002). A fast and elitist multiobjective genetic algorithm: NSGA-II. *IEEE Transactions on Evolutionary Computation*. https://doi.org/10.1109/4235.996017
- Delavaud, E., Cotton, F., Akkar, S., Scherbaum, F., Danciu, L., Beauval, C., ... Theodoulidis, N. (2012). Toward a ground-motion logic tree for probabilistic seismic hazard assessment in Europe. *Journal of Seismology*. https://doi.org/10.1007/s10950-012-9281-z
- Dietzenbacher, E., & Lahr, M. L. (2004). Wassily leontief and input-output economics. Wassily Leontief And Input-Output Economics. Cambridge University Press. https://doi.org/10.1017/CBO9780511493522
- Ditlevsen, O., & Bjerager, P. (1986). Methods of structural systems reliability. *Structural Safety*. https://doi.org/10.1016/0167-4730(86)90004-4

- Dong, Y., & Frangopol, D. M. (2017). Probabilistic assessment of an interdependent healthcare– bridge network system under seismic hazard. *Structure and Infrastructure Engineering*. https://doi.org/10.1080/15732479.2016.1198399
- Doocy, S., Daniels, A., Packer, C., Dick, A., & Kirsch, T. D. (2013). The Human Impact of Earthquakes: A Historical Review of Events 1980-2009 and Systematic Literature Review. *PLoS Currents*. https://doi.org/10.1371/currents.dis.67bd14fe457f1db0b5433a8ee20fb833
- Dueñas-Osorio, L., Craig, J. I., & Goodno, B. J. (2007). Seismic response of critical interdependent networks. *Earthquake Engineering and Structural Dynamics*. https://doi.org/10.1002/eqe.626
- Dueñas-Osorio, L., Craig, J. I., Goodno, B. J., & Bostrom, A. (2007). Interdependent Response of Networked Systems. *Journal of Infrastructure Systems*. https://doi.org/10.1061/(ASCE)1076-0342(2007)13:3(185)
- EERI. (1986). *Reducing Earthquake Hazards: Lessons Learned From Earthquakes*. Earthquake Engineering Research Institute.
- Eidinger, J. M., & Avila, E. A. (1999). Guidelines for the Seismic Evaluation and Upgrade of Water Transmission Facilities. Amercian Society of Civil Engineers; Technical Council on Lifeline Earthquake Engineering.
- Ellingwood, B. R. (2001). Earthquake risk assessment of building structures. *Reliability Engineering and System Safety*. https://doi.org/10.1016/S0951-8320(01)00105-3
- Ellingwood, B. R., Cutler, H., Gardoni, P., Peacock, W. G., van de Lindt, J. W., & Wang, N. (2016). The Centerville Virtual Community: a fully integrated decision model of interacting physical and social infrastructure systems. *Sustainable and Resilient Infrastructure*.

https://doi.org/10.1080/23789689.2016.1255000

- Energy, E. A. C. of the D. of. (2011). Interdependence of Electricity System Infrastructure and Natural Gas Infrastructure.
- Esposito, S. (2011). Systemic seismic risk analysis of gas distribution networks. University of Naples Federico.
- Esposito, S., Giovinazzi, S., Elefante, L., & Iervolino, I. (2013). Performance of the L'Aquila (central Italy) gas distribution network in the 2009 (Mw 6.3) earthquake. *Bulletin of Earthquake Engineering*. https://doi.org/10.1007/s10518-013-9478-8
- Esposito, S., Iervolino, I., d'Onofrio, A., Santo, A., Cavalieri, F., & Franchin, P. (2015). Simulation-Based Seismic Risk Assessment of Gas Distribution Networks. *Computer-Aided Civil and Infrastructure Engineering*, 30(7), 508–523. https://doi.org/10.1111/mice.12105
- Fan, Y., Liu, C., Lee, R., & Kiremidjian, A. S. (2009). Highway Network Retrofit under Seismic Hazard. *Journal of Infrastructure Systems*. https://doi.org/10.1061/(asce)is.1943-555x.0000024
- Fang, Y. P., & Sansavini, G. (2019). Optimum post-disruption restoration under uncertainty for enhancing critical infrastructure resilience. *Reliability Engineering and System Safety*. https://doi.org/10.1016/j.ress.2018.12.002
- Fang, Y., & Zio, E. (2019). Game-Theoretic Decision Making for the Resilience of Interdependent Infrastructures Exposed to Disruptions. In *Critical Infrastructure Security and Resilience* (pp. 97–114). Springer. https://doi.org/10.1007/978-3-030-00024-0_6
- Farahmandfar, Z., Piratla, K. R., & Andrus, R. D. (2016). Resilience Evaluation of Water Supply Networks against Seismic Hazards. *Journal of Pipeline Systems Engineering and Practice*.

https://doi.org/10.1061/(asce)ps.1949-1204.0000251

- Farmer, R. G., & Allen, E. H. (2006). Power system dynamic performance advancement from history of North American blackouts. In 2006 IEEE PES Power Systems Conference and Exposition, PSCE 2006 - Proceedings. https://doi.org/10.1109/PSCE.2006.296322
- FEMA. (2003). Multi-hazard loss estimation methodology, earthquake model, HAZUS-MH 2.1 Technical Manual, 1–699. Washington, DC: Federal Emergency Management Agency.
- Folga, S. M. (2007). *Natural gas pipeline technology overview*. Argonne, IL. https://doi.org/10.2172/925391
- Fonseca, C. M., & Fleming, P. J. (1993). Genetic Algorithms for Multiobjective Optimization: Formulation, Discussion and Generalization. In *Proceedings of the 5th International Conference on Genetic Algorithms*.
- Francis, R., & Bekera, B. (2014). A metric and frameworks for resilience analysis of engineered and infrastructure systems. *Reliability Engineering and System Safety*. https://doi.org/10.1016/j.ress.2013.07.004
- Frangopol, D. M., & Bocchini, P. (2011). Resilience As Optimization Criterion for the Rehabilitation of Bridges Belonging to a Transportation Network Subject to Earthquake. https://doi.org/10.1061/41171(401)178
- Gardoni, Paolo; Der Kiureghian, Armen; Mosalam, K. M. (2002). Probabilistic Models and Fragility Estimates for Bridge Components and Systems. Pacific Earthquake Engineering Research Center. https://doi.org/10.1061/(ASCE)0733-9429(1998)124:7(750)
- Gardoni, P., Mosalam, K. M., & Der Kiureghian, A. (2003). Probabilistic seismic demand models and fragility estimates for RC bridges. *Journal of Earthquake Engineering*.

https://doi.org/10.1080/13632460309350474

- Godschalk, D. R. (2003). Urban Hazard Mitigation: Creating Resilient Cities. *Natural Hazards Review*. https://doi.org/10.1061/(asce)1527-6988(2003)4:3(136)
- Gomez, C., Baker, J. W., Castiblanco, D., & Faber, M. H. (2015). Coupled pre- and post-disaster decisions in the context of infrastructure resilience. In *International Forum on Engineering Decision Making; Eight Forum: Decision Making Beyond the Engineering Comunity.*
- González, A. D., Dueñas-Osorio, L., Sánchez-Silva, M., & Medaglia, A. L. (2016). The Interdependent Network Design Problem for Optimal Infrastructure System Restoration. *Computer-Aided Civil and Infrastructure Engineering*. https://doi.org/10.1111/mice.12171
- Gordon, P., Richardson, H. W., & Davis, B. (1998). Transport-Related Impacts of the Northridge Earthquake. *Journal of Transportation and Statistics*.
- Guidotti, R., Chmielewski, H., Unnikrishnan, V., Gardoni, P., McAllister, T., & van de Lindt, J. (2016). Modeling the resilience of critical infrastructure: the role of network dependencies. *Sustainable and Resilient Infrastructure*, 1(3–4), 153–168. https://doi.org/10.1080/23789689.2016.1254999
- Guidotti, R., Gardoni, P., & Rosenheim, N. (2019). Integration of physical infrastructure and social systems in communities' reliability and resilience analysis. *Reliability Engineering and System Safety*. https://doi.org/10.1016/j.ress.2019.01.008
- Guikema, S. D., Nateghi, R., Quiring, S. M., Staid, A., Reilly, A. C., & Gao, M. (2014). Predicting Hurricane Power Outages to Support Storm Response Planning. *IEEE Access*. https://doi.org/10.1109/ACCESS.2014.2365716
- Gupta, I. D. (2007). Probabilistic seismic hazard analysis method for mapping of spectral

amplitudes and other design-specific quantities to estimate the earthquake effects on manmade structures. *ISET Journal of Earthquake Technology*.

- Gutenberg, B., & Richter, C. F. (1944). Frequency of earthquakes in California. *Bulletin of the Seismological Society of America*, *34*(4), 185–188.
- Haimes, Y. Y., Horowitz, B. M., Lambert, J. H., Santos, J. R., Lian, C., & Crowther, K. G. (2005).
 Inoperability Input-Output Model for Interdependent Infrastructure Sectors. I: Theory and Methodology. *Journal of Infrastructure Systems*. https://doi.org/10.1061/(asce)1076-0342(2005)11:2(67)
- Haimes, Y. Y., & Jiang, P. (2001). Leontief-Based Model of Risk in Complex Interconnected Infrastructures. *Journal of Infrastructure Systems*, 7(1). https://doi.org/10.1061/(asce)1076-0342(2001)7:1(1)
- Hainzl, S., Scherbaum, F., & Beauval, C. (2006). Estimating background activity based on interevent-time distribution. *Bulletin of the Seismological Society of America*. https://doi.org/10.1785/0120050053
- Hamada, M. (1991). Estimation of earthquake damage to lifeline systems in Japan. In 3rd Japan-U.S. Workshop on Earthquake REsistant Design of Lifeline Facilities and Countermeasures for Soil Liquefaction, Technical Rep. NCEER-91-0001 (pp. 6–22). Buffalo: National Center for Earthquake Engineering Research.
- Han, Z. Y., & Weng, W. G. (2011). Comparison study on qualitative and quantitative risk assessment methods for urban natural gas pipeline network. *Journal of Hazardous Materials*, *189*(1–2), 509–518. https://doi.org/10.1016/j.jhazmat.2011.02.067

Healy, P. M., & Palepu, K. G. (2003). The fall of Enron. IEEE Engineering Management Review.

https://doi.org/10.1109/EMR.2003.24934

- Heracleous, C., Kolios, P., Panayiotou, C. G., Ellinas, G., & Polycarpou, M. M. (2017). Hybrid systems modeling for critical infrastructures interdependency analysis. *Reliability Engineering and System Safety*. https://doi.org/10.1016/j.ress.2017.03.028
- Hertz, A., Jaumard, B., Ribeiro, C. C., & Formosinho Filho, W. P. (1994). A multi-criteria tabu search approach to cell formation problems in group technology with multiple objectives. *RAIRO - Operations Research*. https://doi.org/10.1051/ro/1994280303031
- Holland, J. H. (1975). Adaptation in Natural and Artificial Systems: An introductory Analysis with Applications to Biology, Control and Artificial Intelligence. MIT Press. https://doi.org/10.1137/1018105
- Hollnagel, E., Woods, D. D., & Leveson, N. (2006). (2006). *Resilience engineering: Concepts and precepts*. Ashgate Publishing, Ltd.
- Hosseini, S., Barker, K., & Ramirez-Marquez, J. E. (2016). A review of definitions and measures of system resilience. *Reliability Engineering and System Safety*, 145, 47–61. https://doi.org/10.1016/j.ress.2015.08.006
- Hu, X., Daganzo, C., & Madanat, S. (2015). A reliability-based optimization scheme for maintenance management in large-scale bridge networks. *Transportation Research Part C: Emerging Technologies*. https://doi.org/10.1016/j.trc.2015.01.008
- Huling, D., & Miles, S. B. (2015). Simulating disaster recovery as discrete event processes using python. In *Proceedings of the 5th IEEE Global Humanitarian Technology Conference*, *GHTC 2015*. https://doi.org/10.1109/GHTC.2015.7343980

Iervolino, I., Fabbrocino, G., & Manfredi, G. (2004). Fragility of standard industrial structures by

a response surface based method. *Journal of Earthquake Engineering*, 8(6), 927–945. https://doi.org/10.1080/13632460409350515

- Ishibuchi, H., & Murata, T. (1998). A multi-objective genetic local search algorithm and its application to flowshop scheduling. *IEEE Transactions on Systems, Man and Cybernetics Part C: Applications and Reviews*. https://doi.org/10.1109/5326.704576
- Isoyama, R., Ishida, E., Yune, K., & Shirozu, T. (2000). Seismic damage estimation procedures for water supply pipelines. *Water Supply*.
- Javanbakht, P., & Mohagheghi, S. (2014). A risk-averse security-constrained optimal power flow for a power grid subject to hurricanes. *Electric Power Systems Research*. https://doi.org/10.1016/j.epsr.2014.07.018
- Jayaram, N., & Baker, J. W. (2010). Efficient sampling and data reduction techniques for probabilistic seismic lifeline risk assessment. *Earthquake Engineering and Structural Dynamics*. https://doi.org/10.1002/eqe.988
- Jeon, S. S., & O'Rourke, T. D. (2005). Northridge earthquake effects on pipelines and residential buildings. Bulletin of the Seismological Society of America, 95(1), 294–318. https://doi.org/10.1785/0120040020
- Jia, G., Tabandeh, A., & Gardoni, P. (2017). Life-cycle analysis of engineering systems: Modeling deterioration, instantaneous reliability, and resilience. In *Springer Series in Reliability Engineering*. https://doi.org/10.1007/978-3-319-52425-2_20
- Jibson, R. W., Harp, E. L., & Michael, J. A. (2000). A method for producing digital probabilistic seismic landslide hazard maps. *Engineering Geology*. https://doi.org/10.1016/S0013-7952(00)00039-9

- Johansson, J., & Hassel, H. (2010). An approach for modelling interdependent infrastructures in the context of vulnerability analysis. *Reliability Engineering and System Safety*. https://doi.org/10.1016/j.ress.2010.06.010
- Kaegi, M., Mock, R., & Kröger, W. (2009). Analyzing maintenance strategies by agent-based simulations: A feasibility study. *Reliability Engineering and System Safety*. https://doi.org/10.1016/j.ress.2009.02.002
- Kagan, Y. Y. (1991). Seismic Moment Distribution. *Geophysical Journal International*, 106(1), 123–134.
- Kagan, Y. Y. (2002a). Seismic moment distribution revisited: I. Statistical results. *Geophysical Journal International*. https://doi.org/10.1046/j.1365-246x.2002.01594.x
- Kagan, Y. Y. (2002b). Seismic moment distribution revisited: II. Moment conservation principle. *Geophysical Journal International*. https://doi.org/10.1046/j.1365-246X.2002.01671.x
- Kidnay, A. J., Parrish, W. R., & McCartney, D. G. (2011). *Fundamentals of natural gas processing*. CRC Press.
- Kijko, A. (2004). Estimation of the maximum earthquake magnitude, mmax. *Pure and Applied Geophysics*. https://doi.org/10.1007/s00024-004-2531-4
- Kishawy, H. A., & Gabbar, H. A. (2010). Review of pipeline integrity management practices. *International Journal of Pressure Vessels and Piping*. https://doi.org/10.1016/j.ijpvp.2010.04.003
- Knowles, J., & Corne, D. (1999). The Pareto archived evolution strategy: A new baseline algorithm for Pareto multiobjective optimisation. In *Proceedings of the 1999 Congress on Evolutionary Computation, CEC 1999.* https://doi.org/10.1109/CEC.1999.781913

- Koliou, M., van de Lindt, J. W., McAllister, T. P., Ellingwood, B. R., Dillard, M., & Cutler, H.
 (2018). State of the research in community resilience: progress and challenges. *Sustainable and Resilient Infrastructure*, 9689, 1–21. https://doi.org/10.1080/23789689.2017.1418547
- Kong, J., & Simonovic, S. P. (2019). Probabilistic Multiple Hazard Resilience Model of an Interdependent Infrastructure System. *Risk Analysis*. https://doi.org/10.1111/risa.13305
- Krinitzsky, E. L. (2002). How to obtain earthquake ground motions for engineering design. *Engineering Geology*. https://doi.org/10.1016/S0013-7952(01)00098-9
- Lanzano, G., Salzano, E., Santucci de Magistris, F., & Fabbrocino, G. (2014). Seismic vulnerability of gas and liquid buried pipelines. *Journal of Loss Prevention in the Process Industries*, 28, 72–78. https://doi.org/10.1016/j.jlp.2013.03.010
- Lanzano, Giovanni, Salzano, E., De Magistris, F. S., & Fabbrocino, G. (2013). Seismic vulnerability of natural gas pipelines. *Reliability Engineering and System Safety*, 117, 73–80. https://doi.org/10.1016/j.ress.2013.03.019
- Lapajne, J., Motnikar, B. Š., & Zupančič, P. (2003). Probabilistic seismic hazard assessment methodology for distributed seismicity. *Bulletin of the Seismological Society of America*. https://doi.org/10.1785/0120020182
- Lee, E. E., Mitchell, J. E., & Wallace, W. A. (2007). Restoration of services in interdependent infrastructure systems: A network flows approach. *IEEE Transactions on Systems, Man and Cybernetics Part C: Applications and Reviews*. https://doi.org/10.1109/TSMCC.2007.905859

Leontief, W. (1986). Input-ouput economics. In *Input-Output economics*. Oxford University Press. Liao, T. Y., Hu, T. Y., & Ko, Y. N. (2018). A resilience optimization model for transportation
networks under disasters. Natural Hazards. https://doi.org/10.1007/s11069-018-3310-3

- Lim, H. W., & Song, J. (2012). Efficient risk assessment of lifeline networks under spatially correlated ground motions using selective recursive decomposition algorithm. *Earthquake Engineering and Structural Dynamics*. https://doi.org/10.1002/eqe.2162
- Liu, H., Davidson, R. A., & Apanasovich, T. V. (2007). Statistical forecasting of electric power restoration times in hurricanes and ice storms. *IEEE Transactions on Power Systems*. https://doi.org/10.1109/TPWRS.2007.907587
- Liu, W., Li, Z., Song, Z., & Li, J. (2018). Seismic reliability evaluation of gas supply networks based on the probability density evolution method. *Structural Safety*, 70, 21–34. https://doi.org/10.1016/j.strusafe.2017.10.001
- Liu, X., Ferrario, E., & Zio, E. (2017). Resilience Analysis Framework for Interconnected Critical Infrastructures. ASCE-ASME J. Risk and Uncert. in Engrg. Sys., Part B: Mech. Engrg. https://doi.org/10.1115/1.4035728
- Lomnitz-Adler, J., & Lomnitz, C. (1979). A modified form of the Gutenberg-Richter magnitudefrequency relation. *Bulletin of the Seismological Society of America*.
- Main, I. (1996). Statistical physics, seismogenesis, and seismic hazard. *Reviews of Geophysics*. https://doi.org/10.1029/96RG02808
- Main, I. G., & Burton, P. W. (1984). Information theory and the earthquake frequency-magnitude distribution. *Bulletin of the Seismological Society of America*.
- Maison, B., Lee, D., Lau, B., & Eidinger, J. (1995). East Bay municipality utility district water distribution damage in 1989 Loma Prieta earthquake. In *Proceedings of the fourth US* conference on lifeline earthquake engineering, Monograph 6. ASCE. New York.

Manual, T. (n.d.). Hazus –MH 2.1.

- Markowski, A. S., & Mannan, M. S. (2009). Fuzzy logic for piping risk assessment (pfLOPA). Journal of Loss Prevention in the Process Industries, 22(6), 921–927. https://doi.org/10.1016/j.jlp.2009.06.011
- Maruyama, Y., & Yamazaki, F. (2010). Construction of Fragility Curve for Water Distribution
 Pipes Based on Damage Datasets From Recent Earthquakes in Japan. *Proceedings of the 9th*U.S. National and 10th Canadian Conference on Earthquake Engineering, July 25-29, 2010,
 Toronto, Ontario, Canada.
- Maskrey, A. (1989). Disaster Mitigation: A Community Based Approach. Oxfam Publishing. https://doi.org/10.1902/jop.2013.130167
- Masoomi, H., Ameri, M.R., and van de Lindt, J. W. (2017). Wind performance enhancement strategies for wood-frame buildings. *Journal of Performance of Constructed Facilities* (Accepted).
- Masoomi, H., van de Lindt, J.W., Ameri, M.R., Do. T.Q., and W. B. M. (2018). Combined Wind-Wave-Surge Hurricane-Induced Damage Prediction for Buildings. *Journal of Structural Engineering (In Review)*.
- Masoomi, H., & van de Lindt, J. W. (2018a). Community-Resilience-Based Design of the Built Environment. ASCE-ASME Journal of Risk and Uncertainty in Engineering Systems, Part A: Civil Engineering. https://doi.org/10.1061/ajrua6.0000998
- Masoomi, H., & van de Lindt, J. W. (2018b). Restoration and functionality assessment of a community subjected to tornado hazard. *Structure and Infrastructure Engineering*, 14(3), 275–291. https://doi.org/10.1080/15732479.2017.1354030

- Matisziw, T. C., Murray, A. T., & Grubesic, T. H. (2010). Strategic network restoration. *Networks* and Spatial Economics. https://doi.org/10.1007/s11067-009-9123-x
- Maya Duque, P. A., Dolinskaya, I. S., & Sörensen, K. (2016). Network repair crew scheduling and routing for emergency relief distribution problem. *European Journal of Operational Research*. https://doi.org/10.1016/j.ejor.2015.06.026
- McDaniels, T., Chang, S., Peterson, K., Mikawoz, J., & Reed, D. (2007). Empirical Framework for Characterizing Infrastructure Failure Interdependencies. *Journal of Infrastructure Systems*. https://doi.org/10.1061/(ASCE)1076-0342(2007)13:3(175)
- McGuire, R. K. (2001). Deterministic vs. probabilistic earthquake hazards and risks. *Soil Dynamics and Earthquake Engineering*. https://doi.org/10.1016/S0267-7261(01)00019-7
- Memari, M., Attary, N., Masoomi, H., Mahmoud, H., van de Lindt, J., Pilkington, S., and Ameri, M. R. (2017). Minimal Building Fragility Portfolio for Physics-based Damage Assessment of Communities Subjected to Tornadoes. *Journal of Structural Engineering (Accepted)*.
- Merz, H. A., & Cornell, C. A. (1973). Seismic risk analysis based on a quadratic magnitudefrequency law. *Bulletin of the Seismological Society of America*, 63(6–1).
- Michalewicz, Z. (2013). Genetic algorithms + data structures = evolution programs. Springer Science & Business Media,.
- Miles, S. B., & Chang, S. E. (2003). Urban Disaster Recovery: A Framework and Simulation Model.
- Miles, S. B., & Chang, S. E. (2004). Foundations for Modeling Community Recovery from Earthquake Disasters. In 13th world Conference on Earthquake Engineering. https://doi.org/10.1193/1.2192847

- Miles, S. B., & Chang, S. E. (2006). Modeling community recovery from earthquakes. *Earthquake Spectra*. https://doi.org/10.1193/1.2192847
- Miller, M., & Baker, J. (2015). Ground-motion intensity and damage map selection for probabilistic infrastructure network risk assessment using optimization. *Earthquake Engineering and Structural Dynamics*. https://doi.org/10.1002/eqe.2506
- Mulargia, F., Stark, P. B., & Geller, R. J. (2017). Why is Probabilistic Seismic Hazard Analysis (PSHA) still used? *Physics of the Earth and Planetary Interiors*. https://doi.org/10.1016/j.pepi.2016.12.002
- Nan, C., & Sansavini, G. (2015). Multilayer hybrid modeling framework for the performance assessment of interdependent critical infrastructures. *International Journal of Critical Infrastructure Protection*. https://doi.org/10.1016/j.ijcip.2015.04.003
- Nateghi, R., Guikema, S. D., & Quiring, S. M. (2011). Comparison and Validation of Statistical Methods for Predicting Power Outage Durations in the Event of Hurricanes. *Risk Analysis*. https://doi.org/10.1111/j.1539-6924.2011.01618.x
- Noh, H. Y., Lallemant, D., & Kiremidjian, A. S. (2015). Development of empirical and analytical fragility functions using kernel smoothing methods. *Earthquake Engineering and Structural Dynamics*. https://doi.org/10.1002/eqe.2505
- North, M. J. (2001). Smart II: The spot market agent research tool version 2.0. *Natural Resources* and Environmental Issues.
- Nurre, S. G., Cavdaroglu, B., Mitchell, J. E., Sharkey, T. C., & Wallace, W. A. (2012). Restoring infrastructure systems: An integrated network design and scheduling (INDS) problem. *European Journal of Operational Research*. https://doi.org/10.1016/j.ejor.2012.07.010

- O'Rourke, M. J. (1988). Mitigation of seismic effects on water systems. In *Seismic Design and Construction of Complex Civil Engineering Systems* (pp. 65–78). ASCE.
- O'Rourke, M. J., & So, P. (2000). Seismic fragility curves for on-grade steel tanks. *Earthquake* Spectra. https://doi.org/10.1193/1.1586140
- O'Rourke, Michael, & Ayala, G. (1993). Pipeline Damage Due to Wave Propagation. *Journal of Geotechnical Engineering*. https://doi.org/10.1061/(asce)0733-9410(1993)119:9(1490)
- O'Rourke, Michael, & Deyoe, E. (2004). Seismic damage to segmented buried pipe. *Earthquake* Spectra. https://doi.org/10.1193/1.1808143
- O'Rourke, Michael, & Jeon, S.-S. (1999). Factors affecting water supply damage caused by the Northridge earthquake. In 5th U.S. National Conference on Earthquake Engineering. Seattle: EERI.
- O'Rourke, MJ, & Liu, X. (1999). Response of buried pipelines subject to earthquake effects. Retrieved from https://ubir.buffalo.edu/xmlui/handle/10477/588
- O'Rourke, Thomas D., Jeon, S. S., Toprak, S., Cubrinovski, M., Hughes, M., Van Ballegooy, S.,
 & Bouziou, D. (2014). Earthquake response of underground pipeline networks in Christchurch, NZ. *Earthquake Spectra*, 30(1), 183–204. https://doi.org/10.1193/030413EQS062M
- O'Rourke, Thomas Denis, Jeon, S. ., Toprak, S., Cubrinovski, M., & Jung, J. K. (2012). Underground Lifeline System Performance during the Canterbury Earthquake Sequence. *15 Wcee*.
- Omidvar, B., & Kivi, H. K. (2016). Multi-hazard failure probability analysis of gas pipelines for earthquake shaking, ground failure and fire following earthquake. *Natural Hazards*, 82(1),

703-720. https://doi.org/10.1007/s11069-016-2214-3

- Ouyang, M. (2014). Review on modeling and simulation of interdependent critical infrastructure systems. *Reliability Engineering and System Safety*. https://doi.org/10.1016/j.ress.2013.06.040
- Ouyang, M., Dueñas-Osorio, L., & Min, X. (2012). A three-stage resilience analysis framework for urban infrastructure systems. *Structural Safety*, 36–37, 23–31. https://doi.org/10.1016/j.strusafe.2011.12.004
- Ouyang, M., Hong, L., Mao, Z. J., Yu, M. H., & Qi, F. (2009). A methodological approach to analyze vulnerability of interdependent infrastructures. *Simulation Modelling Practice and Theory*. https://doi.org/10.1016/j.simpat.2009.02.001
- Ouyang, M., Pan, Z., Hong, L., & He, Y. (2015). Vulnerability analysis of complementary transportation systems with applications to railway and airline systems in China. *Reliability Engineering and System Safety*. https://doi.org/10.1016/j.ress.2015.05.013
- Ouyang, M., & Wang, Z. (2015). Resilience assessment of interdependent infrastructure systems:
 With a focus on joint restoration modeling and analysis. *Reliability Engineering and System Safety*, 141, 74–82. https://doi.org/10.1016/j.ress.2015.03.011
- Peeta, S., Sibel Salman, F., Gunnec, D., & Viswanath, K. (2010). Pre-disaster investment decisions for strengthening a highway network. *Computers and Operations Research*. https://doi.org/10.1016/j.cor.2009.12.006
- Percoco, M. (2004). Infrastructure and Economic Efficiency in Italian Regions. *Networks and Spatial Economics*. https://doi.org/10.1023/b:nets.0000047113.37578.91

Pereira, V., Rocha, M., & Sousa, P. (2016). Intelligent Distributed Computing IX. In The 9th

International Symposium on Intelligent Distributed Computing (pp. 485–495). Guimaraes, Portugal: Springer International Publishing. Retrieved from 10.1007/978-3-319-25017-5

- Petersen, M. D., Dawson, T. E., Chen, R., Cao, T., Wills, C. J., Schwartz, D. P., & Frankel, A. D. (2011). Fault displacement hazard for strike-slip faults. *Bulletin of the Seismological Society* of America. https://doi.org/10.1785/0120100035
- Pineda-Porras, O., & Ordaz-Schroeder, M. (2004). Seismic Vulnerability Function for High-Diameter Buried Pipelines: Mexico City's Primary Water System Case. https://doi.org/10.1061/40690(2003)131
- Pitilakis, K, Crowley, H., & Kaynia, A. M. (2014). SYNER-G: Typology Definition and Fragility Functions for Physical Elements at Seismic Risk. https://doi.org/10.1007/978-94-007-7872-6
- Pitilakis, Kyriazis, Alexoudi, M., Argyroudis, S., Monge, O., & Martin, C. (2006). Earthquake risk assessment of lifelines. *Bulletin of Earthquake Engineering*. https://doi.org/10.1007/s10518-006-9022-1
- Poe, W. A., & Mokhatab, S. (2016). Introduction to Natural Gas Processing Plants. In *Modeling*, *Control, and Optimization of Natural Gas Processing Plants*. https://doi.org/10.1016/b978-0-12-802961-9.00001-2
- Poljanšek, K., Bono, F., & Gutiérrez, E. (2012). Seismic risk assessment of interdependent critical infrastructure systems: The case of European gas and electricity networks. *Earthquake Engineering and Structural Dynamics*. https://doi.org/10.1002/eqe.1118
- Pope, S. L. (2002). California electricity price spikes: An update on the facts. Retrieved from https://hepg.hks.harvard.edu/publications/california-electricity-price-spikes-update-facts

Portante, E. C., Kavicky, J. A., Craig, B. A., Talaber, L. E., & Folga, S. M. (2017). Modeling

Electric Power and Natural Gas System Interdependencies. *Journal of Infrastructure Systems*, 23(4).

- Porter, K. (2015). A Beginner's Guide to Fragility, Vulnerability, and Risk. In *Encyclopedia of Earthquake Engineering*. https://doi.org/10.1007/978-3-642-36197-5_256-1
- PPD-21. (2013). Presidential policy directive/ PPD-21 Critical infrastructure security and resilience. Washington, DC: The White House.
- Praks, P., Kopustinskas, V., & Masera, M. (2017). Monte-Carlo-based reliability and vulnerability assessment of a natural gas transmission system due to random network component failures. *Sustainable and Resilient Infrastructure*, 2(3), 97–107. https://doi.org/10.1080/23789689.2017.1294881
- Psyrras, N. K., & Sextos, A. G. (2018). Safety of buried steel natural gas pipelines under earthquake-induced ground shaking: A review. *Soil Dynamics and Earthquake Engineering*. https://doi.org/10.1016/j.soildyn.2017.12.020
- Ramachandran, V., Shoberg, T., Long, S., Corns, S., & Carlo, H. (2015). Identifying Geographical Interdependency in Critical Infrastructure Systems Using Open Source Geospatial Data in Order to Model Restoration Strategies in the Aftermath of a Large-Scale Disaster. *International Journal of Geospatial and Environmental Research*.
- Refice, A., & Capolongo, D. (2002). Probabilistic modeling of uncertainties in earthquake-induced landslide hazard assessment. *Computers and Geosciences*. https://doi.org/10.1016/S0098-3004(01)00104-2
- Rinaldi, S. M., Peerenboom, J. P., & Kelly, T. K. (2001). Identifying, understanding, and analyzing critical infrastructure interdependencies. *IEEE Control Systems Magazine*.

https://doi.org/10.1109/37.969131

- Robert, B., & Morabito, L. (2008). The operational tools for managing physical interdependencies among critical infrastructures. *International Journal of Critical Infrastructures*. https://doi.org/10.1504/ijcis.2008.020156
- Rose, A. (1995). Input-output economics and computable general equilibrium models. *Structural Change and Economic Dynamics*. https://doi.org/10.1016/0954-349X(95)00018-I
- Rose, A. (2004). Economic Principles, Issues, and Research Priorities in Hazard Loss Estimation.
 In *Modeling Spatial and Economic Impacts of Disasters* (pp. 13–36). Springer, Berlin, Heidelberg. https://doi.org/10.1007/978-3-540-24787-6_2
- Rosowsky, D. (2019). Defining resilience. Unravelling Sustainability and Resilience in the Built Environment, 31–57. https://doi.org/10.4324/9781315629087-3
- Sajwan, M., Aggarwal, S., & Singh, R. B. (2008). Forensic characterization of HDPE pipes by DSC. *Forensic Science International*. https://doi.org/10.1016/j.forsciint.2007.05.020
- Schwartz, D. P., & Coppersmith, K. J. (1984). Fault behavior and characteristic earthquakes: examples from the Wasatch and San Andreas fault zones (USA). *Journal of Geophysical Research*. https://doi.org/10.1029/JB089iB07p05681
- Shafieezadeh, A, Leelardcharoen, K., & Dueñas-Osorio, L. (2014). A framework for assessing the effectiveness of resilience enhancement strategies for interdependent infrastructure systems.
 In Safety, Reliability, Risk and Life-Cycle Performance of Structures and Infrastructures. https://doi.org/10.1201/b16387-85
- Shafieezadeh, Abdollah, & Ivey Burden, L. (2014). Scenario-based resilience assessment framework for critical infrastructure systems: Case study for seismic resilience of seaports.

Reliability Engineering and System Safety. https://doi.org/10.1016/j.ress.2014.07.021

- Shahi, S. K., & Baker, J. W. (2011). An empirically calibrated framework for including the effects of near-fault directivity in probabilistic seismic hazard analysis. *Bulletin of the Seismological Society of America*. https://doi.org/10.1785/0120100090
- Shahidehpour, M., Yong, F. U., & Wiedman, T. (2005). Impact of natural gas infrastructure on electric power systems. In *Proceedings of the IEEE*. https://doi.org/10.1109/JPROC.2005.847253
- Sharma, N., Tabandeh, A., & Gardoni, P. (2017). Resilience analysis: a mathematical formulation to model resilience of engineering systems. *Sustainable and Resilient Infrastructure*, 9689(October), 1–19. https://doi.org/10.1080/23789689.2017.1345257
- Shinozuka, Masanobu, Feng, M. Q., Lee, J., & Naganuma, T. (2002). Statistical Analysis of Fragility Curves. *Journal of Engineering Mechanics*. https://doi.org/10.1061/(asce)0733-9399(2000)126:12(1224)
- Shinozuka, Masanubu, Chang, S. E., Cheng, T., Feng, M., O'Rourke, T. D., Saadeghvaziri, M. A., ... Pexixin, S. (2003). Resilience of Integrated Power and Water Systems. *Seismic Evaluation* and Retrofit of Lifeline Systems.
- Simon, D. (2013). Evolutionary Optimization Algorithms. The effects of brief mindfulness intervention on acute pain experience: An examination of individual difference. https://doi.org/10.1017/CBO9781107415324.004
- Singhal, A., & Kiremidjian, A. S. (2002). Method for Probabilistic Evaluation of Seismic Structural Damage. *Journal of Structural Engineering*. https://doi.org/10.1061/(asce)0733-9445(1996)122:12(1459)

- Stapelberg, R. (2008). Infrastructure systems interdependencies and risk informed decision making (RIDM): impact scenario analysis of infrastructure risks induced by natural, technological and intentional hazards. *Systemics, Cybernetics and Informatics*, 6(5).
- Stein, S., Tomasello, J., & Newman, A. (2003). Should memphis build for California's earthquakes? *Eos.* https://doi.org/10.1029/2003EO190002
- Straub, D., & Der Kiureghian, A. (2008). Improved seismic fragility modeling from empirical data. *Structural Safety*. https://doi.org/10.1016/j.strusafe.2007.05.004
- Tabucchi, T., Davidson, R., & Brink, S. (2010). Simulation of post-earthquake water supply system restoration. *Civil Engineering and Environmental Systems*. https://doi.org/10.1080/10286600902862615
- Titi, A., Biondini, F., & Frangopol, D. M. (2015). Seismic Resilience of Deteriorating Concrete Structures. In *Structures Congress*. https://doi.org/10.1061/9780784479117.142
- Tothoug, P., & Cornell, C. A. (2007). Probabilistic Seismic Demand Analysis Using Advanced
 Ground Motion Intensity Measures, Attenuation Relationships, and Near-Fault Effects.
 Pacific Earthquake Engineering Research Center.
 https://doi.org/10.1080/07321870601006565
- Tsinidis, G., Di Sarno, L., Sextos, A., & Furtner, P. (2019). A critical review on the vulnerability assessment of natural gas pipelines subjected to seismic wave propagation. Part 1: Fragility relations and implemented seismic intensity measures. *Tunnelling and Underground Space Technology*. https://doi.org/10.1016/j.tust.2019.01.025
- U.S. Energy Information Administration (EIA, 2017) <https://www.eia.gov/>
- U.S. News. (2019). Restoring Gas Heat in Rhode Island Could Take a Week or More. Retrieved

from https://www.usnews.com/news/best-states/rhode-island/articles/2019-01-22/nationalgrid-suspends-gas-to-7-100-during-frigid-temps

- Unnikrishnan, V. U., & van de Lindt, J. W. (2016). Probabilistic framework for performance assessment of electrical power networks to tornadoes. *Sustainable and Resilient Infrastructure*. https://doi.org/10.1080/23789689.2016.1254998
- Urlainis, A., Shohet, I. M., & Levy, R. (2015). Probabilistic Risk Assessment of Oil and Gas Infrastructures for Seismic Extreme Events. *Procedia Engineering*, 123, 590–598. https://doi.org/10.1016/j.proeng.2015.10.112
- USGS. (n.d.). Casualties and damage after the 1906 Earthquake. Retrieved from https://earthquake.usgs.gov/earthquakes/events/1906calif/18april/casualties.php
- USGS. (2017). Unified Hazard Tool. Retrieved from https://earthquake.usgs.gov/hazards/interactive/
- Utne, I. B., Hokstad, P., & Vatn, J. (2011). A method for risk modeling of interdependencies in critical infrastructures. In *Reliability Engineering and System Safety*. https://doi.org/10.1016/j.ress.2010.12.006
- Vamvatsikos, D., & Allin Cornell, C. (2002). Incremental dynamic analysis. *Earthquake Engineering and Structural Dynamics*. https://doi.org/10.1002/eqe.141
- van de Lindt, J. W., Ellingwood, B. R., McAllister, T., Gardoni, P., Cox, D. T., Cutler, H., & Peacock, W. G. (2015). Computational environment for modeling and enhancing community resilience: Introducing the center for risk-based community resilience planning. In *Proceedings of the Second International Conference on Performance-based and Life-cycle Structural Engineering (PLSE 2015)* (pp. 1154–1163). Brisbane, Australia.

https://doi.org/10.14264/uql.2016.555

- Visini, F., Valentini, A., Chartier, T., Scotti, O., & Pace, B. (2019). Computational Tools for Relaxing the Fault Segmentation in Probabilistic Seismic Hazard Modelling in Complex Fault Systems. *Pure and Applied Geophysics*. https://doi.org/10.1007/s00024-019-02114-6
- Vodák, R., Bíl, M., & Křivánková, Z. (2018). A modified ant colony optimization algorithm to increase the speed of the road network recovery process after disasters. *International Journal* of Disaster Risk Reduction. https://doi.org/10.1016/j.ijdrr.2018.04.004
- Wallace, W. A., Mendonça, D., Lee II, E. E., Mitchell, J. E., & Chow, J. H. (2003). Managing Disruptions to Critical Interdependent Infrastructures in the Context of the 2001 World Trade Center Attack. *Impacts of and Human Response to the September 11, 2001 Disasters: What Research Tells Us.* https://doi.org/10.1109/icsmc.2004.1401165
- Wang, S., Hong, L., & Chen, X. (2012). Vulnerability analysis of interdependent infrastructure systems: A methodological framework. *Physica A: Statistical Mechanics and Its Applications*. https://doi.org/10.1016/j.physa.2011.12.043
- Wisner, B., Blaikie, P., Cannon, T., & Davis, I. (2003). At Risk: Natural Hazards, People's
 Vulnerability, and Disasters Second Edition. *Human Ecology*. https://doi.org/10.4324/9780203428764
- Wu, B., Tang, A., & Wu, J. (2016). Modeling cascading failures in interdependent infrastructures under terrorist attacks. *Reliability Engineering and System Safety*. https://doi.org/10.1016/j.ress.2015.10.019
- Xie, X., Symans, M. D., O'Rourke, M. J., Abdoun, T. H., O'Rourke, T. D., Palmer, M. C., & Stewart, H. E. (2013). Numerical modeling of buried HDPE pipelines subjected to normal

faulting: A case study. *Earthquake Spectra*, 29(2), 609–632. https://doi.org/10.1193/1.4000137

- Xu, N., Guikema, S. D., Davidson, R. A., Nozick, L. K., Çağnan, Z., & Vaziri, K. (2007).
 Optimizing scheduling of post-earthquake electric power restoration tasks. *Earthquake Engineering and Structural Dynamics*. https://doi.org/10.1002/eqe.623
- Youn, B. D., Hu, C., & Wang, P. (2011). Resilience-Driven System Design of Complex Engineered Systems. *Journal of Mechanical Design*. https://doi.org/10.1115/1.4004981
- Youngs, R. R., & Coppersmith, K. J. (1985). Implications of fault slip rates and earthquake recurrence models to probabilistic seismic hazard estimates. *Bulletin of the Seismological Society of America*, 75(4), 939–964. https://doi.org/10.1016/0148-9062(86)90651-0
- Yuhua, D., & Datao, Y. (2005). Estimation of failure probability of oil and gas transmission pipelines by fuzzy fault tree analysis. *Journal of Loss Prevention in the Process Industries*, 18(2), 83–88. https://doi.org/10.1016/j.jlp.2004.12.003
- Yusoff, Y., Ngadiman, M. S., & Zain, A. M. (2011). Overview of NSGA-II for optimizing machining process parameters. In *Procedia Engineering*. https://doi.org/10.1016/j.proeng.2011.08.745
- Zanini, M. A., Faleschini, F., Zampieri, P., Pellegrino, C., Gecchele, G., Gastaldi, M., & Rossi, R. (2017). Post-quake urban road network functionality assessment for seismic emergency management in historical centres. *Structure and Infrastructure Engineering*. https://doi.org/10.1080/15732479.2016.1244211
- Zhang, P., & Peeta, S. (2011). A generalized modeling framework to analyze interdependencies among infrastructure systems. *Transportation Research Part B: Methodological*.

https://doi.org/10.1016/j.trb.2010.10.001

- Zhang, X., Mahadevan, S., & Goebel, K. (2019). Network Reconfiguration for Increasing Transportation System Resilience Under Extreme Events. *Risk Analysis*. https://doi.org/10.1111/risa.13320
- Zhang, Y., Yang, N., & Lall, U. (2016). Modeling and simulation of the vulnerability of interdependent power-water infrastructure networks to cascading failures. *Journal of Systems Science and Systems Engineering*. https://doi.org/10.1007/s11518-016-5295-3
- Zio, E., & Sansavini, G. (2011). Modeling interdependent network systems for identifying cascade-safe operating margins. *IEEE Transactions on Reliability*. https://doi.org/10.1109/TR.2010.2104211
- Zitzler, E. (1999). Evolutionary Algorithms for Multiobjective Optimization: Methods and Applications. Berichte aus der Informatik. Shaker Verlag, Aachen-. https://doi.org/citeulike-article-id:4597043
- Zitzler, E., & Thiele, L. (1999). Multiobjective evolutionary algorithms: A comparative case study and the strength Pareto approach. *IEEE Transactions on Evolutionary Computation*. https://doi.org/10.1109/4235.797969
- Zulfikar, C., Erdik, M., Safak, E., Biyikoglu, H., & Kariptas, C. (2016). Istanbul natural gas network rapid response and risk mitigation system. *Bulletin of Earthquake Engineering*, 14(9), 2565–2578. https://doi.org/10.1007/s10518-016-9964-x